DEVELOPMENT OF NEW MATHEMATICAL MODELLING FOR REMEDIATION PROCESS: CASE STUDIES ON REMEDIATION OF COPPER FROM WATER MATRICES USING CELLULOSE NANOWHISKER ADSORBENTS

NOR HAZREN ABDUL HAMID, MEng.

Thesis submitted to the University of Nottingham for the degree of

Doctor of Philosophy

SEPTEMBER 2016

ABSTRACT

Metal pollutants such as copper released into the aqueous environment have been increasing as a result of anthropogenic activities, a topic causing global concern. Adsorption-based treatment technologies offer opportunities to remediate metal pollutants from municipal and industrial wastewater effluent. The aim of this work was to evaluate the capability of modified cellulose nanowhisker (CNW) adsorbents for the remediation of copper from water matrices under realistic conditions using response surface methodology (RSM) and artificial neural network (ANN) models.

The first part of the study explored the preparation and characterisation of modified CNW adsorbents. It also focused on the stability of the modified CNW adsorbents at different time intervals under dry conditions (up to 28 days) and in the water matrix (up to 7 days). The results showed that the modified CNW adsorbents were stable at different time intervals under dry conditions and in the water matrix and proved that the functional groups were permanent and did not degrade under the tested conditions. The stability of these modified CNW adsorbents under these conditions, which is relevant from both the manufacturing and application perspectives, is reported for the first time in this study.

The second part of the work focused on using copper as a case study for heavy metal pollution in a clean water matrix, to evaluate removal by modified CNWs under several conditions and ranges appropriate to wastewater treatment plants (WWTPs), using factorial experimental design. RSM and ANN models were employed in order to optimise the system and to create a predictive model to evaluate the Cu(II) removal performance by the modified CNW adsorbents. Moreover, unseen experiments not belonging to the training data set, located both inside and outside the test parameter system, were performed to test the model suitability. This is also novel, as generally only one or two parameter variations have been tested, without checking the chosen model suitability for parameters lying between the tested parameters, and certainly not for parameters lying outside the tested parameter space, as has been done in this study. The results obtained showed that the ANN model outperformed the RSM model when predicting copper removal from a clean water matrix. The Langmuir and Freundlich isotherm models were applied to the equilibrium data, and the results revealed that the Langmuir isotherm ($R^2 = 0.9998$) had better correlation than the Freundlich isotherm ($R^2 = 0.9461$). Experimental data was also tested in terms of kinetics studies using pseudo-first order and pseudo-second order kinetic models. The results showed that the pseudo-second-order model accurately described the kinetics of adsorption.

The third part of the work was aimed at gaining a deeper understanding of the complexity and variability of the wastewater matrix, including evaluating the impact of the wastewater matrix temporally on adsorbent performance to remediate copper pollutant from a real-world wastewater matrix. This study has demonstrated that the wastewater matrix composition, which is both complex and variable, has an impact on adsorbent capability and performance. A benchmark study was adopted as a 'new' water quality parameter to inform on the effects of the wastewater matrix (wastewater composition and its variability) on the modified CNW adsorbent's capability to remediate copper from this matrix. Since the process of adsorption from wastewater is often complicated due to the variation in wastewater composition, results obtained from the benchmark experiments were included as one of the independent variables in ANN modelling, unlike in other optimisation studies. The performance of the ANN and RSM models was statistically evaluated in terms of coefficient of determination (R^2) , absolute average deviation (AAD), and root mean squared error (RMSE) on predicted experimental outcomes. The ANN model including the variability of wastewater composition fitted the experimental data with excellent accuracy and better prediction ($R^2 = 0.9963$) than both the ANN model that did not include this variability ($R^2 = 0.9945$), and the RSM model ($R^2 = 0.9409$). The outcome of this study showed that by supplying the ANN model with the data obtained from the benchmark experiments as the fourth independent variable, it was possible to improve the predictability of the ANN model.

Continuous flow experiments for remediation of spiked Cu(II) from the wastewater matrix were conducted. However, the physical structure of modified CNW adsorbents renders them unsuitable for use in column operation. Therefore, a more detailed study of the mechanical properties of CNW adsorbents would be necessary in order to improve the strength and stability of the adsorbents. This

work has demonstrated that modified CNW are promising adsorbents to remediate copper from water matrices under realistic conditions including wastewater complexity and variability. The use of models to predict the test parameter system and account for matrix variability when evaluating CNW adsorbents for remediating Cu from a real-world wastewater matrix may also provide the foundation for assessing other treatment technologies in the future.

PUBLICATION AND PRESENTATIONS

Publication

Hamid, H.A., Jenidi, Y., Thielemans, W., Somerfield, C., Gomes, R.L., Predicting the capability of carboxylated cellulose nanowhiskers for the remediation of copper from water using response surface methodology (RSM) and artificial neural network (ANN) models. *Industrial Crops and Products*. (In Press)

List of conference contributions

Hamid, H.A., Jenidi, Y., Thielemans, W., Somerfield, C., Gomes, R.L. Cottoning onto novel adsorbents: Predicting the capability of carboxylated cellulose nanowhiskers for the remediation of copper from water using mathematical modelling. International Water Resources Association (IWRA). May 25th - 29th, 2015, Edinburgh, Scotland. Interactive poster power point presentation

Hamid, H.A., Thielemans, W., Somerfield, C., Gomes, R.L. Cottoning onto novel adsorbent: Modified cotton for remediation of heavy metal from wastewater. University Research Showcase 2014. June 18th, 2014, Nottingham. Poster presentation

Hamid, H.A., Thielemans, W., Somerfield, C., Gomes, R.L. Adsorption studies of heavy metal ions onto TEMPO-mediated oxidation cellulose nanowhiskers by Response Surface Methodology. Sustainable Research Symposium. Sustainable Chemical and Biological Processing, University of Nottingham. April 23rd, 2014, Nottingham. Oral presentation

ACKNOWLEDGEMENTS

First of all, all praises and thanks to Allah, God Almighty for His countless blessings throughout my life and especially for the strength and determination to finish this study.

I would like to express my sincere gratitude and appreciation to Dr Rachel Gomes and Mr Chris Somerfield, my supervisors, for their guidance, support and patience during the course of my study. Without their supervisions, the completion of this study would not have been possible.

Special thanks must be given to Prof. Wim Thelemans, professor in the Renewable Materials and Nanotechnology Research Group, KU Leuven Campus Kulak Kortrijk, Belgium, for his expert advice on modification of cellulose nanowhiskers.

I would especially like to thank Dr Scott D Young for his expertise in ICP-MS analysis, Dr Nigel Neate for TEM, and Miss Vikki Archibald for AAS. Their assistance is very much appreciated.

I am obliged to Dr Liew Soon Yee, Dr Latifah Jasmani, Miss Youla Jenidi, and Mr Ian Clark, for their sincere assistance and help.

I am deeply grateful to Majlis Amanah Rakyat (MARA) for funding and supporting this PhD study.

Thank you to all my friends particularly "geng surau" for always being there when help is needed. Despite the long distances between us, my family have always supported me in whatever and whenever I decided to do and go, therefore, I am forever indebted to them. I extend my deepest gratitude to abah, mama, adik, deedat and ica for their prayers and unconditional support.

TABLE OF CONTENTS

PUBLICATION AND PRESENTATIONS iv ACKNOWLEDGEMENT v TABLE OF CONTENTS vi UIST OF FIGURES xii LIST OF FABLES xvi UIST OF ABBREVIATIONS xvi CHAPTER 1: INTRODUCTION 1 1.1 THESIS OVERVIEW 1.2 RESEARCH BACKGROUND 2 1.3 AIM AND OBJECTIVES 6 CHAPTER 2: LITERATURE REVIEW 7 2.1 IMPORTANCE OF WATER. 7 2.2 WATER POLLUTION BY HEAVY METALS 9 2.2.1 Sources of heavy metals in water. 9 2.2.2 Impact of heavy metals on the aquatic environment 11 2.2.3.1 Sources of Copper to humans and the environment 14 2.3.2 Toxicity of copper to humans and the environment 14 2.3.3 Environmental regulations and standards for discharge of effluent into water. 13 2.3.4 Legislative Limits for Copper in the Aquatic Environment 14 2.3.3 Conventional wastewater treatment approaches for copper removal 18 2.3 ADSORPTION PROCESS 21 2.3.1<	ABSTRA	СТ	i
ACKNOWLEDGEMENT	PUBLICA	ATION AND PRESENTATIONS	iv
TABLE OF CONTENTS vi LIST OF FIGURES xii LIST OF TABLES xvi LIST OF ABBREVIATIONS xx CHAPTER 1: INTRODUCTION 1 1.1 THESIS OVERVIEW 1.2 RESEARCH BACKGROUND 2.2 INAM AND OBJECTIVES 6 CHAPTER 2: LITERATURE REVIEW 7.1 IMPORTANCE OF WATER 7.2 WATER POLLUTION BY HEAVY METALS 9 2.2.1 2.2.2 Impact of heavy metals on the aquatic environment 11 2.3.3 Sources of Copper in the Aquatic Environment 13 2.3.4 Sources of Copper to humans and the environment 14 2.3.3 Environmental regulations and standards for discharge of effluent into water 17 2.3.4 Legislative Limits for Copper in the Aquatic Environment 17 2.3.5 Conventional wastewater treatment approaches for copper removal 18 2.3 ADSORPTION PROCESS 21 3.2.1 Langmuir Isotherm 23 2.3.2 Freundlich Isotherm 23 2.3.3 Factors affecting the rate of adsorption 21		WLEDGEMENT	v
LIST OF FIGURES xii LIST OF TABLES xvi LIST OF ABBREVIATIONS xx CHAPTER 1: INTRODUCTION 1 1.1 THESIS OVERVIEW 1.2 RESEARCH BACKGROUND 2.1.3 AIM AND OBJECTIVES 6 CHAPTER 2: LITERATURE REVIEW 7 2.1 IMPORTANCE OF WATER. 7 2.1 MAPOTANCE OF WATER. 7 2.1 IMPORTANCE OF WATER. 9 2.2.1 Sources of heavy metals in water 2.2 Impact of heavy metals on the aquatic environment 11 2.2.3 Coxices of Copper in the Aquatic Environment 13 2.3.2 Toxicity of copper to humans and the environment 14 2.3.3 Environmental regulations and standards for discharge of effluent into water water 15 2.3.4 Legislative Limits for Copper in the Aquatic Environment 17 2.3.4 Legislative Limits for Copper in the Aquatic Environment 17 12 2.3.5 Conventional wastewater treatment approaches for copper removal 18 18 3 ADSORPTION PROCESS <td>TABLE C</td> <td>DF CONTENTS</td> <td>vi</td>	TABLE C	DF CONTENTS	vi
LIST OF TABLES xvi LIST OF ABBREVIATIONS xx CHAPTER 1: INTRODUCTION 1 1.1 THESIS OVERVIEW 1 1.2 RESEARCH BACKGROUND 2 1.3 AIM AND OBJECTIVES 6 CHAPTER 2: LITERATURE REVIEW 7 2.1 IMPORTANCE OF WATER 7 2.2 WATER POLLUTION BY HEAVY METALS 9 2.2.1 Sources of heavy metals in water. 9 2.2.2 Impact of heavy metals on the aquatic environment 11 2.3.3 Copper as a case study heavy metal 13 2.3.4 Sources of Copper in the Aquatic Environment 14 2.3.3 Environmental regulations and standards for discharge of effluent into water 15 2.3.4 Legislative Limits for Copper in the Aquatic Environment 17 2.3.5 Conventional wastewater treatment approaches for copper removal 18 2.3 Absorption NROCESS 21 3.3.1 Mechanism of adsorption 22 2.3.2 Freundlich Isotherm 23 3.3.2 Factors affecting the rate of adsorption 25	LIST OF	FIGURES	xii
LIST OF ABBREVIATIONS xx CHAPTER 1: INTRODUCTION 1 1.1 THESIS OVERVIEW 1 1.2 RESEARCH BACKGROUND 2 1.3 AIM AND OBJECTIVES 6 CHAPTER 2: LITERATURE REVIEW 7 2.1 IMPORTANCE OF WATER 7 2.1 Sources of heavy metals in water. 9 2.2.1 Sources of heavy metals on the aquatic environment 11 2.2.3 Copper as a case study heavy metal 13 2.3.1 Sources of Copper in the Aquatic Environment 14 2.3.2 Toxicity of copper to humans and the environment 14 2.3.3 Environmental regulations and standards for discharge of effluent into water 15 2.3.4 Legislative Limits for Copper in the Aquatic Environment 17 2.3.5 Conventional wastewater treatment approaches for copper removal 18 2.3 ADSORPTION PROCESS 21 3.1 Mechanism of adsorption 22 2.3.2 Freundlich Isotherm 23 2.3.3 Foctors affecting the rate of adsorption 25 2.4 TRADITIONAL MODELLING OF TH	LIST OF	TABLES	vi
CHAPTER 1: INTRODUCTION 1 1.1 THESIS OVERVIEW 1 1.2 RESEARCH BACKGROUND 2 1.3 AIM AND OBJECTIVES 6 CHAPTER 2: LITERATURE REVIEW 7 2.1 IMPORTANCE OF WATER 7 2.2 WATER POLLUTION BY HEAVY METALS 9 2.2.1 Sources of heavy metals in water 9 2.2.2 Impact of heavy metals on the aquatic environment 11 2.2.3 Copper as a case study heavy metal 13 2.2.3.1 Sources of Copper in the Aquatic Environment 14 2.3.3 Environmental regulations and standards for discharge of effluent into water 15 2.2.3.4 Legislative Limits for Copper in the Aquatic Environment 17 2.3.3 Conventional wastewater treatment approaches for copper removal 18 2.3 Adsorption 21 3.3.1 Mechanism of adsorption 21 2.3.2 Freundlich Isotherm 23 2.3.3 Factors affecting the rate of adsorption 25 2.4 Adsorption Isotherm 23 2.3.4 Legislative Limits GOF THE ADS	LIST OF	ABBREVIATIONS	хх
1.1 THESIS OVERVIEW 1 1.2 RESEARCH BACKGROUND 2 1.3 AIM AND OBJECTIVES 6 CHAPTER 2: LITERATURE REVIEW 7 2.1 IMPORTANCE OF WATER 7 2.1 MORTANCE OF WATER 9 2.2.1 Sources of heavy metals in water 9 2.2.1 Sources of heavy metals on the aquatic environment 11 2.2.3 Copper as a case study heavy metal 13 2.3.1 Sources of Copper in the Aquatic Environment 14 2.3.2 Toxicity of copper to humans and the environment 14 2.3.3 Environmental regulations and standards for discharge of effluent into water 2.3.4 Legislative Limits for Copper in the Aquatic Environment 17 2.3.5 Conventional wastewater treatment approaches for copper removal 18 2.3 AdSorption isotherm 23 2.3.1 Mechanism of adsorption 21 2.3.2 Freundlich Isotherm 23 2.3.1 Langmuir Isotherm 23 2.3.2 Freundlich Isotherm 23 3.3 Factors affecting	СНАРТЕ	R 1: INTRODUCTION	1
1.2 RESEARCH BACKGROUND 2 1.3 AIM AND OBJECTIVES 6 CHAPTER 2: LITERATURE REVIEW 7 2.1 IMPORTANCE OF WATER 7 2.1 MORTANCE OF WATER 9 2.2.1 Sources of heavy metals in water. 9 2.2.2 Impact of heavy metals on the aquatic environment 11 2.3.3 Copper as a case study heavy metal 13 2.3.1 Sources of Copper in the Aquatic Environment 14 2.3.2 Toxicity of copper to humans and the environment 14 2.3.3 Environmental regulations and standards for discharge of effluent into water 15 2.3.4 Legislative Limits for Copper in the Aquatic Environment 17 2.3.5 Conventional wastewater treatment approaches for copper removal 18 2.3 ADSORPTION PROCESS 21 2.3.1 Mechanism of adsorption 22 2.3.2 Freundlich Isotherm 23 2.3.3 Factors affecting the rate of adsorption 25 2.3.4 Legislative Coupler of Adsorption 25 2.3.4 Legislative Coupler 30	1.1 TH	IESIS OVERVIEW	1
1.3 AIM AND OBJECTIVES 6 CHAPTER 2: LITERATURE REVIEW 7 2.1 IMPORTANCE OF WATER 7 2.2 WATER POLLUTION BY HEAVY METALS 9 2.2.1 Sources of heavy metals in water 9 2.2.2 Impact of heavy metals on the aquatic environment 11 2.3 Copper as a case study heavy metal 13 2.3.1 Sources of Copper in the Aquatic Environment 13 2.3.2 Toxicity of copper to humans and the environment 14 2.3.3 Environmental regulations and standards for discharge of effluent into water 15 2.2.3.4 Legislative Limits for Copper in the Aquatic Environment 17 2.3.5 Conventional wastewater treatment approaches for copper removal 18 2.3 ADSORPTION PROCESS 21 2.3.1 Mechanism of adsorption 22 2.3.2 Freundlich Isotherm 23 2.3.3 Factors affecting the rate of adsorption 25 2.4 TRADITIONAL MODELLING OF THE ADSORPTION PROCESS 30 2.5.1 Response Surface Methodology (RSM) 31 2.5.1.1 Theory and steps for RSM application 31 2.5.1.2 Factorial experimental design 31 2.5.1.3 Choice of experimental design 31 2.5.1.4 Evaluation	1.2 RE	ESEARCH BACKGROUND	2
CHAPTER 2: LITERATURE REVIEW 7 2.1 IMPORTANCE OF WATER 7 2.2 WATER POLLUTION BY HEAVY METALS 9 2.2.1 Sources of heavy metals in water 9 2.2.2 Impact of heavy metals on the aquatic environment 11 2.3.3 Copper as a case study heavy metal 13 2.3.1 Sources of Copper in the Aquatic Environment 14 2.3.2 Toxicity of copper to humans and the environment 14 2.3.3 Environmental regulations and standards for discharge of effluent into water 15 2.2.3.4 Legislative Limits for Copper in the Aquatic Environment 17 2.3.5 Conventional wastewater treatment approaches for copper removal 18 2.3 Absorption PROCESS 21 2.3.1 Mechanism of adsorption 21 2.3.2 Freundlich Isotherm 23 2.3.3 Factors affecting the rate of adsorption 25 2.4 TRADITIONAL MODELLING OF THE ADSORPTION PROCESS 30 2.5.1 Response Surface Methodology (RSM) 31 2.5.1.1 Theory and steps for RSM application 31 2.5.1.3	1.3 AI	M AND OBJECTIVES	6
2.1 IMPORTANCE OF WATER. 7 2.2 WATER POLLUTION BY HEAVY METALS 9 2.2.1 Sources of heavy metals in water. 9 2.2.2 Impact of heavy metals on the aquatic environment 11 2.2.3 Copper as a case study heavy metal 13 2.2.3.1 Sources of Copper in the Aquatic Environment 13 2.3.2 Toxicity of copper to humans and the environment 14 2.3.3 Environmental regulations and standards for discharge of effluent into water 15 2.3.4 Legislative Limits for Copper in the Aquatic Environment 17 2.3.5 Conventional wastewater treatment approaches for copper removal 18 2.3 Absorption isotherm 23 2.3.1 Mechanism of adsorption 21 2.3.2 Freundlich Isotherm 23 2.3.2.1 Langmuir Isotherm 23 2.3.2 Freundlich Isotherm 24 2.3.3 Factors affecting the rate of adsorption 25 2.4 TRADITIONAL MODELLING OF THE ADSORPTION PROCESS 30 2.5.1 Response Surface Methodology (RSM) 31 2.5.1.1	СНАРТЕ	R 2: LITERATURE REVIEW	7
2.2 WATER POLLUTION BY HEAVY METALS 9 2.2.1 Sources of heavy metals in water 9 2.2.2 Impact of heavy metals on the aquatic environment 11 2.2.3 Copper as a case study heavy metal 13 2.2.3.1 Sources of Copper in the Aquatic Environment 13 2.2.3.2 Toxicity of copper to humans and the environment 14 2.2.3.3 Environmental regulations and standards for discharge of effluent into water 15 2.3.4 Legislative Limits for Copper in the Aquatic Environment 17 2.3.5 Conventional wastewater treatment approaches for copper removal 18 2.3 Absorption PROCESS 21 2.3.1 Langmuir Isotherm 23 2.3.2 Freundlich Isotherm 23 2.3.3 Factors affecting the rate of adsorption 25 2.4 TRADITIONAL MODELLING OF THE ADSORPTION PROCESS 30 2.5.1 Response Surface Methodology (RSM) 31 2.5.1.1 Theory and steps for RSM application 31 2.5.1.2 Factorial experimental design 31 2.5.1.3 Choice of experimental design 31 </td <td>2.1 IN</td> <td>IPORTANCE OF WATER</td> <td>7</td>	2.1 IN	IPORTANCE OF WATER	7
2.2.1 Sources of heavy metals in water	2.2 W	ATER POLLUTION BY HEAVY METALS	9
2.2.2 Impact of heavy metals on the aquatic environment 11 2.2.3 Copper as a case study heavy metal 13 2.2.3.1 Sources of Copper in the Aquatic Environment 13 2.2.3.2 Toxicity of copper to humans and the environment 14 2.2.3.3 Environmental regulations and standards for discharge of effluent into water 15 2.2.3.4 Legislative Limits for Copper in the Aquatic Environment 17 2.2.3.5 Conventional wastewater treatment approaches for copper removal 18 2.3 ADSORPTION PROCESS 21 2.3.1 Mechanism of adsorption 21 2.3.2 Freundlich Isotherm 23 2.3.2.1 Langmuir Isotherm 23 2.3.2 Freundlich Isotherm 24 2.3.3 Factors affecting the rate of adsorption 25 2.4 TRADITIONAL MODELLING OF THE ADSORPTION PROCESS 30 2.5.1 Response Surface Methodology (RSM) 31 2.5.1.1 Theory and steps for RSM application 31 2.5.1.2 Factorial experimental design 31 2.5.1.3 Choice of experimental design 31 <tr< td=""><td>2.2.1</td><td>Sources of heavy metals in water</td><td>.9</td></tr<>	2.2.1	Sources of heavy metals in water	.9
2.2.3 Copper as a case study heavy metal 13 2.2.3.1 Sources of Copper in the Aquatic Environment 13 2.3.2 Toxicity of copper to humans and the environment 14 2.3.3 Environmental regulations and standards for discharge of effluent into water 15 2.3.4 Legislative Limits for Copper in the Aquatic Environment 17 2.3.5 Conventional wastewater treatment approaches for copper removal 18 2.3 ADSORPTION PROCESS 21 2.3.1 Mechanism of adsorption 21 2.3.2 Adsorption isotherm 23 2.3.2.1 Langmuir Isotherm 23 2.3.2.2 Freundlich Isotherm 23 2.3.3 Factors affecting the rate of adsorption 25 2.4 TRADITIONAL MODELLING OF THE ADSORPTION PROCESS 30 2.5.1 Response Surface Methodology (RSM) 31 2.5.1.2 Factorial experimental design 31 2.5.1.3 Choice of experimental design 31 2.5.1.4 Evaluation of fitted model 33 2.5.1.5 Optimisation of adsorption process using RSM 33 2.5.2 </td <td>2.2.2</td> <td>Impact of heavy metals on the aquatic environment</td> <td>11</td>	2.2.2	Impact of heavy metals on the aquatic environment	11
2.2.3.1 Sources of Copper in the Aquatic Environment 13 2.2.3.2 Toxicity of copper to humans and the environment 14 2.3.3 Environmental regulations and standards for discharge of effluent into water 15 2.3.4 Legislative Limits for Copper in the Aquatic Environment 17 2.3.5 Conventional wastewater treatment approaches for copper removal 18 2.3 ADSORPTION PROCESS 21 2.3.1 Mechanism of adsorption 21 2.3.2 Adsorption isotherm 23 2.3.2.1 Langmuir Isotherm 23 2.3.2.2 Freundlich Isotherm 24 2.3.3 Factors affecting the rate of adsorption 25 2.4 TRADITIONAL MODELLING OF THE ADSORPTION PROCESS 30 2.5.1 Response Surface Methodology (RSM) 31 2.5.1.1 Theory and steps for RSM application 31 2.5.1.2 Factorial experimental design 31 2.5.1.3 Choice of experimental design 31 2.5.1.4 Evaluation of fitted model 33 2.5.1.5 Optimisation of adsorption process using RSM 33 2	2.2.3	Copper as a case study heavy metal	13
2.2.3.2 Toxicity of copper to humans and the environment 14 2.2.3.3 Environmental regulations and standards for discharge of effluent into water 15 2.2.3.4 Legislative Limits for Copper in the Aquatic Environment 17 2.2.3.5 Conventional wastewater treatment approaches for copper removal 18 2.3 ADSORPTION PROCESS 21 2.3.1 Mechanism of adsorption 21 2.3.2 Adsorption isotherm 23 2.3.2.1 Langmuir Isotherm 23 2.3.2.2 Freundlich Isotherm 23 2.3.3 Factors affecting the rate of adsorption 25 2.4 TRADITIONAL MODELLING PROCEDURE 30 2.5.1 Response Surface Methodology (RSM) 31 2.5.1.1 Theory and steps for RSM application 31 2.5.1.2 Factorial experimental design 31 2.5.1.3 Choice of experimental design 31 2.5.1.4 Evaluation of fitted model 33 2.5.2 Artificial Neural Network (ANN) 34 2.5.2.1 Theoretical background of the ANN model 34 2.5.2.2 Applicat	2.2.3.1	Sources of Copper in the Aquatic Environment	13
2.2.3.3 Environmental regulations and standards for discharge of effluent into water .15 .15 2.2.3.4 Legislative Limits for Copper in the Aquatic Environment .17 2.2.3.5 Conventional wastewater treatment approaches for copper removal .18 2.3 ADSORPTION PROCESS .21 2.3.1 Mechanism of adsorption .21 2.3.2 Adsorption isotherm .23 2.3.2.1 Langmuir Isotherm .23 2.3.2.2 Freundlich Isotherm .23 2.3.3 Factors affecting the rate of adsorption .25 2.4 TRADITIONAL MODELLING PROCEDURE .30 2.5.1 Response Surface Methodology (RSM) .31 2.5.1.1 Theory and steps for RSM application .31 2.5.1.2 Factorial experimental design .31 2.5.1.3 Choice of experimental design .31 2.5.1.4 Evaluation of fitted model .33 2.5.2 Artificial Neural Network (ANN) .34 2.5.2.1 Theoretical background of the ANN model .34 2.5.2.2 Application of ANN in process optimisation and modelling of adsorption proces	2.2.3.2	Toxicity of copper to humans and the environment	14
water152.2.3.4Legislative Limits for Copper in the Aquatic Environment172.2.3.5Conventional wastewater treatment approaches for copper removal182.3ADSORPTION PROCESS212.3.1Mechanism of adsorption212.3.2Adsorption isotherm232.3.2.1Langmuir Isotherm232.3.2.2Freundlich Isotherm242.3.3Factors affecting the rate of adsorption252.4TRADITIONAL MODELLING PROCEDURE302.5MATHEMATICAL MODELLING OF THE ADSORPTION PROCESS302.5.1Response Surface Methodology (RSM)312.5.1.1Theory and steps for RSM application312.5.1.2Factorial experimental design312.5.1.3Choice of experimental design312.5.1.4Evaluation of fitted model332.5.2Artificial Neural Network (ANN)342.5.2.1Theoretical background of the ANN model342.5.2.2Application of ANN in process optimisation and modelling of adsorption352.5.3Application of RSM and ANN in wastewater treatment plant process optimisation36	2.2.3.3	Environmental regulations and standards for discharge of effluent into	
2.2.3.4 Legislative Limits for Copper in the Aquatic Environment 17 2.3.5 Conventional wastewater treatment approaches for copper removal 18 2.3 ADSORPTION PROCESS 21 2.3.1 Mechanism of adsorption 21 2.3.2 Adsorption isotherm 23 2.3.2.1 Langmuir Isotherm 23 2.3.2.2 Freundlich Isotherm 23 2.3.3 Factors affecting the rate of adsorption 25 2.4 TRADITIONAL MODELLING PROCEDURE 30 2.5.1 Response Surface Methodology (RSM) 31 2.5.1.1 Theory and steps for RSM application 31 2.5.1.2 Factorial experimental design 31 2.5.1.3 Choice of experimental design 31 2.5.1.4 Evaluation of fitted model 33 2.5.2 Artificial Neural Network (ANN) 34 2.5.2.1 Theoretical background of the ANN model 34 2.5.2.2 Application of ANN in process optimisation and modelling of adsorption processes 35		water	15
2.2.3.5 Conventional wastewater treatment approaches for copper removal 18 2.3 ADSORPTION PROCESS 21 2.3.1 Mechanism of adsorption 21 2.3.2 Adsorption isotherm 23 2.3.2.1 Langmuir Isotherm 23 2.3.2.2 Freundlich Isotherm 23 2.3.3 Factors affecting the rate of adsorption 25 2.4 TRADITIONAL MODELLING PROCEDURE 30 2.5 MATHEMATICAL MODELLING OF THE ADSORPTION PROCESS 30 2.5.1 Response Surface Methodology (RSM) 31 2.5.1.2 Factorial experimental design 31 2.5.1.3 Choice of experimental design 31 2.5.1.4 Evaluation of fitted model 33 2.5.2 Artificial Neural Network (ANN) 34 2.5.2.1 Theoretical background of the ANN model 34 2.5.2.2 Application of ANN in process optimisation and modelling of adsorption processes 35 2.5.3 Application of ANN in wastewater treatment plant process optimisation 36	2.2.3.4	Legislative Limits for Copper in the Aquatic Environment	17
2.3ADSORPTION PROCESS.212.3.1Mechanism of adsorption212.3.2Adsorption isotherm.232.3.2.1Langmuir Isotherm.232.3.2.2Freundlich Isotherm.232.3.2.3Factors affecting the rate of adsorption252.4TRADITIONAL MODELLING PROCEDURE.302.5MATHEMATICAL MODELLING OF THE ADSORPTION PROCESS302.5.1Response Surface Methodology (RSM)312.5.1.2Factorial experimental design312.5.1.3Choice of experimental design312.5.1.4Evaluation of fitted model332.5.2Artificial Neural Network (ANN)342.5.2Application of ANN in process optimisation and modelling of adsorption342.5.2Application of ANN in wastewater treatment plant process optimisation36	2.2.3.5	Conventional wastewater treatment approaches for copper removal	18
2.3.1Mechanism of adsorption212.3.2Adsorption isotherm232.3.2.1Langmuir Isotherm232.3.2.2Freundlich Isotherm242.3.3Factors affecting the rate of adsorption252.4TRADITIONAL MODELLING PROCEDURE302.5MATHEMATICAL MODELLING OF THE ADSORPTION PROCESS302.5.1Response Surface Methodology (RSM)312.5.1.1Theory and steps for RSM application312.5.1.2Factorial experimental design312.5.1.3Choice of experimental design312.5.1.4Evaluation of fitted model332.5.2Artificial Neural Network (ANN)342.5.2.1Theoretical background of the ANN model342.5.2.2Application of ANN in process optimisation and modelling of adsorption processes352.5.3Application of RSM and ANN in wastewater treatment plant process optimisation36	2.3 AI	DSORPTION PROCESS	21
2.3.2Adsorption isotherm	2.3.1	Mechanism of adsorption	21
2.3.2.1Langmuir Isotherm	2.3.2	Adsorption isotherm	23
2.3.2.2Freundlich Isotherm.242.3.3Factors affecting the rate of adsorption252.4TRADITIONAL MODELLING PROCEDURE.302.5MATHEMATICAL MODELLING OF THE ADSORPTION PROCESS302.5.1Response Surface Methodology (RSM)312.5.1.1Theory and steps for RSM application312.5.1.2Factorial experimental design312.5.1.3Choice of experimental design312.5.1.4Evaluation of fitted model332.5.2Artificial Neural Network (ANN)342.5.2.1Theoretical background of the ANN model342.5.2.2Application of ANN in process optimisation and modelling of adsorption processes352.5.3Application of RSM and ANN in wastewater treatment plant process optimisation36	2.3.2.1	Langmuir Isotherm	23
2.3.3Factors affecting the rate of adsorption252.4TRADITIONAL MODELLING PROCEDURE302.5MATHEMATICAL MODELLING OF THE ADSORPTION PROCESS302.5.1Response Surface Methodology (RSM)312.5.1.1Theory and steps for RSM application312.5.1.2Factorial experimental design312.5.1.3Choice of experimental design312.5.1.4Evaluation of fitted model332.5.2Artificial Neural Network (ANN)342.5.2.1Theoretical background of the ANN model342.5.2.2Application of ANN in process optimisation and modelling of adsorption processes352.5.3Application of RSM and ANN in wastewater treatment plant process optimisation36	2.3.2.2	Freundlich Isotherm	24
2.4 TRADITIONAL MODELLING PROCEDURE 30 2.5 MATHEMATICAL MODELLING OF THE ADSORPTION PROCESS 30 2.5.1 Response Surface Methodology (RSM) 31 2.5.1.1 Theory and steps for RSM application 31 2.5.1.2 Factorial experimental design 31 2.5.1.3 Choice of experimental design 31 2.5.1.4 Evaluation of fitted model 33 2.5.1.5 Optimisation of adsorption process using RSM 33 2.5.2 Artificial Neural Network (ANN) 34 2.5.2.1 Theoretical background of the ANN model 34 2.5.2.2 Application of ANN in process optimisation and modelling of adsorption processes 35 2.5.2.3 Application of RSM and ANN in wastewater treatment plant process optimisation 36	2.3.3	Factors affecting the rate of adsorption	25
2.5MATHEMATICAL MODELLING OF THE ADSORPTION PROCESS302.5.1Response Surface Methodology (RSM)312.5.1.1Theory and steps for RSM application312.5.1.2Factorial experimental design312.5.1.3Choice of experimental design312.5.1.4Evaluation of fitted model332.5.1.5Optimisation of adsorption process using RSM332.5.2Artificial Neural Network (ANN)342.5.2.1Theoretical background of the ANN model342.5.2.2Application of ANN in process optimisation and modelling of adsorption processes352.5.3Application of RSM and ANN in wastewater treatment plant process optimisation36	2.4 TF	RADITIONAL MODELLING PROCEDURE	30
2.5.1Response Surface Methodology (RSM)312.5.1.1Theory and steps for RSM application312.5.1.2Factorial experimental design312.5.1.3Choice of experimental design312.5.1.4Evaluation of fitted model332.5.1.5Optimisation of adsorption process using RSM332.5.2Artificial Neural Network (ANN)342.5.2.1Theoretical background of the ANN model342.5.2.2Application of ANN in process optimisation and modelling of adsorption processes352.5.3Application of RSM and ANN in wastewater treatment plant process optimisation36	2.5 M	ATHEMATICAL MODELLING OF THE ADSORPTION PROCESS	30
2.5.1.1Theory and steps for RSM application312.5.1.2Factorial experimental design312.5.1.3Choice of experimental design312.5.1.4Evaluation of fitted model332.5.1.5Optimisation of adsorption process using RSM332.5.2Artificial Neural Network (ANN)342.5.2.1Theoretical background of the ANN model342.5.2.2Application of ANN in process optimisation and modelling of adsorption processes352.5.3Application of RSM and ANN in wastewater treatment plant process optimisation36	2.5.1	Response Surface Methodology (RSM)	31
2.5.1.2Factorial experimental design312.5.1.3Choice of experimental design312.5.1.4Evaluation of fitted model332.5.1.5Optimisation of adsorption process using RSM332.5.2Artificial Neural Network (ANN)342.5.2.1Theoretical background of the ANN model342.5.2.2Application of ANN in process optimisation and modelling of adsorption processes352.5.3Application of RSM and ANN in wastewater treatment plant process optimisation36	2.5.1.1	Theory and steps for RSM application	31
2.5.1.3Choice of experimental design	2.5.1.2	Factorial experimental design	31
2.5.1.4Evaluation of fitted model332.5.1.5Optimisation of adsorption process using RSM332.5.2Artificial Neural Network (ANN)342.5.2.1Theoretical background of the ANN model342.5.2.2Application of ANN in process optimisation and modelling of adsorption processes352.5.3Application of RSM and ANN in wastewater treatment plant process optimisation36	2.5.1.3	Choice of experimental design	31
 2.5.1.5 Optimisation of adsorption process using RSM	2.5.1.4	Evaluation of fitted model	33
 2.5.2 Artificial Neural Network (ANN)	2.5.1.5	Optimisation of adsorption process using RSM	33
 2.5.2.1 Theoretical background of the ANN model	2.5.2	Artificial Neural Network (ANN)	34
 2.5.2.2 Application of ANN in process optimisation and modelling of adsorption processes	2.5.2.1	Theoretical background of the ANN model	34
2.5.3 Application of RSM and ANN in wastewater treatment plant process optimisation	2.5.2.2	Application of ANN in process optimisation and modelling of adsorption processes	35
	2.5.3	Application of RSM and ANN in wastewater treatment plant process optimisation	36

2.5.4	Comparison of RSM and ANN models	37
2.5.5	RSM and ANN advantages and limitations	41
2.6 C	ELLULOSE AS A SOURCE MATERIAL	42
2.6.1	Structure and properties of cellulose	42
2.6.2	Surface morphology	43
2.6.3	Sources of cellulose	44
2.7 C	ELLULOSE NANOWHISKERS AS POTENTIAL ADSORBENT (AS AN EXAMPLE CASE)	45
2.7.1	Structure and properties of CNws	47
2.7.2	Modification of cellulose nanowhiskers	48
2.8 C	HARACTERISATION OF CNW ADSORBENTS	49
2.8.1	Fourier Transform Infrared Spectroscopy (FTIR)	49
2.8.2	Zeta potential	50
2.8.3	Electron microscopy (SEM and TEM)	51
2.8.4	Brunauer-Emmett-Teller (BET) method	51
2.9 EI	NVIRONMENTAL ANALYSIS	52
2.9.1	THEORETICAL BACKGROUND TO INSTRUMENTATION	53
2.9.1.1	Atomic absorption spectrometry instrumentation (AAS)	53
2.9.1.2	Inductively Coupled Plasma Mass Spectrometry (ICP-MS)	54
2.9.2	Performance comparison of instrumentation techniques for analysis	55
2.10 IS	SUES ARISING FROM LITERATURE REVIEW	59
СНАРТЕ	R 3: MATERIALS AND METHODS	61
3.1 IN	ITRODUCTION	61
3.1 IN 3.2 №	ITRODUCTION	61 61
3.1 IN3.2 Ⅳ3.2.1	ITRODUCTION IATERIALS Reagents	61 61
 3.1 IN 3.2 N 3.2.1 3.2.2 	ITRODUCTION IATERIALS Reagents Cotton	61 61 61
 3.1 IN 3.2 N 3.2.1 3.2.2 3.2.3 	ITRODUCTION IATERIALS Reagents Cotton Washing procedure	61 61 61 61
 3.1 IN 3.2 N 3.2.1 3.2.2 3.2.3 3.2.4 	ITRODUCTION IATERIALS Reagents Cotton Washing procedure Wastewater samples	61 61 61 64 64
 3.1 IN 3.2 IN 3.2.1 3.2.2 3.2.3 3.2.4 3.2.4.1 	ITRODUCTION IATERIALS Reagents Cotton Washing procedure Wastewater samples Stability of effluent	61 61 61 64 64 65
3.1 IN 3.2 N 3.2.1 3.2.2 3.2.3 3.2.4 3.2.4.1 3.2.4.2	ITRODUCTION IATERIALS Reagents Cotton Washing procedure Wastewater samples. Stability of effluent Wastewater characteristics	61 61 61 64 64 65 65
3.1 IN 3.2 N 3.2.1 3.2.2 3.2.3 3.2.4 3.2.4.1 3.2.4.1 3.2.4.2 3.2.4.2	ITRODUCTION IATERIALS Reagents Cotton Washing procedure Wastewater samples. Stability of effluent Wastewater characteristics Total suspended solids (TSS).	61 61 64 64 65 65 65
3.1 IN 3.2 N 3.2.1 3.2.2 3.2.3 3.2.4 3.2.4.1 3.2.4.2 3.2.4.2 3.2.4.3 3.2.4.3 3.2.4.4	ITRODUCTION IATERIALS Reagents Cotton Washing procedure Wastewater samples Stability of effluent Wastewater characteristics Total suspended solids (TSS) pH, conductivity and total dissolved solids (TDS)	61 61 61 64 65 65 65 65
3.1 IN 3.2 N 3.2.2 3.2.3 3.2.4 3.2.4.1 3.2.4.2 3.2.4.2 3.2.4.3 3.2.4.3 3.2.4.3 3.2.4.5	ITRODUCTION IATERIALS Reagents Cotton. Washing procedure Wastewater samples. Stability of effluent Wastewater characteristics Total suspended solids (TSS) pH, conductivity and total dissolved solids (TDS) Chemical oxygen demand (COD)	61 61 64 64 65 65 65 65 66
3.1 IN 3.2.1 N 3.2.2 .2.3 3.2.4 .2.4.1 3.2.4.2 .2.4.2 3.2.4.3 .2.4.3 3.2.4.4 .2.4.5 3.2.4.5 .2.4.5	ITRODUCTION IATERIALS Reagents Cotton Washing procedure Wastewater samples Stability of effluent Wastewater characteristics Total suspended solids (TSS) pH, conductivity and total dissolved solids (TDS) Chemical oxygen demand (COD)	61 61 61 64 65 65 65 65 66 66 66
 3.1 IN 3.2 N 3.2.1 3.2.2 3.2.3 3.2.4 3.2.4.1 3.2.4.2 3.2.4.3 3.2.4.4 3.2.4.5 3.3 EX 3.3.1 	ITRODUCTION IATERIALS Reagents Cotton Washing procedure Wastewater samples Stability of effluent Wastewater characteristics Total suspended solids (TSS) pH, conductivity and total dissolved solids (TDS) pH, conductivity and total dissolved solids (TDS) Chemical oxygen demand (COD) KPERIMENTAL METHODOLOGY FOR ADSORBENT PREPARATION Preparation of cellulose nanowhiskers	61 61 64 64 65 65 65 66 66 67
3.1 IN 3.2 N 3.2.1 N 3.2.2 3.2.3 3.2.4 3.2.4.1 3.2.4.2 3.2.4.2 3.2.4.3 3.2.4.3 3.2.4.5 3.2.4.5 3.3.1 3.3.1 3.3.2 3.3.2	ITRODUCTION IATERIALS	61 61 64 65 65 65 65 66 67 67 67
 3.1 IN 3.2 N 3.2.1 3.2.2 3.2.3 3.2.4 3.2.4.1 3.2.4.2 3.2.4.3 3.2.4.4 3.2.4.5 3.3 EX 3.3.1 3.3.2 3.4 CI 	ITRODUCTION IATERIALS Reagents Cotton Washing procedure Wastewater samples Stability of effluent Wastewater characteristics Total suspended solids (TSS) pH, conductivity and total dissolved solids (TDS) pH, conductivity and total dissolved solids (TDS) Chemical oxygen demand (COD) Chemical oxygen demand (COD) Preparation of cellulose nanowhiskers Preparation of surface-carboxylated cnws (oxidised CNWs) via tempo-mediated oxidation. HARACTERISATION METHODS OF CELLULOSE NANOWHISKERS	61 61 64 65 65 65 65 66 66 67 67 67 68 69
 3.1 IN 3.2 N 3.2.1 3.2.2 3.2.3 3.2.4 3.2.4.1 3.2.4.2 3.2.4.3 3.2.4.4 3.2.4.5 3.3 EX 3.3.1 3.3.2 3.4 CI 3.4.1 	ITRODUCTION IATERIALS Reagents Cotton Washing procedure Wastewater samples Stability of effluent Wastewater characteristics Total suspended solids (TSS) pH, conductivity and total dissolved solids (TDS) pH, conductivity and total dissolved solids (TDS) Chemical oxygen demand (COD) Chemical oxygen demand (COD) RPERIMENTAL METHODOLOGY FOR ADSORBENT PREPARATION Preparation of cellulose nanowhiskers Preparation of surface-carboxylated cnws (oxidised CNWs) via tempo-mediated oxidation. HARACTERISATION METHODS OF CELLULOSE NANOWHISKERS Fourier Transform Infra-Red spectroscopy.	61 61 64 65 65 65 65 66 66 67 67 67 67 67 67 67 67
 3.1 IN 3.2 IN 3.2.1 3.2.2 3.2.3 3.2.4 3.2.4.1 3.2.4.2 3.2.4.3 3.2.4.4 3.2.4.5 3.3.1 3.3.2 3.4.1 3.4.1 3.4.2 	ITRODUCTION IATERIALS Reagents Cotton Washing procedure Wastewater samples Stability of effluent Wastewater characteristics Total suspended solids (TSS) pH, conductivity and total dissolved solids (TDS) Chemical oxygen demand (COD) Chemical oxygen demand (COD) Chemical oxygen demand (COD) Preparation of cellulose nanowhiskers Preparation of cellulose nanowhiskers Preparation of surface-carboxylated cnws (oxidised CNWs) via tempo-mediated oxidation	61 61 61 64 65 65 65 65 66 66 67 67 67 67 67 67 67 67 67 67 67 67
3.1 IN 3.2 N 3.2.1 3.2.2 3.2.3 3.2.4 3.2.4.1 3.2.4.2 3.2.4.3 3.2.4.5 3.2.4.5 3.3 3.2.4.5 3.3.1 3.3.2 3.3.2 3.4.1 3.4.2 3.4.2 3.4.3	ITRODUCTION IATERIALS Reagents Cotton Washing procedure Wastewater samples Stability of effluent Wastewater characteristics Total suspended solids (TSS) pH, conductivity and total dissolved solids (TDS) pH, conductivity and total dissolved solids (TDS) Chemical oxygen demand (COD) KPERIMENTAL METHODOLOGY FOR ADSORBENT PREPARATION Preparation of cellulose nanowhiskers Preparation of surface-carboxylated cnws (oxidised CNWs) via tempo-mediated oxidation HARACTERISATION METHODS OF CELLULOSE NANOWHISKERS Fourier Transform Infra-Red spectroscopy Zeta potential. Determination of carboxylate contents	61 61 64 64 65 65 65 65 66 67 67 67 67 67 67 67 67 67 67 67 67
 3.1 IN 3.2 IN 3.2.1 3.2.2 3.2.3 3.2.4 3.2.4.2 3.2.4.3 3.2.4.4 3.2.4.5 3.3 IN 3.3.1 3.3.2 3.4 CIN 3.4.1 3.4.2 3.4.3 3.4.4 	ITRODUCTION INTERIALS. Reagents Cotton. Washing procedure Wastewater samples. Stability of effluent Wastewater characteristics Total suspended solids (TSS) pH, conductivity and total dissolved solids (TDS) pH, conductivity and total dissolved solids (TDS) Chemical oxygen demand (COD) Chemical oxygen demand (COD) Preparation of cellulose nanowhiskers Preparation of cellulose nanowhiskers Preparation of surface-carboxylated cnws (oxidised CNWs) via tempo-mediated oxidation. HARACTERISATION METHODS OF CELLULOSE NANOWHISKERS Fourier Transform Infra-Red spectroscopy. Zeta potential. Determination of carboxylate contents BET method	61 61 61 64 65 65 65 65 66 67 67 67 67 67 67 67 67 67 67 67 67 67 67
3.1 IN 3.2.1 N 3.2.2 3.2.3 3.2.4 3.2.4.1 3.2.4.2 3.2.4.2 3.2.4.3 3.2.4.5 3.2.4.5 B.3 3.3.1 3.3.2 3.4.1 3.4.2 3.4.1 3.4.2 3.4.3 3.4.4 3.4.4 3.4.5	ITRODUCTION IATERIALS Reagents Cotton Washing procedure Wastewater samples Stability of effluent Wastewater characteristics Total suspended solids (TSS) pH, conductivity and total dissolved solids (TDS) pH, conductivity and total dissolved solids (TDS) Chemical oxygen demand (COD) CPERIMENTAL METHODOLOGY FOR ADSORBENT PREPARATION Preparation of cellulose nanowhiskers Preparation of surface-carboxylated cnws (oxidised CNWs) via tempo-mediated oxidation. HARACTERISATION METHODS OF CELLULOSE NANOWHISKERS Fourier Transform Infra-Red spectroscopy. Zeta potential. Determination of carboxylate contents BET method Electron microscopy (SEM).	61 61 64 64 65 65 65 65 66 67 67 67 67 67 67 67 67 67 67 67 67 67 67
3.1 IN 3.2 N 3.2.1 3.2.2 3.2.3 3.2.4 3.2.4.1 3.2.4.2 3.2.4.3 3.2.4.4 3.2.4.5 3.3 3.2.4.3 3.2.4.4 3.2.4.5 3.3 3.3.1 3.3.2 3.4.1 3.4.2 3.4.2 3.4.3 3.4.3 3.4.4 3.4.5 3.4.6	ITRODUCTION IATERIALS. Reagents Cotton. Washing procedure Wastewater samples. Stability of effluent Wastewater characteristics Total suspended solids (TSS) pH, conductivity and total dissolved solids (TDS) Chemical oxygen demand (COD) Chemical oxygen demand (COD) Chemical oxygen demand (COD) Chemical oxygen demand (COD) Preparation of cellulose nanowhiskers Preparation of cellulose nanowhiskers Preparation of surface-carboxylated cnws (oxidised CNWs) via tempo-mediated oxidation. HARACTERISATION METHODS OF CELLULOSE NANOWHISKERS Fourier Transform Infra-Red spectroscopy. Zeta potential. Determination of carboxylate contents BET method. Electron microscopy (SEM) Electron microscopy (TEM).	61 61 61 64 65 65 65 65 66 66 67 67 67 67 67 67 67 67 67 67 67 70 70 71

3.4.7 Conductometric titration	2
3.5 EXPERIMENTAL SET UP FOR BATCH ADSORPTION STUDIES	2
3.5.1 Preparation of adsorbate solution	3
3.5.1.1 Clean water matrix	3
3.5.1.2 Wastewater effluent spiked with Cu(II)	3
3.5.2 Batch adsorption studies using clean water matrix	4
3.5.3 Batch adsorption studies using wastewater effluent	4
3.5.4 Benchmark study7	5
3.5.5 Selection of independent variables and ranges through scoping experiments	5
3.5.5.1 Effect of pH7!	5
3.5.5.2Effect of contact time	6
3.5.5.3 Effect of temperature	6
3.5.5.4 Effect of sorbent dosage	6
3.5.6 Adsorption isotherm studies	7
3.5.7 Adsorption kinetics studies	7
3.6 INSTRUMENTAL ANALYSIS OF COPPER	8
3.6.1 Atomic absorption spectrometry (AAS)	8
3.6.2 Preparation of AAS standard solutions	9
3.6.3 Determination of detection limits and quantification limit79	9
3.7 DESIGN OF EXPERIMENTS	0
3.7.1 Central composite design (CCD)	1
3.7.2 Application of ccd for adsorption from the clean water matrix	1
3.7.3 Application of ccd for adsorption from the wastewater effluent8.	2
3.8 MATHEMATICAL MODELLING OF Cu(II) REMOVAL FROM THE WATER MATRICES	2
3.8.1 Response surface methodology (RSM)8.	2
3.8.2 Artificial Neural Network (ANN)	3
3.9 DATA ANALYSIS AND STATISTICAL TECHNIQUES	4
3.9.1 Evaluated of the models	4
3.9.2 Test and validation of the model	5
3.10 EXPERIMENTAL SET UP FOR FIXED BED ADSORPTION	7
CHAPTER 4: CHARACTERISATION OF CELLULOSE NANOWHISKERS	9
4.1 INTRODUCTION	9
4.2 HIGHLIGHTS	9
4.3 RESULTS AND DISCUSSION	0
4.3.1 Fourier Transform Infra-Red (FTIR) spectroscopy	0
4.3.2 Zeta potential determination	2
4.3.3 BET analysis	3
4.3.4 TEM analysis	4
4.3.5 Determination of carboxylate contents	6
4.3.6 Stability test	9
CHAPTER 5: PREDICTING THE CAPABILITY OF OXIDISED CNW ADSORBENTS FOR THE	
REMEDIATING OF COPPER FROM WATER USING RSM AND ANN MODELS	1
5.1 INTRODUCTION	1

5.2	HIGHLIGHTS
5.3	RESULTS AND DISCUSSION
5.3.1	Control experiments for the effect of filter102
5.3.2	Control experiments to determine parameter ranges for the central composite design (CCD)
5.3.2	.1 Initial Cu(II) concentration
5.3.2	.2 Contact time of the adsorption process
5.3.2	.3 Cu(II) solution pH in the adsorption process
5.3.2	.4 Determination of sorbent dosage (Oxidised CNW adsorbents)
5.3.2	.5 Temperature of the adsorption medium
5.4	MODELLING OF THE ADSORPTION PROCESS
5.4.1	Central composite design (CCD)
5.4.2	RSM model
5.4.2	1 Thermodynamic study116
5.4.3	ANN model
5.5	STATISTICAL COMPARISON AND PERFORMANCE OF RSM AND ANN MODELS122
5.6	ADSORPTION ISOTHERM
5.7	ADSORPTION KINETICS
СНАР	TER 6: UNDERSTAND THE WASTEWATER MATRIX AND CHARACTERISATION134
6.1	INTRODUCTION
6.2	HIGHLIGHTS
6.3	RESULTS AND DISCUSSION
6.3.1	Sources of contamination of icp-ms instrumental analysis
6.3.2	The influence of storage time141
6.3.3	Physicochemical data analysis of water samples142
6.3.4	Chemical Oxygen Demand (COD)146
6.3.5	Heavy metal concentration and its variation in influent and effluent wastewater samples
6.3.6	benchmarking the wastewater matrix from the pollutant and adsorbent perspective . 151
СНАР	PTER 7: PREDICTING THE CAPABILITY OF OXIDISED CNW ADSORBENTS FOR THE
REM	EDIATING OF COPPER FROM WASTEWATER EFFLUENT USING RSM AND ANN MODELS .154
7.1	INTRODUCTION
7.2	HIGHLIGHTS
7.3	SCOPING EXPERIMENTS FOR EVALUATING REMOVAL PERFORMANCE IN THE
74	MATHEMATICAL MODELLING OF SPIKED CU(III) REMOVAL FROM WASTEWATER
/	EFFLUENT
7.4.1	Central composite design (CCD)
7.4.2	Response surface methodology (RSM)
7.4.3	Artificial neural network (ANN)
7.5	MULTIPLE LINEAR REGRESSION (MLR)
7.6	STATISTICAL COMPARISON AND PERFORMANCE OF MODELS FOR WASTEWATER
	EFFLUENT

7.6.1	Model validation using unseen experiments	.177
7.7 P	ROCESS OPTIMISATION AND OPTIMUM PARAMETERS	.181
7.7.1	Performance of continuous flow experiment under optimal operating conditions	.184
СНАРТІ	ER 8: CONCLUSIONS AND RECOMMENDATION FOR FUTURE RESEARCH	.186
8.1 C	ONCLUSIONS	.186
8.1.1	Characterisation of CNW adsorbents	.186
8.1.2	Predicting the capability of oxidised CNW adsorbents for the remediation of copper f	rom
	clean water using RSM and ANN models	. 188
8.1.3	Understanding the wastewater matrix and characterisation	. 188
8.1.4	Predicting the capability of oxidised CNW adsorbents for remediation of spiked coppe	er
	from wastewater effluent using RSM and ANN models	. 189
8.2 R	ECOMMENDATION FOR FUTURE RESEARCH	.190
8.2.1	Manufacturing oxidised CNW adsorbents	.190
8.2.2	Suitability of adsorbents for continuous flow column application	. 190
8.2.3	Regeneration of adsorbent	.191
8.2.4	Recovery of heavy metals from the adsorbent	.191
8.2.5	Preparation of dual and multi-functional groups on the surface of CNWs	. 192
REFERE	NCES	.192
APPEND	DIX	.215

LIST OF FIGURES

Figure 2.1: Sources of water pollution in the urban water cycle
Figure 2.2: Schematic of a typical wastewater treatment plant
Figure 2.3: Mechanism of adsorption process
Figure 2.4: Theoretical solubility of heavy metals
Figure 2.5: Experimental design based on the study for the optimisation of three
variables (a) Box-Behnken design and (b) central composite design.
Figure 2.6: Architecture of the developed artificial neural network (ANN)35
Figure 2.7: Molecular structure of cellulose43
Figure 2.8: Schematic of amorphous and crystalline cellulose45
Figure 2.9: Acid hydrolysis mechanism
Figure 2.10: The schematic of hierarchical structure of a wood47
Figure 2.11: Schematic illustrating Smoluchowski and Hückel's approximations
used to convert from electrophoretic mobility into zeta potential50
Figure 2.12: AAS schematic diagram
Figure 2.12: AAS schematic diagram 54 Figure 2.13: ICP-MS block diagram 55 Figure 2.14: Typical detection limit ranges for the major atomic spectroscopy techniques 56 Figure 3.1: Process flow for CNWs production 68 Figure 3.2: The process flow for production of oxidised CNWs via TEMPO-mediated oxidation 69 Figure 3.3: Design of experiment (Define significant factors and response) 73
Figure 2.12: AAS schematic diagram 54 Figure 2.13: ICP-MS block diagram 55 Figure 2.14: Typical detection limit ranges for the major atomic spectroscopy techniques 56 Figure 3.1: Process flow for CNWs production 68 Figure 3.2: The process flow for production of oxidised CNWs via TEMPO-mediated oxidation 69 Figure 3.3: Design of experiment (Define significant factors and response) 73 Figure 3.4: Design of experiment (selection of experimental design) 80
Figure 2.12: AAS schematic diagram
Figure 2.12: AAS schematic diagram 54 Figure 2.13: ICP-MS block diagram 55 Figure 2.14: Typical detection limit ranges for the major atomic spectroscopy techniques 56 Figure 3.1: Process flow for CNWs production 68 Figure 3.2: The process flow for production of oxidised CNWs via TEMPO-mediated oxidation 69 Figure 3.3: Design of experiment (Define significant factors and response) 73 Figure 3.4: Design of experiment (selection of experimental design) 80 Figure 3.5: Design of experiment (verification of the model) 84 Figure 4.1: The FTIR spectra of CNWs 90
Figure 2.12: AAS schematic diagram
Figure 2.12: AAS schematic diagram 54 Figure 2.13: ICP-MS block diagram 55 Figure 2.14: Typical detection limit ranges for the major atomic spectroscopy techniques 56 Figure 3.1: Process flow for CNWs production 68 Figure 3.2: The process flow for production of oxidised CNWs via TEMPO-mediated oxidation 69 Figure 3.3: Design of experiment (Define significant factors and response) 73 Figure 3.4: Design of experiment (selection of experimental design) 80 Figure 3.5: Design of experiment (verification of the model) 84 Figure 4.1: The FTIR spectra of CNWs 90 Figure 4.2: FTIR spectra of the CNWs and oxidised CNWs 92 Figure 4.3: A typical TEM image of negatively stained preparations of

Figure 4.4: A typical image of negatively stained preparations of oxidised CNW
adsorbents95
Figure 4.5: Typical conductometric titration curve
Figure 4.6: Determination of the total number of carboxyl groups in oxidised CNW suspension by conductometric titration
Figure 5.1: Variations in the removal efficiency of Cu(II) with contact time for different sorbent dosages: Initial Cu(II) concentration = 60 mg/L; T = 25 °C; pH = 6
Figure 5.2: Variations in the removal efficiency of Cu(II) with contact time for different initial concentrations of Cu(II) solution: sorbent dosage = 10 g/L; T = 25 °C; pH = 6.0106
Figure 5.3: Variations in Cu(II) removal efficiency according to sorbent dosage (Oxidised CNW): initial Cu(II) concentration; C _i =10 mg/L; T=25 °C; pH=6109
Figure 5.4: The experimentally obtained Cu(II) removal compared to that predicted by the response surface methodology (RSM)
Figure 5.5: Source distribution in the removal percentage (adjusted sum of squares) versus source
Figure 5.6: The experimentally obtained result for Cu(II) removal compared to that predicted by the artificial neural network (ANN)118
Figure 5.7: Surface plots (left) and corresponding contour plots (right) showing the effects of adsorption parameters on the Cu(II) removal as predicted by the ANN model, with temperature held constant at 15.5°C (A), initial Cu(II) concentration held constant 35 mg/L (B) and sorbent dosage held constant at 5.10 g/L (C)120
Figure 5.8: Plot representing the residuals obtained from the predictions of RSM and ANN modelling
Figure 5.9: Plot representing error distribution obtained from the predictions of RSM and ANN modelling

Figure 5.10: 3D scatter plots showing a comparison of the CCD with unseen
experiments within the systems (A) for this work; (B) for Ghosh et
al., (2015); (C) Bingol et al. (2012)
Figure 5.11: Comparison of the experimental and predicted results for unseen
experiments between the RSM and ANN models
Figure 5.12: Linear plot, first-order rate equation, using oxidised CNW
adsorbents (sorbent dosage = 4 g/L, temperature = 25 °C , pH = 6)
Figure 5.13: Linear plot, second-order rate equation, using oxidised CNW
adsorbents (sorbent dosage = 4 g/L, temperature = 25 °C , pH = 6).
Figure 6.1: Variation in temperature for sampled wastewater influent and effluent
(February to March 2016) from Stoke Bardolph WWTP143
Figure 6.2: Variation in pH for influent and effluent (February to March 2016)
Figure 6.3: Variation in conductivity for influent and effluent (February to March
2016)
Figure 6.4: Variation in COD for influent and effluent (February to March 2016).
Figure 6.5: The concentrations of Cu in influent and effluent
Figure 6.6: The concentrations of Cr in influent and effluent
Figure 6.7: The concentrations of Cd in influent and effluent
Figure 6.8: The concentrations of Fe in influent and effluent 150
Figure 6.9: The concentrations of Pb in influent and effluent
Figure 7.1. The experimentally obtained remediation of spilled (w(II) from
Figure 7.1. The experimentally obtained remediation of spiked Cu(II) from
surface methodology (RSM) 160
Eigure 7.2: Source distribution in removal percentage (adjusted sum of
rigure 7.2. Source distribution in removal percentage (adjusted sum of squares vs
50urce)

Figure 7.3: Comparison between the Cu(II) removal predicted by the artificial
neural network (ANN) model and the experimentally determined
removal; (WW) wastewater166
Figure 7.4: Surface plots (left) and corresponding contour plots (right) showing
the effects of adsorption parameters on the spiked Cu (II) removal
from wastewater effluent as predicted by the ANN model with initial
concentration of wastewater effluent spiked with Cu(II) held constant
3 mg/L (A), pH held constant at 6.5 (B) and sorbent dosage held
constant at 5.25 g/L (C)169
Figure 7.5: Comparison of the actual and predicted values by MLR and ANN
models (including WW variability)
Figure 7.6: Comparison of the actual and predicted values by MLR and ANN
models
Figure 7.7: Comparison of the actual and predicted values by MLR and ANN
models (including WW variability)176
Figure 7.8: 3D scatter plots showing: (A) comparison of the CCD with unseen
experiments within the systems for this work; (B) solid-phase
extraction for determination of gold from industrial wastewater
(Ebrahimzadeh et al., 2012); (C) Lead adsorption from industrial
sludge leachate (Geyikci et al., 2012)179
Figure 7.9: Effect of pressure on Cu(II) removal from wastewater effluent184
Figure A.1: Calibration Curve of Cu(II) from standard solutions of Cu (2-10
mg/L)216
Figure A.2: Calibration Curve of Cu(II) from standard solutions of Cu (10-70
mg/L)217
Figure B.1: BET surface area plot of (a) unmodified and (b) modified CNWs .218
Figure B.2: N ₂ adsorption-desorption isotherm at 77K of (a) unmodified and (b)
modified CNWs
Figure C.1: Langmuir isotherm plots for the adsorption of Cu(II) onto modified
CNW adsorbents (temperature=; pH=; contact time=; sorbent dosage)

Figure C.2: Freundlich isotherm plots for the adsorption of Cu(II) onto modified
CNW adsorbents (temperature=; pH=; contact time=; sorbent dosage)
Figure D.1: Experimental and ANN predicted Cu(II) removal for the training set.
Condition for ANN: training algorithm: Levenberg-Marquardt,
number of inputs: 4, number of hidden layer: 6, number of output: 1

LIST OF TABLES

Table 2.1: The sources of anthropogenic pollutants
Table 2.2: Significant heavy metal sources from industrial activities
Table 2.3: The health hazards of various toxic heavy metals 12
Table 2.4: Copper concentration in industrial and municipal wastewater, and the
range reported for clean water matrix
Table 2.5: Maximum guideline concentration of copper in drinking water and wastewater discharges 17
Table 2.6: Dissolved copper concentration (mg/L Cu) for different water hardness
values between 10 and 300 mg/L CaCO ₃ 18
Table 2.7: The main advantages and disadvantages of the conventional methods
for treatment of heavy metals in wastewater
Table 2.8: General characteristics of physisorption and chemisorption
Table 2.9: Adsorption parameters of various adsorbents
Table 2.10: Summary of the modified biomass as adsorbent for the removal of
heavy metal ions from the water matrix
Table 2.11: Various applications of RSM in wastewater treatment process
optimisation
Table 2.12: Various applications of ANN in wastewater treatment process
optimisation40
Table 2.13: Differences between SEM and TEM 51
Table 2.14: The strength and limitations of atomic spectroscopy techniques58
Table 3.1: List of reagents and chemicals 62
Table 3.2: Operating parameters for metal analysis using Atomic Absorption
Spectrometry (AAS)78
Table 3.3: Experimental ranges and levels of the independent variables for clean
water
Table 3.4: Experimental ranges and levels of the independent variables for
wastewater effluent
Table 4.1: Assignment of bands in IR spectra of CNWs

Table 4.2: Zeta potential of TEMPO-oxidised cellulose from different sources93
Table 4.3: BET surface areas and total pore volume of the samples
Table 4.4: Carboxylate contents of cellulose before and after TEMPO-mediated
oxidation98
Table 4.5: The amount of carboxylate content for oxidised CNWs for different
batches
Table 4.6: The carboxylate content of oxidised CNWs under dry conditions100
Table 4.7: The sorbent ability under wet conditions (T = 21.3°C, C_i = 50.3 mg/L,
m = 2.1g/L)
Table 5.1: Data for the control experiments to determine appropriate filter103
Table 5.2: The equilibrium time for Cu(II) removal by common adsorbents107
Table 5.3: The effect of Cu(II) solution pH on the removal of Cu(II); $C_0 = 9$
mg/L , $m = 4 g/L$, $T = 25^{\circ}C$, $t = 30 min108$
Table 5.4: Ranges of the investigated parameters. 111
Table 5.5: CCD Experimental ranges and levels of the independent variables112
Table 5.6: The experimentally obtained percentage of Cu(II) removal compared
to that predicted by the central composite design (CCD) models113
Table 5.7: Analysis of variance (ANOVA) for Cu(II) removal prediction using
the RSM model
Table 5.8: The experimentally obtained percentage Cu(II) removal compared to
that predicted by the artificial neural network (ANN)119
Table 5.9: Validation data for 14 unseen experiments (CCD)
Table 5.10: Comparison of the predictive abilities of the RSM and ANN models.
Table 5.11: Comparison of maximum adsorption capacities of Cu(II) ions by
different adsorbents
Table 5.12: Adsorption kinetic parameters of Cu(II) onto oxidised CNW
adsorbents132
Table 6.1: Detection capabilities of ICP-MS for wastewater analysis

Table 6.2: Concentration of selected elements in Milli-Q water from different
control experiments
Table 6.3: Concentration of selected elements in reagents and chemicals 140
Table 6.4: Comparison of selected heavy metal concentrations after 0, 72 and 120
hours storage of effluent samples141
Table 6.5: Characteristics of influent and effluent wastewater
Table 6.6: Data for water quality parameters for influent and effluent wastewater samples during February-March 2016
Table 6.7: Data over time of Cu(II) removal as evaluated by benchmark studies
(Initial concentration of wastewater effluent spiked with Cu(II)=4.0
mg/L, sorbent dosage=1.0 g/L, pH=6.0)151
Table 7.1: CCD Experimental ranges and levels of independent variables157
Table 7.2: The experimentally obtained for spiked Cu(II) removal from the
wastewater effluent compared to that predicted by the response surface
methodology (RSM)159
Table 7.3: Analysis of variance (ANOVA) of Cu(II) removal prediction by using
RSM model161
Table 7.4: The experimentally obtained removal of Cu(II) compared to that
predicted by ANN, and ANN including wastewater (WW) variability,
as defined by benchmark experiments conducted for each sampling trip
Table 7.5: The experimentally obtained removal of Cu(II) compared to that
predicted by ANN, and ANN including wastewater (WW) variability,
as defined by benchmark experiments conducted for each sampling trip
Table 7.6: Validation data for 10 unseen experiments
Table 7.7: Comparison of the predictive abilities of RSM and ANN model180
Table 7.8: Optimised operating conditions for spiked Cu(II) removal from
wastewater effluent
Table 7.9: Effect of pressure on spiked Cu(II) removal efficiency from
wastewater effluent by oxidised CNW adsorbents

Table A.1: Data collected using certified 1000 mg/L calibration standard solution
of Cu (in 2 % HNO ₃) of known concentration
Table A.2: Data collected using certified 1000 mg/L calibration standard solution
of Cu (in 2 % HNO ₃) of known concentration217
Table C.3: Kinetic parameters obtained from pseudo-first order for Cu(II)
removal using modified CNW adsorbents at different initial Cu(II)
concentration (a) 10 mg/L and (b) 60 mg/L221
Table C.4: Kinetic parameters obtained from pseudo-second order for Cu(II)
removal using modified CNW adsorbents at different initial Cu(II)
concentration (a) 10 mg/L and (b) 60 mg/L222
Table D.1: ANN preliminary optimisation studies for wastewater matrix (a) at
constant initial Cu(II) concentration, (b) at constant pH, and (c) at
constant sorbent dosage

LIST OF ABBREVIATIONS

AA	Annual Average
AAD	Absolute average deviation
AAS	Atomic absorption spectrometry
AC	Activated carbon
ACC	Activated carbon clothe
ACF	Activated carbon fibrous
ANN	Artificial neural network
ANOVA	Analysis of variance
BBD	Box-behnken design
BET	Brunauer, Emmett and Teller
BJH	Barrett, Joyner and Halenda
BOD	Biological oxygen demand
CCD	Central composite design
CEC	Council European Communities
CIP	Chemical Investigation Programme
CNT	Carbon nanotubes
CNW	Cellulose nanowhisker
COD	Chemical Oxygen Demand
DO	Dissolved oxygen
DoE	Design of experiment
EPA	Environmental Protection Agency
EQS	Environmental Quality Standards
EU	European Union
FTIR	Fourier transform infra-red spectroscopy
GAC	Granular activated carbon
GFAAS	Graphite furnace atomic absorption spectrometry
HRT	Hydraulic retention time
ICP-MS	Inductively coupled plasma mass spectrometry
ICP-OES	Inductively coupled plasma optical emission spectrometry
IR	Infrared spectroscopy
LoD	Limit of detection
LoQ	Limit of quantification

MAC	Maximum Allowable Concentration
MLP	Multilayer perceptron
PAC	Powder activated carbon
PHS	Priority hazardous substance
PS	Priority substance
RMSE	Root mean square error
rpm	Revolutions per minute
RSM	Response surface methodology
SEM	Scanning Electron Microscopy
SPE	Solid phase extraction
TDS	Total dissolved solid
TEM	Transmission Electron Microscopy
TEMPO	2,2,6,6-tetramethyl-1-piperidinyloxy (TEMPO) free radical
TOC	Total organic carbon
TSS	Total suspended solid
UKWIR	UK Water Industry Research
WHO	World Health Organisation
WWTP	Wastewater treatment plant

1.1 THESIS OVERVIEW

This study focuses on the modification of cellulose nanowhisker (CNW) and evaluates the capability of this modified adsorbents for the remediation of copper from the water matrices. Chapter Two begins with a review of the literature relating to the sources of copper in water, and their impact upon health and the environment. The amount of copper pollutant released into the aqueous environment is increasing as a result of anthropogenic activities, with the effects of this pollutant on the ecosystem causing global concern. This chapter also introduces treatment technologies for copper removal from water and wastewater, followed by current knowledge on CNW as a potential adsorbent. This chapter also presents a review on the application of response surface methodology (RSM) and artificial neural network (ANN) in developing an approach for the evaluation of heavy metals adsorption process. The results chapters commence with the characterisation of CNW before and after the modification. In Chapter Four, the size, morphology, surface charge, and stability of unmodified CNW and oxidised CNW adsorbents are summarised. The potential of oxidised CNW adsorbents for the remediation of Cu(II) from a clean water matrix using RSM and ANN models is described in Chapter Five. A deeper understanding of the wastewater matrix (wastewater composition and its variability) and the remediation of spiked Cu(II) from a real wastewater effluent are summarised in Chapters Six and Seven, respectively. Considerations for design and application to remediate Cu(II) from wastewater by developing a continuous flow experiment are described in Chapter Eight.

1.2 RESEARCH BACKGROUND

Adsorption is a complex process, as it involves the interaction of various parameters. Moreover, the complexity and variability of the wastewater matrix also has a direct impact on the process performance (Hanafiah *et al.*, 2014; Ileri *et al.*, 2014). In addition, studies tend to focus on evaluating one single parameter at a time, which assumes each parameter operates independently. This is non-practical because parameter interactions cannot be elucidated using this approach (Turan *et al.*, 2013b).

Developing a new mathematical modelling for remediation process not only can reduce cost and time in wastewater treatment, but also to investigate the effects of input variables or factors on an output variable or response. The capabilities of these models were applied to the case study: remediation of copper from water matrices using modified cellulose nanowhisker (CNW) adsorbents. In this study, the central composite design (CCD) was selected because as it has been extensively applied in adsorption studies (Bingol *et al.*, 2012; Shanmugaprakash and Sivakumar, 2013). The mathematical models, response surface methodology (RSM), and artificial neural network (ANN) are among the most popular models used in research on remediation of heavy metals from the clean water matrix (Li *et al.*, 2014; Mandal *et al.*, 2015). This approach has also been used for the study of adsorption of dyes and organic compounds from water matrices (Gengec *et al.*, 2013; Sinha *et al.*, 2013).

Copper is a naturally occurring element which is found in water, air, and soil, and is considered as one of the most dangerous substances found in the environment (CEC, 1976). It has been identified in many water matrices originating from industrial and municipal waste, mining operations, coal fired power generation, and urban runoff (Kadirvelu *et al.*, 2001; Pazwash, 2011; Pereira *et al.*, 2009). Although small amounts of copper are essential for human health, high concentrations of copper in water are recognised as a major problem that can cause physiological and health effects (Lim and Schoenung, 2010). The maximum guideline concentration limit for copper discharge to water has been established by the Water Framework Directive of water policy discharge to inland

surface water directive (2000/60/EC) (CEC, 2000). With these strict guidelines for the regulated levels of copper for wastewater discharge (1–28 μ g/L), proper and suitable treatment is required in order to meet these discharge limits.

Treatment technologies such as chemical precipitation, ion exchange, reverse osmosis, membrane technologies, and electrochemical treatments have been proposed for the removal of copper from industrial effluent (Gavrilescu, 2004; Wang and Chen, 2009). These technologies have their own limitations, for instance, high operating costs, high energy requirements, low adsorption capacity, and the production of toxic sludge (Wang and Chen, 2009).

Adsorption has become one of the alternative treatments for the removal of low concentrations of heavy metals from the water matrices. The adsorption process has been practiced widely in the wastewater treatment due to operational simplicity, regeneration and reuse of adsorbent, and low cost treatment technology (Cojocaru *et al.*, 2009). Activated carbon, for instance, is a widely employed adsorbent due to its high porosity and surface area. However, various studies have demonstrated the limitations of activated carbon as an expensive and non-renewable adsorbent for heavy metals removal (Geyikci *et al.*, 2012; Reddy *et al.*, 2012). From a sustainability and economical perspective, adsorbents derived from renewable and sustainable resources are an interesting option. Fly ash (Javadian *et al.*, 2015), natural zeolites (Ansari *et al.*, 2015), clay (Unuabonah *et al.*, 2013) and agricultural biomass (Thirumavalavan *et al.*, 2010) are the examples of promising sources of adsorbents due to their low cost and abundance.

Nevertheless, without physical and chemical modifications, these untreated or unmodified adsorbents can cause several problems, for example, high chemical oxygen demand (COD), biological oxygen demand (BOD) and total organic carbon (TOC) due to the release of soluble organic compounds contained in plant materials (Himanshu and Vashi, 2014; O'Connell *et al.*, 2008a). Therefore, chemical modifications and pre-treatment methods have been developed in order to increase the adsorption capacity and improve the adsorption performance. Chemical modifications are carried out by attaching the potential functional groups onto the adsorbent surface. The attachment of different functional groups onto chemically modified biomass adsorbents such as fruit peels (Liu *et al.*, 2013), wood sawdust (Pereira *et al.*, 2009), egg shell (Wang *et al.*, 2013), jute fibre (Shukla and Pai, 2005b), and pine bark (Argun *et al.*, 2009) have been reported. Furthermore, some plant waste requires separate pre-treatment/purification to remove impurities and colours which could affect the performance of the adsorbents (Bulut and Tez, 2007).

Cellulose has been identified as a promising adsorbent for the remediation of heavy metals from the water matrix (Alves *et al.*, 2009; Isobe *et al.*, 2013). In comparison with other biomass adsorbents, cellulose is the most abundant natural biopolymer on earth (Brown, 2004). Numerous chemical modifications and pretreatment methods for isolating CNW from various plant fibres have been attempted. One of the most common methods is acid hydrolysis using sulfuric acid, through which CNWs with length of 100-320 nm and width of 6-70 nm could be extracted from various plant fibres (Elazzouzi-Hafraoui *et al.*, 2008; Hsieh, 2013). Although CNWs have been recognised due to their high surface area and high reactive group density on the surface, only limited research has been published on using CNWs as an adsorbent, as the majority of the literature has mainly focused on macroscopic lignocellulosic biomass such as jute, orange peel, wood sawdust, wood pulp, and sugarcane bagasse fibres, rather than pure cellulose (Eyley and Thielemans, 2014; Reddy, 2012).

In recent years, 2,2,6,6-tetramethylpyperidine-1-oxyl TEMPO-mediated oxidation has frequently been used to introduce carboxyl functional groups on the surface of nanowhiskers or native cellulose without affecting the crystallinity or changing the original fibrous morphology (Saito *et al.*, 2005). This study not only focuses on the modification of CNW through a controlled surface oxidation, but also on the stability of the oxidised CNW adsorbents under dry conditions and in the water matrix. Although the carboxyl group can also be introduced through esterification, the reported modification using succinic anhydride is not a suitable and sustainable process, as pyridine is a well-known problem in the chemical industry as it may cause harmful health effects (Health, 1992; Xu *et al.*, 2015).

While the application of oxidised CNW adsorbents in removing heavy metals is still in its development phase, other processes and modifying adsorbents are now rapidly taking place, in line with the fast-paced growth in the field (Chand *et al.*, 2015; Duan *et al.*, 2013; Góes *et al.*, 2016). However, in many studies, the range of the investigated parameters for the adsorption process is often not representative of the actual conditions in a wastewater treatment plant (WWTP) (Thirumavalavan *et al.*, 2010). Moreover, most of these previous studies focused on the adsorption process for the remediation of heavy metals from a clean water matrix, and little exists on real-world wastewater matrices. As the remediation process from wastewater is often complicated due to the variation in wastewater composition, studying this factor and its influence on adsorption behaviour and removal capability in the case of varying wastewater matrix composition could lead to a better understanding of the actual application performance of an adsorbent for remediating pollutants.

However, most of the previous literature focuses attention on adsorption studies by using either RSM or ANN, without comparing the performances between these two models. Furthermore, the testing of both RSM and ANN using new sets of experiments not belonging to the training data set has only been undertaken by a limited number of studies on biomass adsorption, and without consideration of how the additional experiments represent the system and provide a more accurate indicator of performance (Bingol et al., 2012; Ghosh et al., 2015). Therefore, model suitability for interpolated and extrapolated experimental parameters was tested, which is rare in the existing literature, but provides valuable insights into applicability of the approaches tested in this work. Moreover, no work in the reported literature has included matrix complexity and the variability of the wastewater as one of the independent variables in ANN modelling. Therefore, in order to study the effect of variation in wastewater composition on removal of spiked Cu(II) from wastewater effluent, a benchmark experiment was conducted for each wastewater sample used in the experiments, which determined Cu(II) removal by oxidised CNW adsorbents for that particular water matrix in time and space. Studying the effect of matrix complexity and the variability of the wastewater, along with applying realistic conditions for WWTP, could lead to the establishment of a good knowledge based on adsorption behaviour and provide the foundation for further studies.

1.3 AIM AND OBJECTIVES

The aim of this research is to develop a new mathematical modelling using response surface methodology (RSM) and artificial neural network (ANN) models, to evaluate the capability of modified cellulose nanowhisker adsorbents for the remediation of copper from water matrices under realistic conditions, and to show the potential of these models to deal with the complexity and variability of the wastewater matrix.

The specific objectives of this research are:

- 1. To characterise the adsorbents through various physical and chemical analytical methods, in order to check the suitability of CNW adsorbents as an example of potential adsorbent to remove copper form water matrices.
- 2. To build two models, RSM and ANN, and assess their abilities to determine the effectiveness of oxidised CNW adsorbents functionalised with carboxylate functionalities at removing copper ions from water matrices.
- 3. To understand the complexity and variability of the wastewater matrix, including evaluating the impact of the wastewater matrix temporally on adsorbent performance to remediate copper pollutant from the wastewater matrix, in order to include matrix complexity and the variability of the wastewater as one of the independent variables in ANN modelling.
- To optimise the adsorption process parameters such as pH, temperature, sorbent dosage and initial Cu(II) concentration using RSM and ANN for the removal of copper from water matrices
- 5. To test if CNWs functionalised with carboxylate functionalities could work in real application by developing a continuous flow experiment under optimum adsorption conditions.

2.1 IMPORTANCE OF WATER

Water is a crucial part of everyday life as every living thing on earth, such as microorganisms, plants, animals and humans consist mostly of water, and require access to water in order to function. Moreover, water is also important in many industries and domestically, which in turn generate used water or wastewater. The wastes generated from anthropogenic activities decrease the quality of water and impact the natural ecosystems that support human health, food production and biodiversity (Singh *et al.*, 2015). This is because untreated wastewater consists of many pollutants that can adversely affect organisms and flora that come into contact with that wastewater. Hence wastewater needs to be treated before being reused, whether this involves discharge into water bodies or not, and legislation such as The Urban Wastewater Treatment Directive (91/271/EEC) and the Water Framework Directive (2000/60/EC) clearly support the need for treatment prior to reuse (Hughes and Gray, 2013b).

In general, the hydrological cycle is a model that describes the storage and circulation of water between the biosphere, atmosphere, lithosphere and hydrosphere. However, the hydrological cycle lacks the anthropogenic interaction which renders it more complex and is then known as the urban water cycle (Marsalek *et al.*, 2008). The simple concept of the urban water cycle can be defined as a water balance where water that is used should be able to be treated, reused, and returned to nature as purified wastewater. Moreover, urban sources of water pollution have often been mentioned as the major cause of poor water quality (Eslamian, 2016). A growing population in urban areas for instance, will create more domestic waste and wastewater, placing increasing burdens on treatment systems and creating the need to meet over-stringent legislation aimed at reducing the pollutant load of treated water entering the receiving rivers and water bodies.

Urbanisation and industrialisation have caused serious ecological problems, such as water pollution in urban areas (Pazwash, 2011). Various pollution sources, for example toxic waste, agricultural activities, household wastes and transportation activities can contaminate ground and surface water (Ramwell *et al.*, 2014; Revitt *et al.*, 2014). Storm water is also considered a major pollution source and affects the quality of water (He *et al.*, 2014). The sources of water pollution in the urban water cycle are illustrated in Figure 2.1.



Figure 2.1: Sources of water pollution in the urban water cycle (Eckhardt, 2012)

Anthropogenic pollutants are listed as harmful and dangerous to the environment and to life by most legislation (Chernen'kova *et al.*, 2014; Liu *et al.*, 2014c). The impact of human activities from agricultural and urban runoff, municipal wastewater, industrial trade, main drainage and landfill leachate often causes serious water pollution and affects water quality. Therefore, knowledge regarding anthropogenic pollutants is widely regarded as necessary to the understanding of the effect of these pollutants in any environmental system. These pollutants are categorised under three major groups: nonpolar organic compounds (pesticides, polycyclic aromatic hydrocarbons), polar organics (pharmaceuticals), and heavy metals (e.g. copper and cadmium). The major pollutants associated with anthropogenic activities are summarised in Table 2.1.

Activity / source category	Pollutants	References
Agriculture	Phosphates, nitrates, pesticides	(Munoz <i>et al.</i> , 2009; Yang <i>et al.</i> , 2015)
 Urban activities Transportation Construction and building Misconnections (wastewater draining to the wrong place) Run-off activities (e.g.car 	Phosphates, heavy metals, hydrocarbons	(Carlson <i>et al.</i> , 2011; Hasenmueller and Criss, 2013)
washing)		
Industrial	Polycyclic aromatic hydrocarbons (PAH), heavy metals	(Gao <i>et al.</i> , 2014; Nhapi, 2011; Pereira <i>et al.</i> , 2009)
Waste disposal	Nitrate, ammonia, phosphate	(Divers <i>et al.</i> , 2014; Rhind, 2009)
Mining (abandoned mines)	Heavy metals	(Gavrilescu, 2004)

Table 2.1: The sources of anthropogenic pollutants

2.2 WATER POLLUTION BY HEAVY METALS

2.2.1 SOURCES OF HEAVY METALS IN WATER

Sources of heavy metals derive in the environment mainly from anthropogenic activities as well as from natural constituents of the earth's crust. Heavy metals are not only emitted from industrial activities, but also from major urban sources including urban storm water runoff, domestic effluents and municipal wastewater (Karamouz *et al.*, 2010). For example, urban storm water runoff contains heavy metals such as lead (Pb), zinc (Zn), copper (Cu) and cadmium (Cd) that enter the municipal wastewater treatment plants or otherwise contaminate surface water and ground water sources (Liu *et al.*, 2014c).

Moreover, natural processes can affect the environment and cause pollution through metal corrosion, atmospheric deposition, soil erosion and leaching of heavy metals, sediment re-suspension and metal evaporation (Chen *et al.*, 2016). Heavy metals in soils resulting from weathering of the underlying bedrock are

normally harmless, and natural sources of metals become potentially toxic only when these heavy metals are transported by water in a dissolved or particulate state and enter the water environment through rivers or land runoff (He *et al.*, 2014). For example, monitoring studies in the United Kingdom (UK) have reported that the concentration of Cu, Pb and Zn from managed and unmanaged woodland exceed the required discharge limit, leading to metal concentrations in water reaching the threshold for adverse effects (Dore *et al.*, 2014).

Humans contribute largely to the sources of metals entering the environment through a variety of activities including industrial activities, mining, burning fossil fuels and domestic effluents. Major industrial sources, as listed in Table 2.2, contribute significant amounts of heavy metals such as Cu, Cd, Pb, Zn, As, Fe, Hg, and Ni to the water environment. Although heavy metals present in wastewater are mainly industry-related, other factors such as household effluents, drainage water, business effluents (car washes, dental uses), and transportation activities can also provide a significant contribution to heavy metal loads (Carlson *et al.*, 2011).

Sources	Cu	Cd	Pb	Zn	As	Fe	Hg	Ni
Non-ferrous metal production		\checkmark	\checkmark				\checkmark	
Electroplating	\checkmark							
Chemical manufacturing		\checkmark	\checkmark				\checkmark	\checkmark
Paint and pigments		\checkmark	\checkmark					
Metal manufacturing, refining, and finishing works			√	√	\checkmark	√	\checkmark	\checkmark
Leather tanning	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark		

Table 2.2: Significant heavy metal sources from industrial activities (Barakat, 2011; Das, 2008)

2.2.2 IMPACT OF HEAVY METALS ON THE AQUATIC ENVIRONMENT

Numerous studies showed that heavy metals are contained in our drinking water (conc), air and soil (Mudhoo *et al.*, 2011; Revitt *et al.*, 2014; Yanna *et al.*, 2014). This is because metals cannot be subjected to chemical degradation beyond the elemental state (Lysenko *et al.*, 2010). Being non-biodegradable, they can therefore cause a variety of toxicities towards aquatic and other ecosystems (Lim and Schoenung, 2010). Moreover, heavy metals can cause serious problems that can affect human health and aquatic life even at low concentrations (Ashraf *et al.*, 2011). For example, low concentrations of lead are capable of inhibiting the growth of plants due to instability in ion uptake by plants (Zhu *et al.*, 2011). Therefore, in order to protect the environment, wastewater must be sufficiently treated before being discharged. Table 2.3 summarises the effects of toxic heavy metals on human health and aquatic life.

Heavy metals	Health and risks	Daily intake (µg/person)	References
Cadmium (Cd)	 Renal dysfunction Lung disease, bone defects Effects on the myocardium in animals Larval mortality 	0.93	(Rao <i>et al.</i> , 2012)
Copper (Cu)	Anaemia, liver and kidney damageStomach and intestinal irritation	3.00	(Sengil and Oezacar, 2008)
Iron (Fe)	ConjunctivitisPneumoconiosis	0.07	(Abdel-Sabour et al., 2001)
Lead (Pb)	 Chronic damage to the nervous system Effects on haemoglobin synthesis and anaemia Gastrointestinal tract damage Joint and reproductive system harm 	3.54	(Azila <i>et al.</i> , 2008)
Mercury (Hg)	TremorsGingivitisCongenital malformations	0.36	(Ratcliffe et al., 1996)
Nickel (Ni)	Decreased body weightHeart and liver damage	1.40	(Oezer <i>et al.</i> , 2008)
Zinc (Zn)	• Metal fume fever, headache, aches, cough	0.05	(Lim and Schoenung, 2010)

Table 2.3: The health hazards of various toxic heavy metals

2.2.3 COPPER AS A CASE STUDY HEAVY METAL

Copper is a reddish-brown coloured metal, tough, ductile and corrosion resistant. It is the third most used metal in the world, and has widespread commercial uses in a range of applications (Bini and Bech, 2014). Copper is of major environmental concern (CEC, 1976) as it and its compounds are abundant in the environment and have become a threat to human health due to their water solubility and ready binding to natural organic materials (Sengil and Oezacar, 2008). Moreover, copper compounds do not break down easily in the environment, thus can accumulate in animals and plants (Melignani *et al.*, 2015). With a wide range of applications and ubiquitous usage, copper is one of the most common metals found in common industrial and municipal wastewater and urban runoff (Gardner *et al.*, 2012; Liu *et al.*, 2002).

2.2.3.1 Sources of Copper in the Aquatic Environment

Sources of copper in the aquatic environment can be divided into two categories: point and non-point emission sources.

- A point source is where the pollutants come from a single location, such as a pipe or drain and commonly from industrial waste. These include operations that produce copper, consume copper or use copper compounds in product manufacture. The point sources include mining operations, metal processing, coal fired power generation, electrical applications and chemical manufacture (Kadirvelu *et al.*, 2001; Pereira *et al.*, 2009).
- Non-point sources usually refer to indirect copper sources where the metal is introduced into the aquatic environment via storm run-off. Examples of some non-point sources include transport (mechanical wear and tear of brake pads and tyres), household plumbing, metallic building products, pavements, roads, and highways (Gavrilescu, 2004; He *et al.*, 2014; Pazwash, 2011; Wang *et al.*, 2009).

2.2.3.2 Toxicity of copper to humans and the environment

Copper is essential for humans and must be absorbed in small amounts on a daily basis by eating and drinking in order to maintain good health. However, high levels of copper can be harmful to health and cause anaemia, liver and kidney damage, and stomach and intestinal irritation (Sengil and Oezacar, 2008). Furthermore, copper pipe and tubing are widely used for domestic water systems, and contamination of drinking water can easily occur from corrosion of household pipes (Hu *et al.*, 2012; Turek *et al.*, 2011). Therefore, it cannot be easily removed from the water system. The data on copper concentration in wastewater from different industries and municipal wastewater, and the range of copper concentration in a clean water matrix are shown in Table 2.4.

Table 2.4: Copper concentration in	industrial	and	municipal	wastewater,	and	the
range reported for clean water matr	ix					

Wastewater	Copper concentration (mg/L)	References
	20.0	(Sciban et al., 2007)
Flootroplating	60.0	(Malakootian et al., 2011)
Electroplating	20.0	(Da'na and Sayari, 2012)
	22.57	(Kulbat et al., 2003)
Copper plating industries	126.0	(Kadirvelu et al., 2001)
Pharmaceutical	17.6	(Hamid <i>et al</i> .)
	0.5	(Hughes and Gray, 2013a)
Municipal	0.2	(Kulbat et al., 2003)
	0.02	(Gardner et al., 2012)
	25–250	(Aksu and Isoglu, 2005)
Clean water matrix	10–400	(Zhu et al., 2009)
	25–200	(Amarasinghe and Williams, 2007)
Copper concentration varies in wastewater both temporally and spatially. Other constituents, which are pollutants themselves, also vary temporally and spatially (within a WWTP and between different wastewater streams). Wastewater entering a treatment works is not of constant composition; how treatment technology performs in the face of that matrix variability needs consideration in order to understand how to develop technologies for their abilities to treat wastewater. However, to date no work has examined the effect of wastewater composition and its variability on the remediation of Cu(II) from the wastewater samples in their studies, assumptions that the real wastewater composition is constant, or that its composition has no effect on adsorbent capability, are questionable and may lead to inaccurate and non-trustworthy data being presented.

2.2.3.3 Environmental regulations and standards for discharge of effluent into water

Copper pollution in the water environment has been reported for a number of years and has been a topic of concern in UK legislation for decades (Hutton and Symon, 1986). Besides proper treatment being provided by waste management, the other way to improve environmental management is by reducing the production of wastes and restricting their discharge. Therefore, to support and improve the understanding of the environmental impacts of hazardous chemicals and pollutants, European Union (EU) directives have improved the scope of pollution control measures required to protect surface water. There are also a number of specific EU directives for heavy metal pollutants in the water environment. Such directives include:

- Dangerous Substances Directive (76/464/EEC) (CEC, 1976)
- The Sewage Sludge Directive (86/278/EEC) (CEC, 1986)
- Water Framework Directive (2000/60/EC) (CEC, 2000)
- Fish Directive (2006/44/EC) (CEC, 2006)
- Environmental Quality Standards (2008/EC) (CEC, 2008)

The Dangerous Substances Directive (76/464/EEC) was one of the first water related, aquatic environment related, directives to be adopted. It covers pollution caused by certain dangerous substances, including heavy metals. The directive established the concept of List I and List II, with the purpose of eliminating pollution by the dangerous substances in List I and of reducing pollution by the dangerous substances in List I and of reducing pollution by the dangerous substances are kept clean, European water policy has undergone a thorough restricting process. As a result, a new Water Framework Directive (2000/60/EC) was adopted in 2000 and will be the operational tool for community action in the field of water policy.

Environmental Quality Standards (2008/EC) have been defined by EU directives including the Water Framework Directive (WFD) and the Priority Substances Daughter Directive, which specify the maximum permissible concentration of priority substances (e.g. metals, polyaromatic hydrocarbons, biocides) in the water environment. Environmental Quality Standards (EQS) are benchmarks, expressed as an annual average value (AA) or maximum allowable concentration (MAC), or other limit value of likely interest such as a predicted no effect value (PNEC), used as guidelines to ensure that the concentrations within the effluent, measured at the monitoring point, do not exceed the standards set (CEC, 2008).

While EU Directives have established EQS to set maximum admissible concentrations of over 30 substances at EU level, the Chemical Investigation Programme (CIP) is a programme of the UK Water Industry Research (UKWIR) organisation, which assists in improving the management of wastewater treatment across the UK (Gardner *et al.*, 2013). The main objective of this programme is to monitor the final effluents from over 160 wastewater treatment works and to determine the final concentrations of chemicals discharged to the water environment. The CIP covers more than 70 target chemicals, including nine metals (Ni, Pb, Cu, Zn, Cd, Hg, Fe, Al and Ag). The metals for which concentrations were reported to be above the standards were Cd, Cu, Ni, Pb and Zn (Gardner *et al.*, 2012). Since the regulations regarding the discharge of wastewater are becoming ever more stringent, wastewater needs to be treated prior to discharge in order to meet the discharge limits of heavy metals and other pollutants.

There are few examples of approaches that have been implemented as part of the CIP in order to improve wastewater effluent quality. Weise *et al.*, (1997) reported that after 30 years of implementation of activated sludge treatment in the Beckton wastewater treatment plant in London, significant improvement in water quality has resulted. Moreover, the Water Industry Act 1991 gave the occupiers of trade premises the right to an effective trade effluent collection and treatment service. Thus, trade effluent produced from commercial and industrial operations will be monitored and controlled by water companies prior to discharge (Act, 1991).

2.2.3.4 Legislative Limits for Copper in the Aquatic Environment

Copper (Cu) has also been listed in European Directives under dangerous substances (76/464/EC), as a metals that is dangerous and toxic to the aquatic environment (CEC, 1976). Moreover, the World Health Organisation International Standards for Drinking Water has suggested a guideline value that has been adopted by the 1963 and 1971 International Standards as a maximum allowable or permissible Cu concentration in drinking water (Donohue, 2004). The maximum guideline concentration limits of Cu in drinking water and water discharge are summarised in Table 2.5, which includes a comparison between values in the WHO guideline value for Cu in drinking water, Water Framework Directive (WFD) on water policy (2000/60/EC) and PNEC values (UK, 2008).

Copper regulatory limit					
Drinking water	W	astewater discharge			
WHO	WFD (2000/60/EC	C) PNEC			
$2 m \sigma/I$	1.0.ug/I	2.6-5 µg/L (Salt water)			
2 mg/L	1.0 μg/L	1-28 µg/L (Fresh water)			
WHO=World Health	Organization; WFD=W	Water Framework Directive; PNEC=			
predicted no effect valu	e				

Table 2.5: Maximum guideline concentration of copper in drinking water and wastewater discharges

Moreover, the Fish Directive (2006/44/EC) also provides guideline values for copper in fresh water, which are used to ensure that such water is capable of supporting aquatic life (CEC, 2006). The guidelines for dissolved copper concentrations for different water hardness values are shown in Table 2.6.

Table 2.6: Dissolved copper concentration (mg/L Cu) for different water hardness values between 10 and 300 mg/L CaCO₃

Water hardness (mg/L CaCO ₃)	10	50	100	300
Copper concentration (mg/L)	0.005	0.002	0.04	0.112

2.2.3.5 Conventional wastewater treatment approaches for copper removal

The main objective of wastewater treatment is to allow municipal discharges (from domestic sources) and industrial waste to be treated, to reduce pollutant load so that the treated water can be discharged to the water environment without adversely affecting the aquatic organisms within, or human health. The wastewater treatment plant (WWTP) comprises several treatment stages known as preliminary, primary, secondary and in rare cases, tertiary/advanced treatment (WEF, 2007). Preliminary treatment is where the solids and large materials often found in raw wastewater are removed using coarse screening and grit removal. Since organic solid matter and fine particles still remain suspended in the wastewater, these remaining particles settle out to form sludge. The clarified water then enters secondary treatment which involves the removal of biodegradable dissolved and colloidal organic matter through aerobic biological treatment processes (Yu *et al.*, 2013a). The WWTP is summarised in Figure 2.2.



Figure 2.2: Schematic of a typical wastewater treatment plant (Spellman, 2013)

Tertiary and/or advanced wastewater treatment is only applied when specific wastewater pollutants cannot be removed or treated by the prior treatment stage and there is a legislative driver (CEC, 2000). It is very rarely used. Tertiary/advanced treatment technologies including precipitation, ion exchange, electrochemical reverse osmosis, filtration. treatment. and membrane technologies have been proposed and applied to the removal of metal ions from wastewater effluent (Gavrilescu, 2004; Wang and Chen, 2009; Wang et al., 2013). However, these methods are considered economically inefficient due to the high operating cost of treating large volumes of wastewater and water with low concentrations of metal ions (Wang and Chen, 2009). Table 2.7 shows the main advantages and disadvantages of the conventional methods in removing heavy metals from the water matrix.

Amongst these technologies, adsorption using natural sorbents is considered a popular and effective process for the remediation of heavy metals from wastewater (Ali and Gupta, 2006). The adsorption process is effective and simple compared to other treatments technologies, especially in removing low concentrations of heavy metals from the water matrix (Ashraf *et al.*, 2011). Removal of these pollutants by an adsorption process also offers the opportunity to consider waste as a resource, by recovering the heavy metals for reuse by regenerating the adsorbent. This is in line with the UK Water Industry Research (UKWIR) Road Map that looks to recovering opportunities within WWTPs (UKWIR, 2015).

Conventional treatments	Advantages	Disadvantages	References
Chemical precipitation	• Low capital cost	• Sludge generation	(O'Connell et al., 2008a)
	• Simple procedure	• High cost for sludge disposal	
Coagulation-	• Good sludge settling and dewatering	Chemical consumption	(Oloibiri et al., 2015)
Flocculation	characteristic	• Sludge generation	
Electrodialysis	High separation selectivity	• High operational cost due to	(Wang and Chen, 2009)
	 No chemical consumption 	membrane fouling	
		• Energy consumption	
Ion exchange	• High metal selectivity	• High maintenance cost	(Barakat, 2011)
Membrane filtration	High separation selectivity	• High initial capital cost	(Kurniawan, 2006)
	Small space requirement	• High maintenance cost due to	
	• Low solid waste generation	membrane fouling	
Reverse osmosis	• Effective in both small and large scale	• High capital and maintenance cost	(Fu and Wang, 2011)

Table 2.7: The main advantages and disadvantages of the conventional methods for treatment of heavy metals in wastewater

2.3 ADSORPTION PROCESS

2.3.1 MECHANISM OF ADSORPTION

Adsorption is a mass transfer process by which a substance (in this case, a heavy metal) is transferred from the liquid phase to the surface of a solid (the adsorbent), and becomes bound by physical and/or chemical interactions (Ali and Gupta, 2006; Iqbal *et al.*, 2005). Adsorption of ions can be simply represented as a physical and chemical reaction:

$$A + B \leftrightarrow A.B \tag{2.1}$$

where A is the adsorbate (pollutant), B is the adsorbent and A.B is the adsorbed compound. Moreover, the sorption process can be described in three main steps (as shown in Figure 2.3):

- i. The transport of the pollutant from the bulk solution to the sorbent surface
- ii. Adsorption on the particle surface or pores
- iii. Transport within the sorbent particle.
- iv. Adsorption is different from absorption, in which a substance diffuses into a liquid or solid to form a solution.



Figure 2.3: Mechanism of adsorption process

The term 'sorption' was first introduced in 1908, and described selective transfer to the surface and/or into the bulk of a solid or liquid (McBain, 1909). Several mechanisms can be classified in the adsorption process based on the type of bonding involved, such as physical adsorption involving Van der Waals forces and chemical adsorption involving strong chemical bonds between adsorbent and adsorbate (Gorelov *et al.*, 1994).

In physical sorption or physisorption, no exchange of electrons is observed and the adsorbate is bound to the surface by relatively weak Van der Waal's forces; hence the adsorbate is not fixed to a specific site and it is relatively free to move on the surface (Jiang *et al.*, 2013). Moreover, this type of adsorption is a reversible process that can be effected by increasing the temperature or decreasing the pressure. Chemical sorption or chemisorption involves strong forces between adsorbate and adsorbent, resulting in a change in the chemical form of the adsorbate. The bond formed is much stronger and stable at high temperature than that derived from Van der Waal's physisorption (Rouquerol *et al.*, 2013). The characteristics of physical and chemical sorption are presented in Table 2.8.

Physical sorption	Chemical adsorption
No electron transfer although polarisation of adsorbate may occur	Electron transfer leading to bond formation between adsorbate and adsorbent
The heat of adsorption is low, at about 20-40 kJ/mol	The heat of adsorption is high, at about 40-400 kJ/mol
Low activation energy	High activation energy
Multilayer adsorption	Monolayer adsorption
Non-dissociative and reversible process	Irreversible process

Table 2.8: General characteristics of physisorption and chemisorption (Rouquerol *et al.*, 2013)

2.3.2 ADSORPTION ISOTHERM

In general, an adsorption isotherm describes the dynamic adsorptive behaviour of any substance moving from the liquid phase or aquatic environment to a solid phase at a constant temperature and pH (Foo and Hameed, 2010). Therefore, isotherms play a crucial functional role in predictive modelling procedure for analysis and design of adsorption systems (sorption capacity of the sorbent). A variety of isotherm adsorption models have been applied in WWTPs to evaluate the design of adsorption systems and to describe the adsorption capacities of modified adsorbents (Shojaeimehr *et al.*, 2014). For example, two common adsorption isotherms, namely Langmuir and Freundlich isotherm models, have been widely applied. These isotherm models have been applied due to their simplicity and their good description of experimental behaviour under a large range of operating conditions (Ho, 2004). Below, a brief description of the two common adsorption isotherms is presented.

2.3.2.1 Langmuir Isotherm

The Langmuir isotherm was first established in the work by Langmuir in 1918, to describe gaseous adsorption on planar surfaces (Langmuir, 1918). According to Foo and Hameed (2010), the Langmuir isotherm is considered the simplest and the most widely used model for physical and chemical adsorption from both gaseous and liquid solutions. The derivation of this isotherm is based on certain implicit assumptions (Foo and Hameed, 2010):

- The adsorption occurs at a fixed number of definite localised sites
- Monolayer adsorption is formed on the surface of the adsorbent.
- The surface of the adsorbent is homogeneous
- There are no lateral interactions between neighbouring adsorbed adsorbate molecules

The Langmuir adsorption model is given by (Shojaeimehr et al., 2014):

$$q_e = \frac{q_m K_L C_e}{1 + K_L C_e} \tag{2.2}$$

where $q_e (mg/g)$ is the equilibrium mass of Cu(II) adsorbed per unit mass of sorbent, C_e is the equilibrium concentration (mg/L), q_m (mg/g) is the maximum adsorption capacity to form a complete Cu(II) ion monolayer bound to the surface and K_L is the Langmuir constant which related to the affinity of the binding sites. The characteristics of the Langmuir isotherm can be expressed in terms of a dimensionless equilibrium parameter (R_L), which is given by:

$$R_L = \frac{1}{1 + K_L C_o}$$
(2.3)

Where K_L is the Langmuir constant and C_o is the maximum adsorbate concentration (mg/L). The value of R_L indicates the type of the isotherm: unfavourable ($R_L>1$), linear ($R_L=1$), favourable ($0 < R_L < 1$) or irreversible ($R_L=0$).

2.3.2.2 Freundlich Isotherm

The Freundlich isotherm is often used to define adsorption to a heterogeneous surface across a wide range of adsorbate concentrations. This isotherm represents the relationship between the mass of metal adsorbed per unit mass of the adsorbent (q_e) and the concentration of the metal ion in solution at equilibrium (Limousin *et al.*, 2007). The Freundlich isotherm is expressed as:

$$q_e = K_f C_e^{1/n} \tag{2.4}$$

where K_f and n are the Freundlich constants that indicate adsorption capacity and adsorption intensity respectively. The n value is an indication of how favourable the adsorption process is, usually ranging from (1<n<10). K_f and n are determined from the linear plot of log q_e versus log C_e .

The Cu(II) adsorption isotherm of modified cellulose based adsorbents usually exhibits typical Langmuir behaviour, thus showing characteristics of physical and chemical adsorption (Chen *et al.*, 2009; Liu *et al.*, 2009; Liu *et al.*, 2002; O'Connell *et al.*, 2006a). This observation suggests the adsorption of a monolayer on the surface of the adsorbent. Moreover, Langmuir isotherm models have also been successfully applied to other adsorbents, such as carbon nanotubes (CNTs) and activated carbon (Liu *et al.*, 2014b). For example, Mobasherpour *et al.*,

(2011) used CNTs for the removal of copper from aqueous solution. Isotherm studies indicated that the Langmuir model fitted the experimental data better than the Freundlich model.

2.3.3 FACTORS AFFECTING THE RATE OF ADSORPTION

There are several key parameters that can influence the adsorption process. Experimental parameters such as pH, initial metal ion concentration, sorbent dosage, and temperature, are the important factors affecting heavy metal adsorption (Reddy, 2012). Moreover, the surface area of the adsorbent, pore structure and adsorbent particle size are also factors responsible for influencing heavy metal adsorption. Several studies have shown the effect of adsorbent particle size in the removal of heavy metals from a water matrix (Nadeem *et al.*, 2009; Yang *et al.*, 2011). Sengil and Oezacar (2008) carried out a series of experiments with different particle sizes of mimosa tannin resin adsorbent (Sengil and Oezacar, 2008). The results indicate that the uptake of Cu(II) by tannin resin is increased as the particle size decreases. This is due to the large surface area of small particles (100 μ m), while larger particles (250 μ m) result in a lower driving force per unit surface area for mass transfer during the adsorption process.

The pH of the adsorbate or solution is an important factor in the adsorption process, as the number of available sorption sites on the adsorbent is pH dependent (Reddy, 2012). The optimum pH for metal removal from the water matrix varies, depending on the target metal. However, the optimum pH range (2.0 - 9.0) for the adsorption process is not representative of the wastewater environment and is considered too acidic, as this pH range increases competition between protons and metal ions for active sites (Reddy, 2012). Results of adsorption capability for removing heavy metals from a wastewater matrix in this pH range will not be a true reflection of capability in a WWTP. Thus, it is important to note here that the removal of Cu(II) may decrease when pH is more acidic than the optimum. Moreover, the choice of a range of pH that does not consider metal hydroxide precipitation will also affect the removal of Cu(II) from the water matrix (Thirumavalavan *et al.*, 2010). For example, in Figure 2.4, the



lowest dissolved concentration of Cu(II) is approximately 0.1 mg/L, which occurs at a pH value of 8.0.

Figure 2.4: Theoretical solubility of heavy metals (EPA, 1995)

The initial concentration of metals in aqueous solution is also an important factor affecting the remediation of heavy metals from the water matrix. However, in many studies, the range of the investigated parameters for the adsorption process is often not representative of the actual conditions in a WWTP (Thirumavalavan *et al.*, 2010). For example, the majority of the studies have been performed with a high initial metal ion concentration (100-1000 mg/L), which is unrealistic for actual commercial adsorption processes as they are generally applied to low concentration streams. This is because the majority of conventional technologies are impractical for treating heavy metal pollutants at low concentrations due to high operating and maintenance costs (Ashraf *et al.*, 2011).

Temperature is another factor observed to influence the adsorption process, and which can alter the metal removal efficiency depending on the exothermic and endothermic nature of process (Reddy, 2012). The percentage removal of Pb(II), Cd(II) and Ni(II) by sawdust adsorbent increased with increasing temperature due to enlargement of pore size and activation of the adsorbent surface (Bulut and

Tez, 2007). However, the reported experiments were conducted under conditions unrealistic in a wastewater treatment environment, using temperatures up to 45 °C. Not only would such temperatures accelerate decomposition of chelating efficiency, leading to the decrease of the adsorption ratio (Sahan *et al.*, 2010), but by increasing the temperature of wastewater above ambient would be expensive. The studies of the adsorption parameters by various modified adsorbents are summarised in Table 2.9 and 2.10.

Due to the drawbacks of other treatment technologies discussed previously, adsorption is regarded as the most appropriate for removing heavy metals (Fu and Wang, 2011). The adsorption process is very effective, especially when removing low concentrations of heavy metal pollutants from the water matrix, where common technologies are either economically unfavourable or technically complicated (Barakat, 2011). In addition, this process is also an effective treatment for wastewater because the adsorbents utilised have good adsorption capacity, high selectivity, and the ability to be regenerated (O'Connell *et al.*, 2008b; Wang and Chen, 2009). Several adsorption studies have shown that the adsorption process is very efficient in removing heavy metals from the water matrix without releasing any harmful by-products (from the adsorbent) into the treated water (Shukla and Pai, 2005a; Wong *et al.*, 2003; Yu *et al.*, 2013b).

Adsorbent	Heavy metal	рН	Initial concentration (mg/L)	Contact time	Temperature (°C)	Adsorption capacity (mg/g)	References
Black cumin	Pb(II)	2.0-6.0	20	60 min	20-50	7.8	(Bingol et al., 2012)
Calledaa	Cd(II)					562.7	
bydrogel	Pb(II)	2.0-6.0	200-2000	6 h	30	825.7	(Zhou et al., 2012)
nyuroger	Ni(II)					380.1	
Fungal	Pb(II)					0.103	
Trichoderma	Cd(II)	2.0-6.0	20-100	60 min	20-40	0.139	(Singh et al., 2010)
viride	Cu(II)					0.360	
Lemon peel	Pb(II)	2.0-8.0	100-600	25 h	28	277.78	(Thirumavalavan <i>et al.</i> , 2010)
Moringa	Cd(II)	2000	10.40	100 min	20.40	171.37	$(\mathbf{D}_{\mathbf{add}\mathbf{r}}, 2012)$
oleifera leaves	Cu(II)	2.0-9.0	10-40	100 min	20-40	167.90	(Reddy, 2012)
	Cd(II)						(Thimmereleven of
Orange peel	Cu(II)	2.0-9.0	100-600	20 min	25	227.27	(1) mrumavalavan et $al = 2010$)
	Ni(II)						, 2010)

Table 2.9: Adsorption parameters of various adsorbents

Adsorbent	Heavy metals	Modifying agents	Adsorption capacities (mg/g)	References
Fruit peels: Lemon peel	Cu(II) Zn(II) Cd(II) Pb(II)	Oxalic acid Sodium hydroxide	227.27 196.08 172.41 204.08	(Thirumavalavan <i>et al.</i> , 2010)
Konjac glucomannan (polysaccharide)	Cu(II) Pb(II)	Methyl acrylate Methyl methacrylate	64.5 191.3	(Liu et al., 2009)
Moringa oleifera leaves	Cu(II) Cd(II) Ni(II)	Citric acid Sodium hydroxide	167.90 171.37 163.88	(Reddy, 2012)
Palmate-tuber salep	Zn(II) Cu(II)	Chlorosulphonic acid- dimethylformamide	1156 594	(Pourjavadi et al., 2013)
Pine cone powder	Cu(II)	Sodium hydroxide	26.32	(Ofomaja <i>et al.</i> , 2010a)
Spent grain	Cu(II) Pb(II) Zn(II) Cd(II)	Citric acid Sodium hypophosphite	104.13 293.30 232.10 296.61	(Li <i>et al.</i> , 2010)
Sugarcane bagasse	Cu(II) Cd(II) Pb(II)	1,3-diisopropylcarbodiimide acetic anhydride	92.6 149.0 333.0	(Karnitz et al., 2009)

Table 2.10: Summary of the modified biomass as adsorbent for the removal of heavy metal ions from the water matrix

2.4 TRADITIONAL MODELLING PROCEDURE

Adsorption is a complex process dependent on various parameters and outputs, which requires a large number of experiments to investigate the relationship between those factors and the process performance output (Ranjan *et al.*, 2011). Traditionally, optimisation of an adsorption process has been performed by applying one factor at a time to an experimental response, where the other factors remain constant (Bezerra et al., 2008). This method is known as one variable at a time (OVAT). The disadvantages of this method are that it is time consuming and requires a large number of experiments, which lead to an increase in expense and in the utilisation of reagents and materials (Bashir et al., 2015). Moreover, OVAT does not take account of interactions between the selected factors and does not describe the complete effects of those factors on the response and process performance. Thus, to overcome this difficulty, factorial experimental design can be employed to optimise the conditions of adsorption of heavy metals from a water matrix. Response Surface Methodology (RSM) and Artificial Neural Network (ANN) modelling are methods that are applied extensively in industry for the optimisation of process design parameters (Geyikci et al., 2012; Witek-Krowiak et al., 2014; Ye et al., 2014).

Although RSM and ANN are widely used in the study of adsorption processes, studies on Cu(II) removal from real wastewater samples tend to focus on one single parameter at a time (Pereira *et al.*, 2009; Saiano *et al.*, 2005). For example, the potential of amine-functionalised SBA-15 as an adsorbent to remove Cu(II) ions from river water, tap water and electroplating wastewater (Da'na and Sayari, 2012), and the potential of *Ulothrix Zonate* algae to remove Cu(II), Pb(II) and Cd(II) from industrial wastewater (Malakootian *et al.*, 2011) have focused only on one single parameter at a time. However, the adsorption capacity and selectivity in both studies were investigated through batch kinetic experiments, and Langmuir and Freundlich models were used to describe the equilibria between metal ions and adsorbent. Therefore, the adsorption of copper from real wastewater samples was studied; this is rare in the existing literature, but provides valuable insights into the applicability of the approaches tested in this thesis.

2.5 MATHEMATICAL MODELLING OF THE ADSORPTION PROCESS

2.5.1 RESPONSE SURFACE METHODOLOGY (RSM)

2.5.1.1 Theory and steps for RSM application

RSM is a practical method for studying the effects of multiple parameters or variables that influence the process response, by varying them simultaneously and thus reducing the number of experiments required. This method was developed by Box and Wilson, and has been widely used as a technique for experimental design (Box and Wilson, 1951). The RSM approach can be divided into six stages: (1) selection of independent variables and possible responses; (2) selection of experimental design strategy; (3) execution of experiments and obtaining results; (4) fitting the model equation to experimental data; (5) obtaining response graphs and model verification (ANOVA); and (6) determination of optimal conditions.

2.5.1.2 Factorial experimental design

The mechanism of adsorption in water treatment is a complex chemical process due to the interaction of various parameters and variation in wastewater composition (Ebrahimzadeh *et al.*, 2012). Thus, before applying these models, it is important to select the most important independent variables and their ranges, which cover the realistic conditions in a wastewater treatment environment. The independent variables, such as pH, temperature, sorbent dosage and solute concentration are the ones that usually influence adsorption performance. After the independent variables have been properly selected via screening or scoping experiments, several design methods can be applied for adsorption optimisation.

2.5.1.3 Choice of experimental design

Among the various experimental designs, it was found that two common designs, central composite design (CCD) and Box-Behnken design (BBD), have frequently been used for the final optimisation of desired processes (Turan *et al.*,

2013b; Zolgharnein *et al.*, 2013a). Figure 2.5 (a and b) shows representations of BBD and CCD for three-variable optimisation, respectively.



Figure 2.5: Experimental design based on the study for the optimisation of three variables (a) Box-Behnken design and (b) central composite design (Witek-Krowiak *et al.*, 2014).

A comparative study of BBD and CCD has been carried out on the adsorption of Pb(II) by *Robinia* tree leaves, which concluded that the CCD model is more efficient and dependable (Zolgharnein *et al.*, 2013a). Furthermore, within the field of adsorption, there are many examples where CCD has been applied to discover an appropriate model, leading to the successful prediction of optimum conditions for maximum removal of heavy metals from a water matrix (Azila *et al.*, 2008; Sugashini and Begum, 2013; Zolgharnein *et al.*, 2013b). Therefore, in this study, CCD was selected because it has better predictive capabilities and has been applied extensively in adsorption studies (Bingol *et al.*, 2012; Shanmugaprakash and Sivakumar, 2013).

CCDs are used to fit the low and high values of the parameters determined in the experiment to $-\alpha$, -1, 0, +1, and $+\alpha$ in terms of coded factors. Although the optimisation and modelling of biosorption is a recognised method in biosorption research literature, examples of building a model that covers the realistic conditions in a wastewater treatment environment are limited. Cao *et al.*, (2014) presented the effects of temperature (10–50 °C) and showed that the adsorption of Cr(VI) increases with increasing temperature, even when the range is larger than would be expected in a wastewater treatment environment, and features other

unrealistic conditions. Moreover, the range of pH (1.3–8.7) applied for the optimisation of Cd(II) uptake using marine algae is also not representative of the actual conditions in a WWTP (Ghorbani *et al.*, 2008). Therefore, the range of parameters in this study was decided based on the literature reports and the scoping experiments, which in turn are based on those used in industrial design and environmental regulations.

2.5.1.4 Evaluation of fitted model

After carrying out the experiments according to the selected experimental matrix, the results obtained are fitted to a mathematical equation to describe the behaviour of the response. The response can be displayed as a three-dimensions (3D), or as a contour, plot. This graphical representation of modelling results is the fastest way for a single response and if optimal response is within experimental boundaries (Bashir *et al.*, 2015).

The more reliable and efficient method to evaluate the quality of the developed model is by the application of analysis of variance (ANOVA). ANOVA is a set of statistical methods that are used to identify the significance of individual factors and to inform if the experimental results are meaningful (Anupam *et al.*, 2011). Furthermore, ANOVA also offers the sequential F-test, lack of fit test, p-value, significance of regression, and diagnostic plots, such as predicted versus actual plot and normal plot of residuals (Azila *et al.*, 2008). A model will be well fitted to the experimental data if it presents a significant regression and a non-significant lack of fit. Most variation related to residuals is due to pure error and not to the lack of fit, which is directly related to the model quality (Barker and Milivojevich, 2016).

2.5.1.5 Optimisation of adsorption process using RSM

The RSM approach, along with CCD, has been used to provide the significant factors, modelling and optimisation of various adsorption processes. The majority of processes used batch experiments in their studies of the independent variables and possible responses. For example, Savic *et al.*, (2012) declared that CCD is the most commonly used RSM design. Ghosh *et al.*, (2013) applied RSM with CCD

to investigate the removal of Cu(II) from aqueous solution using modified orange peel, and their study showed that pH, sorbent dosage and initial metal ion concentration influenced the adsorption process. The adsorption of Tl(I) with modified *Ulmus carpinifolia* tree leaves has also been modelled and optimised using an RSM model by Zolgharnein *et al.*, (2011) with similar independent variables.

In order to study the accuracy of different experimental designs, Zolgharnein *et al.* (2013a) applied three different experimental designs; BBD, CCD, and Doehlert, for modelling and optimisation of Pb(II) biosorption by *Robinia* tree leaves. The results obtained showed that CCD to be the most accurate design for this study, but the selected design was case-dependent and should not be considered as a rule. Furthermore, the Plackett-Burman design and CCD have been used for the optimisation of biosorption processes. Sahan *et al.* (2010) applied these two designs in his study of Cu(II) removal from aqueous solution by *Trametes versicolor*. The use of the CCD was selected as the suitable design, with a maximum quantity of removal of Cu(II) ion of 39.87 mg/g.

2.5.2 ARTIFICIAL NEURAL NETWORK (ANN)

2.5.2.1 Theoretical background of the ANN model

Artificial neural networks (ANNs) are mathematical models that predict the output based on input data without a clearly understood relationship between them. Therefore, the utilisation of ANNs in the field of adsorption processes using biomass has recently attracted interest, given the difficulty that can be encountered in fully characterising all the functionalities found in commonly used biomass (Shojaeimehr *et al.*, 2014). Although there are many well-known ANN types, such as multilayer perceptron, radial basis function networks, linear networks, Bayesian networks and Kohonen networks, currently the most popular network architecture is multilayer perceptron (MLP) (Savic *et al.*, 2012). This network consists of three or more layers of neurons with one hidden layer, and is commonly applied in the performance prediction of many processes, as shown in Figure 2.6 (Pilkington *et al.*, 2014; Witek-Krowiak *et al.*, 2014).



Figure 2.6: Architecture of the developed artificial neural network (ANN)

2.5.2.2 Application of ANN in process optimisation and modelling of adsorption processes

This approach has been widely used by many researchers for modelling the adsorption process involving different sorbents and sorbates. ANNs have been successfully used to model the biosorption of Pb(II) by black cumin (Bingol *et al.*, 2012), the removal of fluoride by bone char (Tovar Gomez *et al.*, 2013), and the removal of Cu(II) by sunflower shells (Oguz and Ersoy, 2010). Furthermore, these optimisation studies did not focus only on the removal of heavy metals from aqueous solutions, but also from leachate. Turan and his co-workers studied the potential combinations of liner materials (zeolite, bentonite and pumice) and evaluated the potential of these adsorbents for the removal of Cu(II) and Zn(II) from industrial leachate. They suggested that ANN topology was found to be effective in modelling the experimental design, and used it to display the significance levels of the analysed liner materials on removal efficiency (Turan *et al.*, 2013a).

Besides heavy metals, this approach has also been utilised in the study of adsorption of dyes and organic compounds (Witek-Krowiak *et al.*, 2014). A study by Aghav *et al.* (2011) worked on the performance of multiple adsorbents (activated carbon, wood charcoal and rice husk ash) in the competitive adsorption of organic compounds from aqueous solution. They found that the ANN technique based on the Levenberg-Marquardt (LM) training was successful when

applied to the prediction of organic compound uptake in competitive adsorption processes.

The ANN approach can not only be applied in batch adsorption process, but can also be used for the prediction of adsorption in fixed-bed and packed-bed adsorption systems. Texier *et al.* (2002) presented the ability of ANN in predicting the biosorption of lanthanide ions from aqueous solution in a fixed-bed system using cells of *Pseudomonas aeruginosa* immobilised in polyacrylamide gel. They found that the analysis of the performance of a model based on ANN showed a low divergence between predicted and experimental data. Cavas *et al.* (2011) compared ANN to a Thomas model in the modelling of methylene blue adsorption by dead leaves of beach waste *Posidonia oceanica* (L). The results of the investigation showed that both ANN and Thomas models led to similar conclusions, which confirmed that the ANN model provided satisfactory predictions for the fixed-bed adsorption of methylene blue from aqueous solution.

2.5.3 APPLICATION OF RSM AND ANN IN WASTEWATER TREATMENT PLANT PROCESS OPTIMISATION

The complexity and variability of wastewater is difficult to model and simulate using traditional modelling procedures. Because of the interaction between a numbers of adsorption variables/factors, the resulting relationships are highly non-linear and require a large number of experiments. This has placed increasing demands on both research and process optimisation, and has resulted in the increased use of RSM and ANN modelling tools. The statistical aspects of RSM and ANN enable the identification of factors that have a significant effect on the adsorption process and are able to provide a large amount of knowledge from a small number of experimental runs.

RSM is an efficient tool to predict the best performance conditions with a minimum number of experiments. It has also been effectively and widely applied in water and wastewater treatment optimisation, such as of textile dye wastewater, tannery wastewater, industrial paint wastewater, landfill leachate, and palm oil mill effluent. Moreover, removal chemical oxygen demand (COD), biological oxygen demand (BOD), colour and nitrate were also optimised via

both RSM and technological treatment. Bashir *et al.* (2010b) found that ion exchange treatment with RSM application not only showed maximum removal of COD and colour, but also removed turbidity from landfill leachate.

ANN has been successfully employed in environmental engineering, due to its superior ability to learn and classify data and its reliable and robust characteristics in capturing the non-linear relationships of variables in a complex system such as an adsorption process. An ANN model was developed by Krishna and Sree (2013) to predict the removal efficiency of Cr(VI) from aqueous solution using coir powder as adsorbent. They found that the model and the test data showed a high R² value (0.992), and the ANN model successfully tracked the non-linear behaviour of percentage removal of Cr(VI) versus independent variables, with low relative percentage error. Oguz and Ersoy (2010) studied the feasibility of sunflower shell for the removal of Cu(II) from aqueous solution in a fixed-bed adsorption column with an ANN approach. They noted that ANN effectively predicted the removal efficiency of Cu(II) using sunflower shell as adsorbent.

Moreover, ANN is also a reliable model for predicting the performance of WWTPs and in forming a basis for controlling the operation of the process. It is used as a valuable performance assessment tool for plant operators and decision makers. A study by Nasr *et al.* (2012) signifies that an ANN can effectively predict plant performance and act as an efficient analysis and diagnostic tool to understand and stimulate the non-linear behaviour of the plant. Tables 2.11 and 2.12 summarise various studies of wastewater treatment performed using conventional and RSM methods for the purpose of treatment process parameter optimisation.

2.5.4 COMPARISON OF RSM AND ANN MODELS

Interestingly, most of the previous literature has focused its attention on adsorption studies by using either RSM or ANN, without comparing the performances of these two models. Furthermore, the testing of both RSM and ANN, using new sets of experiments not belonging to the training data set, has only been undertaken in a limited number of studies on biomass adsorption, and without consideration of how the additional experiments represent the system and give a more accurate indicator of performance (Ghosh *et al.*, 2013; Saha, 2013). Therefore, model suitability for interpolated and extrapolated experimental parameters was tested. This is rare in the existing literature, but provides valuable insights into the applicability of the approaches tested in this work. The performance of the ANN and RSM models were statistically evaluated using a continuous error metric, such as the coefficient of determination (R²), absolute average deviation (AAD), and root mean squared error (RMSE). The AAD and RMSE are defined as follows (Geyikci *et al.*, 2012):

$$AAD = \left(\frac{1}{n}\sum_{i=1}^{n} \left(\frac{y_p - y_e}{y_e}\right)\right) \times 100$$
(2.5)

$$RMSE = \left(\frac{1}{n}\sum_{i=1}^{n} (y_p - y_e)^2\right)^{1/2}$$
(2.6)

Where n is the number of points, y_p is the predicted value and y_e is the experimental value. For adsorption studies, the majority of researchers measure the performance prediction of both models by using more than one error metric (Pilkington *et al.*, 2014; Shojaeimehr *et al.*, 2014), resulting in a more trustworthy evaluation.

Treatment process	Wastewater type	Independent variables	Response	References
Ion exchange	Landfill leachate	Anionic dosage, pH, reaction time, shaking speed	Removal efficiency (%)	(Bashir <i>et al.</i> , 2010b)
Ion exchange	Landfill leachate	Cation dosage, reaction time, shaking speed	Removal efficiency (%)	(Bashir <i>et al.</i> , 2010a)
Electrochemical oxidation	Dairy wastewater	Current density, NaCl dosage, electrolysis time, pH	Removal efficiency of COD (%)	(Aleboyeh <i>et al.</i> , 2008; Kushwaha <i>et al.</i> , 2010)
Adsorption	Textile dyes	Initial dye concentration, pH, temperature	Adsorption capacity (mg/g)	(Santos and Boaventura, 2008)
Fenton oxidation	Landfill leachate	pH, reaction time, initial concentration of H ₂ O ₂ , ferrous ion concentration	COD and colour removal efficiency (%)	(Mohajeri <i>et al.</i> , 2011)
Upflow anaerobic sludge blanket (UASB) bioreactor	Petroleum refinery effluent	Hydraulic retention time (HRT) influent COD, upflow velocity	COD removal, rate of biogas production	(Rastegar <i>et al.</i> , 2011)

	Table 2.11: Variou	is applications	s of RSM in	wastewater	treatment	process o	ptimisati	on
--	--------------------	-----------------	-------------	------------	-----------	-----------	-----------	----

Treatment process	Wastewater type	Independent variables	Response	References
Adsorption (Dye)	Textile dyes	Initial dye concentration, pH, temperature	Adsorption capacity (q)	(Saha, 2013)
Adsorption (Organic compound)	Clean water	pH, sorbent dosage, temperature	Adsorption capacity (q)	(Aghav et al., 2011)
Adsorption (Heavy metal)	Landfill leachate	Different combinations of liner materials	Removal efficiency (%)	(Turan <i>et al.</i> , 2013a)
Fenton process	Antibiotic aqueous solution	Reaction time, H_2O_2/COD molar ratio, H_2O_2/Fe^{2+} molar ratio, pH, COD concentration	COD removal	(Elmolla <i>et al.</i> , 2010)
Decolourisation	Acid orange 52 dye solution	Dye concentration, pH, H ₂ O ₂ concentration, temperature, reaction time	Removal efficiency (%)	(Aleboyeh <i>et al.</i> , 2008)

Table 2.12: Various applications of ANN in wastewater treatment process optimisation

2.5.5 RSM AND ANN ADVANTAGES AND LIMITATIONS

Recently, response surface methodology (RSM) and artificial neural network (ANN) methods have been used together for both modelling and optimisation applications in wastewater treatment and environmental studies (Antonopoulou *et al.*, 2012; Pakravan *et al.*, 2015). Generally, by applying these models, the number of experimental trials is reduced, which requires the evaluation of multiple parameters and their interactions. Furthermore, it is less laborious and time consuming than the conventional 'one variable at time' (OVAT) approach (Witek-Krowiak *et al.*, 2014). Factorial experimental designs such as CCD and BBD provide more information per experiment than OVAT approaches. DOE allows the identification of interactions among experimental variables within the range studied, providing better knowledge of the process and hence reducing research time and costs (Podstawczyk *et al.*, 2015).

ANNs are algorithms that can be used to perform nearly all types of nonlinear statistical modelling and provide a number of advantages, while RSM is suitable only for quadratic estimations (Ghosh *et al.*, 2015). ANN is a simple nonlinear model that is easy to use and to understand compared to other statistical methods. This model requires less formal statistical training, is able to implicitly detect complex nonlinear relationships between dependent and independent variables, to detect interactions between the variables, and to determine the availability of multiple training algorithms (Shanmugaprakash and Sivakumar, 2013). Moreover, ANN works well for large data sets and reduces drastically the processing time compared to other models.

However, ANN is also known as a 'black box', the development of which is mainly a trial and error process, and which is poor in interpreting the relationship between input and output, and in handling uncertainties (Meireles *et al.*, 2003). Thus, the calculated model can only be used within the experimental range and cannot be used for extrapolation. Furthermore, it is believed that an ANN model requires a larger number of experiments for training to build an efficient model than does RSM (Bezerra *et al.*, 2008). There is also no exact method in order to determine the minimum number of experiments for ANN training (Witek-

Krowiak *et al.*, 2014). Therefore, it is troublesome while designing the experiments. However, with scoping experiments and realistic conditions in real WWTPs, ANN can also work well with less data, if that data is well distributed in the design. Thus, the experimental data (20 CCD experiments) of RSM should be sufficient to build an effective ANN model.

2.6 CELLULOSE AS A SOURCE MATERIAL

Anselm Payen was the first to coin the term 'cellulose' for a substance with a molecular formula $C_6H_{10}O_5$ (Payen, 1838). After the identification of cellulose by Payen, it has been revealed that cellulose exists not only as glucose residues but interestingly, the glucose units are covalently linked to each other forming long molecular chains (Zugenmaier, 2008). As it can be obtained from various sources (trees, algae, fungi, tunicates and bacteria), cellulose is an almost unlimited, sustainable raw material with an estimation of more than 10^{11} tons synthesised each year on earth (Brown, 2004). Cellulose in the form of wood and plant fibres has been used for building materials, clothing, paper and as a promising renewable energy source (Welker *et al.*, 2015).

The development of adsorption technology is for a large part focused on the development of the most efficient adsorbent. Cellulose is one such adsorbent that has been investigated for the adsorption of heavy metals (Alila and Boufi, 2009; Isobe *et al.*, 2013; Karnitz *et al.*, 2009). It is the most abundant natural, renewable, and biodegradable polymer and as a raw material for the preparation of various functional polymers is available at relatively low cost (Eyley and Thielemans, 2011; Morandi *et al.*, 2009).

2.6.1 STRUCTURE AND PROPERTIES OF CELLULOSE

Chemically, cellulose is a polymer of glucose (β -1,4-glycan). Natural cellulose materials consist of a mixture of crystalline and amorphous regions (Kasnejad *et al.*, 2012). Moreover, the molecular structure of cellulose also provides the characteristics of hydrophilicity, chirality, and degradability (O'Connell *et al.*, 2008a). Cellulose is a linear polymer of repeating β -D-glucopyranose units linked by acetal functions between hydroxyl groups on C-1 and C-4 (Oksman *et al.*,

2006). The repeating unit in a cellulose polymer chain, known as cellobiose, is stabilised in the chain direction by two hydrogen bonds. The C-4 hydroxyl at the end with a closed ring structure is known as the non-reducing end, whereas the opposite end, the C-1 hydroxyl, is known as the reducing end. The presence of three reactive hydroxyl groups, at the C-2, C-3 and C-6 positions; the alcohols at positions C-2 and C-3 are secondary alcohols while the alcohol at the C-6 position is a primary alcohol (Klemm *et al.*, 2004). These reactive hydroxyl groups are responsible for the reactivity of cellulose. The molecular structure of cellulose is illustrated in (Figure 2.7).





2.6.2 SURFACE MORPHOLOGY

There are several allomorphic forms of crystalline cellulose which are denoted as cellulose I, II, III, and IV with the possibility of conversion from one form to another (Eichhorn *et al.*, 2009). The different allomorphic forms of crystalline cellulose have a great influence on the chemical pre-treatment and modification processes (Ciolacu *et al.*, 2011). Cellulose I is the most common form found in nature, known as 'natural' cellulose, and is composed of a mixture of two related crystalline forms, namely cellulose I α and I β (Atalla and Vanderhart, 1984).

Cellulose I has attracted the interest of the scientific community in it attempts to elucidate its crystal structure (Saito and Isogai, 2004; Siqueira *et al.*, 2010). Cellulose I α is found in algae and bacteria, while cellulose I β is dominant in higher plants and animals (Atalla and Vanderhart, 1984). Cellulose I can be converted to cellulose II through treatment with sodium hydroxide (Moon *et al.*, 2011). New allomorphs, cellulose III_I and cellulose III_{II} are formed when cellulose I or cellulose II are exposed to liquid ammonia. In addition, cellulose

 IV_I and IV_{II} can be derived from cellulose III_I or cellulose III_{II} through heat treatment in glycerol (Dufresne, 2012).

2.6.3 SOURCES OF CELLULOSE

Lignocellulosic materials refer to plants that consist of cellulose, hemicellulose and lignin. In order to obtain the cellulose from these sources, the unwanted lignin, hemicellulose and other extractives are removed through chemical pulping. A wood fibre, which contains approximately 40% cellulose, is the most abundant biomass resource on earth due to the availability of global wood supplies all year round (Moon *et al.*, 2011). Non-woody materials such as plant waste also serve as substitute cellulose sources. Although woody plants could contribute a large amount of cellulose, the need for environmentally friendly processes avoiding deforestation has encouraged researchers to examine plant waste as a new alternative (Ohwoavworhua and Adelakun, 2010).

Besides plants, cellulose can also be synthesised from non-plant materials such as bacteria, algae, yeast and fungi (Wang and Chen, 2006). The cell wall of microorganisms is composed of a network of cellulose fibrils, which serve as a structural support. Chen *et al.*, (2010) reported that bacterial cellulose from bacteria of genus *Acetobacter* has a molecular structure similar to plant cellulose and has been investigated in many fields such as medical, food manufacture and paper production. In addition, bacterial cellulose has also been used as an adsorbent to remove heavy metals from aqueous solution (Chen *et al.*, 2009; Oshima *et al.*, 2008).

Of all the sources, cotton is the purest, containing 90 - 99% cellulose (Schäfer *et al.*, 2005), and was thus used as starting material in this study. However, cotton requires further processing and purification through chemical treatments to produce a form of purified cellulose such as bleached cellulose pulp (Sjöström, 1993).

2.7 CELLULOSE NANOWHISKERS AS POTENTIAL ADSORBENT (AS AN EXAMPLE CASE)

A nanoparticle is a microscopic particle with at least one dimension less than 100 nm. The properties of materials at this scale are of great scientific interest, due to the high surface area to bulk ratio (Bondeson *et al.*, 2006). In nature, the cellulose microfibril structure consists of crystalline and amorphous regions, as shown in Figure 2.8. There are different terminologies to describe the crystalline rod-like nanoparticle: these include nanowhiskers, nanocrystals, nanoparticles and even microcrystallites (Siqueira *et al.*, 2010). 'Cellulose nanowhiskers (CNWs)' will be used throughout this thesis.

A study pioneered by Rånby described the production of a suspension of CNWs from different types of cellulose using strong mineral acids (Gezelius and Rånby, 1957). In this process, the acid attacked the disordered amorphous region preferably, leaving the crystalline region relatively untouched (procedure described in Section 3.3.1). The cellulose chains in the amorphous regions are randomly oriented in a spaghetti-like arrangement, leading to a lower density with more free volume than the crystalline region, making it susceptible to attack by acids (Eyley and Thielemans, 2014). In general, acid hydrolysis of native cellulose causes a rapid decrease in its geometrical dimensions. By controlled acid hydrolysis, rod-like CNWs are produced. Therefore, this process will be used to produce an adsorbent which offers a combination of nano-dimensions with high surface area and provides great potential for a new and green route to solve current heavy metal contamination problems (Cao *et al.*, 2012).



Figure 2.8: Schematic of amorphous and crystalline cellulose (Bhattacharya *et al.*, 2008)

Several strong acids have been used in attempts to hydrolyse cellulose. A study by Araki *et al.*, (2001) has shown the effects of sulphuric and hydrochloric acid to produce stable suspensions of CNWs. The study reported that sulphuric acid hydrolysis provided more stable aqueous suspensions of nanowhiskers than did hydrochloric acid. It has been observed that the use of sulphuric acid for the preparation of CNWs causes the formation of negatively charged sulphate esters on the surface of CNWs, resulting in electrostatic stabilisation of the suspension of these nanowhiskers (Jiang *et al.*, 2013). Furthermore, sulphuric acid as compared to hydrochloric acid is more commonly used for the hydrolysis due to its cost effectiveness and its extensive use in industry (Bhattacharya *et al.*, 2008). The mechanism of acid hydrolysis is shown in Figure 2.9.



Figure 2.9: Acid hydrolysis mechanism

There are several conditions that have a profound effect on the hydrolysis process, such as acid concentration, reaction time, and temperature. Dong *et al.*, (1998) investigated the effect of temperature, reaction time and duration on the subsequent ultrasonic properties of CNWs. It was reported that increasing hydrolysis time decreased the whiskers length and increased the total sulphur

content and surface charge of the nanowhiskers. In order to optimise the isolation of CNWs, Bonderson *et al.*, (2006) selected five independent parameters that had the greatest influence over the dimension and yield of whiskers. From the results, it has been observed that extended hydrolysis time and high acid concentration lead to reduction in the length of CNWs.

2.7.1 STRUCTURE AND PROPERTIES OF CNWS

As mentioned in the previous section, the dimensions of CNWs produced by the acid hydrolysis process are dependent on the origin of cellulose and the process conditions. The morphologies of CNWs have become a topic of interest and have been studied by several researchers (Landry *et al.*, 2011; Lu *et al.*, 2006). Non-wood materials such as algae and tunicate yield much longer nanowhiskers than do wood microfibrils, with lengths in the hundreds of nanometres (Beck-Candanedo *et al.*, 2005; Liu *et al.*, 2014a). Normally, the width of nanowhiskers is a few nanometres while the length is in the hundreds of nanometres. The schematic of hierarchical structure of wood from tree to nanoscale is shown in Figure 2.10.



Figure 2.10: The schematic of hierarchical structure of a wood (Postek *et al.*, 2011)

CNWs from wood are reported to be 3 - 15 nm in width and 100 - 200 nm in length; tunicate, a sea animal, can yield whiskers of 1000 - 2000 nm in length and 10 - 20 nm in width, while the average dimensions of CNWs produced from cotton are approximately 4 - 10 nm in diameter and 100 - 300 nm in length (Dong *et al.*, 1998; Dufresne, 2006).

2.7.2 MODIFICATION OF CELLULOSE NANOWHISKERS

Chemical modification of cellulose by grafting of functional groups has the potential to improve its adsorption capacity and to enhance its performance under desired conditions (O'Connell *et al.*, 2006a; O'Connell *et al.*, 2006b). Although the modification process will cause a decrease in biomass, many researchers have focused more on the adsorption capacity and efficiency than on overall mass yield (Karnitz *et al.*, 2007; Pereira *et al.*, 2009; Xu *et al.*, 2011). The modification enhances not only the adsorption capacity but also other properties of cellulose, such as its hydrophilic or hydrophobic characteristics, elasticity, water sorbency, and thermal resistance (McDowall *et al.*, 1984).

Carboxylic acid groups are one example of a functional group that can be introduced onto the cellulose surface by TEMPO (2,2,6,6-tetramethylpiperidine-1-oxyl)-mediated oxidation; this oxidised cellulose adsorbent has been shown to be capable of adsorbing 465.1 mg/g Pb(II) from aqueous solution (Yu *et al.*, 2013b). TEMPO-mediated oxidation has frequently been used to introduce carboxyl functional groups onto the surface of nanowhiskers or native cellulose without affecting the crystallinity or changing the original fibrous morphology (Saito *et al.*, 2005).

Besides TEMPO-mediated oxidation, the carboxyl functional groups can also be introduced through esterification, which may increase the number of carboxylic acid functionalities on the cellulose surface, as the secondary hydroxyl groups can also be converted to carboxylates, thus increasing the adsorption capacity of the oxidised adsorbent. For example, CNWs were chemically modified with succinic anhydride to obtain carboxylated CNWs. However, the reported modification process using succinic anhydride as an active agent was time consuming and not very sustainable, as the process required 12 hours under pyridine reflux in order to obtained the final modified adsorbent (Yu *et al.*, 2013b). Moreover, pyridine is well known for causing problems in the chemical industry and is avoided as much as possible as it may cause harmful health effects (Health, 1992). In addition, refluxing in pyridine, where pyridine vapour is generated, is not considered sustainable (Xu *et al.*, 2015). Therefore, oxidised CNWs via TEMPO-mediated oxidation have been identified as a more suitable adsorbent for remediation of copper from water matrices.

2.8 CHARACTERISATION OF CNW ADSORBENTS

In characterising the adsorbent, several analytic techniques have been used prior to surface modification. It is important to study the changes observed in CNWs after TEMPO-mediated oxidation. The success of modifications can be judged by comparing the characterisation results before and after modification. Techniques that have been used for characterisation include: (1) Fourier Transform Infrared Spectroscopy (FTIR); (2) zeta potential; (3) transmission electron microscopy (TEM) and scanning electron microscopy (SEM); and (4) Brunauer, Emmett and Teller (BET) analysis.

2.8.1 FOURIER TRANSFORM INFRARED SPECTROSCOPY (FTIR)

IR spectroscopy is widely used in determining functional groups and in ascertaining the chemical structure of organic molecules. Theoretically, when IR radiation is passed through a sample, energy is absorbed, causing a change in the vibrational and rotational energy states of the molecules. An IR spectrometer will then measure the relative amount of energy absorbed, as a function of its wavelength or frequency. Normally, the wavenumbers of organic compounds are located between 400-4000 cm⁻¹ (Silverstein *et al.*, 2014). Since each chemical group absorbs a specific frequency, IR spectrometer can then identify the chemical bonds and molecular structure of a material (Settle, 1997).

2.8.2 ZETA POTENTIAL

Zeta potential analysis is a technique to determine the colloidal stability of nanoparticles in solutions. This analysis is important in understanding the state of the nanoparticle surface in predicting the long-term stability of nanoparticle, and to assess the surface charge of CNWs and oxidised CNWs. The zeta potential of suspended particles is calculated by using the mathematical model of electrophoretic mobility (μ) (Kaszuba *et al.*, 2010):

$$\zeta(mV) = \frac{3\mu\eta_o}{2f_{\kappa_a}\varepsilon} \tag{2.7}$$

Where ζ is the zeta potential; η_o is the medium viscosity; ε is the medium permittivity; and f_{κ_a} = Henry's function in which f_{κ_a} = 1 for the Hückel model, or 1.5 for Smoluchowski. The model is selected based on the prevailing conditions, particle size and ionic strength (Greenwood and Kendall, 1999). κ is termed as the Debye length is the reciprocal length and κ^{-1} is often taken as a measure of the thickness of the electrical double layer, while 'a' refers to the radius of the particle and thus κ_a measures the ratio of the particle radius to electrical double layer thickness. This is illustrated schematically in Figure 2.11.



Figure 2.11: Schematic illustrating Smoluchowski and Hückel's approximations used to convert from electrophoretic mobility into zeta potential
2.8.3 ELECTRON MICROSCOPY (SEM AND TEM)

In electron microscopy, an electron beam is focused onto a small area of a specimen. Images are formed either by electrons passing through a thin sample – Transmission Electron Microscopy (TEM) – or by secondary emission from a thick sample –Scanning Electron Microscopy (SEM) (Goldstein *et al.*, 2012). Both SEM and TEM are commonly used to examine the surface morphology and chemical composition of the nanomaterial samples. Although similar electron gun and lenses are used for both electron microscopes, SEM and TEM also exhibit some differences, which are summarised in Table 2.13.

	SEM	TEM
Electron	Based on scattered electrons	Based on transmitted
		electrons
Image	Produce the image of samples	Electron are directly pointed
	after the microscope collects	towards the sample
	and counts the scattered	
	electrons	
Focus	On the sample's surface and its	On the sample's surface and
	composition	provides the details about
	-	internal composition
Dimensional	Provides 3-dimensional image	Provides 2-dimensional image
Magnification	Offers 2 million as a maximum	Up to a 50 million
	magnification level	magnification level

Table 2.13: Differences between SEM and TEM ((Goldstein et al., 2012))
---	--------------------------	---

2.8.4 BRUNAUER-EMMETT-TELLER (BET) METHOD

The BET technique is the most common method for the precise specific surface area evaluation of materials by isotherm nitrogen adsorption analysis (Ishizaki *et al.*, 2013). Moreover, this analysis provides information about the porosity and pore size, which are very important for nanomaterial adsorbents. The Brunauer-Emmett-Teller (BET) multipoint technique (Brunauer *et al.*, 1938; Rouquerol *et al.*, 2013) was applied for surface area determination and the Barrett, Joyner and Halenda (BJH) method (Barrett *et al.*, 1951) for pore diameter analysis. The method of determining these values involves studying the adsorption and desorption of nitrogen gas to and from the surface of the solid at liquid

temperature and a relative vapour pressure of 0.05–1, following the BJH model. The BET surface area can be calculated by (Rouquerol *et al.*, 2013):

$$BET\left(\frac{m^2}{g}\right) = \frac{3}{\rho_s R} = \frac{6}{\rho_s D}$$
(2.8)

Where D is particle diameter, R is radius and ρ_s is the mass of the solid divided by the volume of the solid excluding open and closed pores.

The total pore volume can be calculated by assuming that the density of liquid nitrogen in the pores is the same as that of bulk of liquid nitrogen, such that (Rouquerol *et al.*, 2013):

$$TOPV\left(\frac{cm^{3}}{g}\right) = \max \ loading \ \left(\frac{cm^{3} \ STP}{g}\right) \times \left(\frac{1 \ mole}{22,414 \ cm^{3} \ STP}\right)$$
(2.9)
$$\times MW\left(\frac{g}{mole}\right) \times \frac{1}{\rho_{L}}\left(\frac{cm^{3}}{g}\right)$$

where MW and ρ_L are the molecular weight and density of the probe molecule being used, which for N₂ adsorbed at 77 K have values of 28.01 g/mol and 0.807 g/cm³, respectively.

2.9 ENVIRONMENTAL ANALYSIS

Environmental analysis is very important in order to determine the presence of contaminants in the aqueous environment. The amount of pollutants released into the environment has been increasing due to anthropogenic activities, with potential exposure of these pollutants to the ecosystem. The identification of heavy metals and the determination of their concentration range in wastewater are very important in order to meet the discharge limits of heavy metals. For wastewater treatment studies, atomic absorption spectrometry (AAS), graphite furnace atomic absorption spectrometry (GFAAS), and inductively coupled plasma optical emission and mass spectrometry (ICP-OES and ICP-MS) are the most commonly spectroscopic methods used for elemental measurements in environmental analysis of the water matrix. AAS and ICP-MS have been used throughout this thesis.

2.9.1 THEORETICAL BACKGROUND TO INSTRUMENTATION

2.9.1.1 Atomic absorption spectrometry instrumentation (AAS)

AAS is a widely used method in analytical chemistry to determine the concentration of a particular element in different samples. AAS is based on the absorption of ultraviolet or visible radiation by free atoms in the gas phase. The sample, which is in solution, is aspirated into a flame. Then, with appropriate flame conditions, the atoms in the ground state will absorb the light produced by the source lamp at a specific analytical wavelength. Hence, by measuring the amount of light absorbed by the ground state atoms, a quantitative determination of the amount of metal present can be measured (Beaty and Kerber, 1978). The single element quantitative analysis in AAS is based on Beer-Lambert's law. Lambert's Law states that absorbance of a sample is proportional to its path length, and Beer's Law states that absorbance is proportional to concentration of the element (Equation 2.8) (Perkin Elmer, 1996):

$$A = -\log\left(\frac{I}{I_o}\right) = -\log T = abc \tag{2.10}$$

Where A is the absorbance, a is the absorption coefficient, b is the length of the light path and c is the concentration of the absorbing species in the light path. The term T in the equation refers to transmittance, I_0 is the incident light intensity and I is the transmitted light intensity.

An atomic absorption instrument consists of primary light source, an atom source, a wavelength selector to isolate the specific wavelength of light required, a detector, electronics to process the data signal and a data display to show the results, as shown in Figure 2.12. A hollow cathode lamp (HCL) is normally used as the light source, with a different lamp used for each element to be determined. The atomisation unit consisting of a nebuliser, spray chamber and flame produces free analyte atoms from the sample. The heat energy from the flame, commonly in the form of an air/acetylene or nitrous oxide/acetylene flame facilitates atomisation. By adjusting the burner head, the light beam can pass through the flame and hence the sample, and is then directed onto the detector where the photons are detected and measured. The absorbance of the element in the solution is measured and the concentration of the element is then calculated from the calibration curve.



Figure 2.12: AAS schematic diagram

2.9.1.2 Inductively Coupled Plasma Mass Spectrometry (ICP-MS)

ICP-MS is considered the most efficient technique for multi-elemental analysis, and is capable of determining more than 73 elements per minute in an individual sample, depending on factors such as the concentration levels and the required precision (Thomas, 2013). An ICP-MS instrument uses argon inductively coupled plasma (ICP) as the ionisation source which is directed into a mass spectrometer (MS) to detect and determine the number of ions produced. Generally, ICP-MS provides better detection limits (0.001-1 μ g/L) compared to other spectroscopic techniques, for a wide variety of elements, and is used for its ease-of-use, robustness and speed (Perkin Elmer, 2008).

The block diagram of an ICP-MS is illustrated in Figure 2.13. In ICP-MS, the aqueous sample solutions are converted into an aerosol form by a nebuliser. Then, the aerosol transported into the plasma together with Ar gas, which usually operates at temperatures of 5000-10000 K at atmospheric pressure. However, in order to increase the precision and sensitivity of the instrument, the large droplets that may be produced by the nebuliser are removed in the spray chamber. Interface conditions exist in the ICP and MS system, namely atmospheric pressure and a vacuum environment (Wang *et al.*, 2006).

Consequently, the sample ions in the interface will flow into the MS system at high speed and the plasma expands in the vacuum. The high vacuum conditions in the MS system results in an efficient environment to prevent collisions between sample ions and air molecules (Thomas, 2013). For MS measurements, a number

of voltages must be set with respect to the type of ion optics used in order to separate the isotopes according their mass to charge ratios. Usually, a quadrupolebased mass analyser is used in ICP-MS instrumentation due to its simplicity, relatively low cost and good performance (Perkin Elmer, 2008). The ions are directed to the detector to record the number of ions per mass present.



Figure 2.13: ICP-MS block diagram (Agilent Technologies, 2010)

2.9.2 PERFORMANCE COMPARISON OF INSTRUMENTATION TECHNIQUES FOR ANALYSIS

In selecting a suitable technique for environmental analysis it is necessary to examine the particular analytical requirements. Therefore, there are several critical performance factors that need to be considered in selecting the appropriate technique for elemental determinations in environmental analysis of aqueous solutions (Perkin Elmer, 2004): detection limits, analytical working range, sample throughput and interferences.

The detection limits achievable for targeted elements are important in selecting an analytical technique for a given analytical problem. As shown in Figure 2.14, it is clear that Flame atomic absorption spectroscopy (FAAS) is inapplicable to trace and ultra-trace level analysis, while ICP-MS has the best detection limits of the other techniques listed.



Figure 2.14: Typical detection limit ranges for the major atomic spectroscopy techniques (Perkin Elmer, 2008)

Sample throughput is based not only on the number of samples that can be analysed but also on the number of elements that can be determined per unit of time. However, for most techniques, the analysis time will vary depending on factors such as concentration levels and required precision. When these two factors are not limiting, the number of elements to be determined per sample and the analytical technique chosen will determine the sample throughput. For example, FAAS shows relatively high sample throughput when analysing a large number of samples with limited numbers of elements. However, when multiple elements are to be determined, FAAS requires specific light sources and optical parameters for each element to be determined, which may cause low sample throughput. As such, FAAS is generally considered to be a single-element technique (Perkin Elmer, 2004). Therefore, a multi-element technique such as ICO-OES or ICP-MS has excellent sample throughput, which can typically determine more than 73 elements per minute in individual samples (Thomas, 2013).

Furthermore, problematic interference in determination of some elements also depends on the method being used. The determination of certain elements by FAAS may cause interference when the flame is not sufficiently hot to dissociate the molecules; this is known as chemical interference (Agilent Technologies, 2015). Interference in these atomic spectroscopy techniques can also occur due to contaminations such as humic acids, and to physical matrix effects such as variations in the physical characteristics of the sample, which may affect the accuracy of the analysis (Beaty and Kerber, 1978). The comparisons between these instrumental techniques are shown in Table 2.14.

	FAAS	GFAAS	ICP-OES	ICP-MS
Detection limit	Very good for some elements	Excellent for some elements	Very good for most elements	Excellent for most elements
Ease of use	Very easy	Skill required	Skill required	Skill required
Unattended operation	No	Yes	Yes	Yes
Capital costs	Low	Medium to high	High	Very high
Running costs	Low	Medium	High	High
Sample throughput	15 seconds per element	4 min per element	5-30 elements per min	All elements in 2-6 min
Applications	Ideal for laboratories analysing large number of samples for a limited number of elements	Ideal for laboratories analysing a limited number of elements and requiring better detection limits	Ideal for laboratories analysing multiple elements in a moderate or large number of samples	Ideal for laboratories analysing multiple elements in a large number of samples and requiring a system capable of determining trace and ultra-trace analyte concentrations.

Table 2.14: The strength and limitations of atomic spectroscopy techniques (Perkin Elmer, 2008)

FAAS= Flame Atomic absorption spectrometry; GFAAS=Graphite furnace atomic absorption spectrometry; ICP-OES-Inductively coupled plasma optical emission; ICP-MS= Inductively coupled plasma mass spectrometry

2.10 ISSUES ARISING FROM LITERATURE REVIEW

The literature has identified that though oxidised CNWs have been used as adsorbents in many studies, these studies have focused mainly on the suitability of this adsorbent for removing heavy metals under conditions unrealistic to the wastewater treatment plant. Moreover, there are no papers at present that test the stability of oxidised CNW adsorbent at different time intervals under dry conditions and in the water matrix, which are relevant from the perspective of manufacture and WWTP application respectively. In addition, only a limited number of studies have focused on the removal of copper from real wastewater samples. The issue with these studies are the assumptions that the real wastewater composition does not vary with time, is the same for each sample, or that its composition has no influence on the removal capability of the process used. Therefore, the complexity of wastewater and its variability in terms of heavy metal contamination and water quality indicators should be considered, because they may influence adsorption efficiency. Benchmark studies performed in this current study will then provide a better understanding of the impacts of wastewater composition on the remediation of Cu(II) from the matrix.

Studies reported in the literature have tended to focus on one single parameter at a time; this will basically require a longer time to determine optimum adsorption conditions. Therefore, experimental factorial design has been implemented to optimise the adsorption of Cu(II) from water matrices. Furthermore, the testing of both the RSM and ANN models, using new sets of experiments that lie both inside and outside the test parameter system, have not been reported in other papers. Generally, only a couple of parameter variations have been tested, without checking the chosen model's suitability for parameters lying within the tested parameters, and certainly not for parameters lying outside the tested parameter space. Statistical mathematical models have been used to fit experimental data within a system, comparing factors and ranges, and in some cases predictive capability, but rarely test system boundaries. The literature does not account for matrix variability as an influencing factor when evaluating potential of an adsorbent for remediating heavy metal pollutants from a wastewater matrix. In addition, no work to date has presented potential mathematical models that could

take into account the influence of matrix variability on adsorbent performance, which would provide valuable insights into the applicability of approaches tested in this thesis.

3.1 INTRODUCTION

This chapter provides the list of chemicals and reagents, preparation and modification of the adsorbent, and the adsorption process. It is divided into three sections. The first section includes the list of chemicals and reagents used for adsorbate solution and adsorbent preparation. The second section contains the experimental procedure for the preparation and modification of the adsorbent, and the characterisation of modified adsorbent. This section also describes the scoping batch studies and the analysis system used to determine the concentration of the solutions. The final section contains the procedure for batch adsorption studies by using design of experiment (DoE), model development and analysis of variance (ANOVA).

3.2 MATERIALS

The materials used are divided into three main parts, comprising the chemicals to prepare adsorbate solution, the raw materials and chemicals to prepare adsorbent, and wastewater effluent.

3.2.1 REAGENTS

All the chemical reagents used in these studies were analytical grade, with the list given in Table 3.1. Milli-Q ultrapure water (Milli-Q, 18.2 M Ω -cm resistivity, Millipore, Bedford, MA, USA) was used for the preparation, dilution and analytical purposes of solutions.

3.2.2 COTTON

Cotton wool (BP grade, 100%), which contains 95–99% cellulose, was chosen as the raw material for the production of cellulose products because of its high purity of cellulose over other cellulose sources. Moreover, cotton wool requires no treatment processes such as bleaching prior to hydrolysis, as it can be readily purchased in pure form.

Chemicals/reagents	Formula	Supplier	Purity grade	Purpose
2,2,6,6-tetramethyl-1-piperidinyloxy free radical	TEMPO	Sigma-Aldrich 98%		2
Amberlite MB6113 mixed bed ion exchange resin		Alfa Aesar		2
Copper sulphate pentahydrate	CuSO ₄ .5H ₂ O	Sigma-Aldrich	98–99%	3
Copper, standard solution 1000 mg/L	CuSO ₄	ROMIL Ltd	Analytical reagent, 1000 mg/L Cu in 2% nitric acid	3
Methanol	CH ₃ OH	Fisher Scientific (UK)	99.8%	2
Milli-Q ultrapure water		Millipore, Bedford, MA, USA	18.2 MΩ-cm resistivity	2, 3
Nitric acid	HNO ₃	Sigma-Aldrich	ACS reagent	2, 3
Potassium bromide	KBr	Fisher scientific (UK)	>99%	1
Sodium bromide	NaBr	Sigma-Aldrich	>99%	2
Sodium chloride	NaCl	Fisher Scientific (UK)	ACS reagent, >99%	1
Sodium hydroxide	NaOH	Fisher Scientific (UK)	98–99%	2, 3
Sodium hypochlorite	NaClO	Sigma-Aldrich	Reagent grade, 10–15% available chlorine	2

Table 3.1: List of reagents and chemicals

Sulphuric acid	H_2SO_4	Fisher Scientific (UK) >95%	2,3
Uranyl acetate	UO ₂ (OCOCH ₃) ₂ .2H ₂ O	Agar Scientific	1

Purpose:

1. Characterisation of adsorbent

2. Adsorbent preparation and modification

3. Adsorption process and preparation

3.2.3 WASHING PROCEDURE

The laboratory wares and sample tubes were washed with Milli-Q water followed by 20% by volume HNO₃. The washed laboratory wares and tubes were filled with 50% by volume HNO₃ and soaked for no less than 3 hours. Then the laboratory wares and tubes were rinsed with Milli-Q water 2–3 times and were dried in the oven prior to use.

3.2.4 WASTEWATER SAMPLES

The sample collection was carried out at the Severn Trent Water Stoke Bardolph wastewater treatment plant (WWTP) in Nottingham, UK, which has a population of 7.7 million. The plant, located on the west bank of the River Trent, serves most of the Nottingham area. Wastewater effluent was collected twice a week using a grab-sampling approach. Grab sampling was selected due to concerns over sample stability for stored composites; it allows the analysis of unstable parameters such as pH, dissolved oxygen and temperature (APHA, 1998).

The wastewater influent and effluent were collected twice a week in the morning from Stoke Bardolph WWTP. The samples were collected from November 2015 to March 2016. Water samples were collected in a 2.5 L amber glass bottle with little or no headspace volume to minimise aerobic biodegradation of organic substrates, and transported to the laboratory. In order to minimise contamination, the bottles were rinsed 3 times with the effluent before the samples were collected. Dissolved oxygen (DO) and temperature were measured by a DO meter (Jenway 970, Staffordshire, UK) and thermocouple thermometer (Digi-Sense, Cole-Parmer Instrument Ltd., UK), respectively at the sampling point to avoid any changes during storage and transport.

The effluents were filtered using glass microfiber paper (Fisherbrand, G261, 1.2 μ m) ready for further characterisation in the laboratory. Samples analysed for trace elements by an atomic absorption spectrometry (AAS) and inductively coupled plasma mass spectrometry (ICP-MS) analysis were filtered through 0.2 μ m surfactant-free cellulose acetate membrane syringe filter (Minisart-Plus filters, Sigma-Aldrich), acidified with nitric acid and stored in polypropylene tube

bottles. In order to minimise the potential for volatilization or biodegradation between sampling and analysis, the samples were stored in a fridge at 4 °C.

3.2.4.1 Stability of effluent

The stability of the effluent was tested at different storage times prior to analysis by ICP-MS. The effluents were first filtered through a 0.45 μ m surfactant-free cellulose acetate membrane syringe filter and stored in acid-washed sample tubes with no headspace. The samples were acidified to 2% by volume HNO₃, with pH lower than 2.0, and stored in a fridge at 4°C before analysis. Then ICP-MS analysis was carried out on the effluents stored for different storage times (24, 72 and 120 hours).

3.2.4.2 Wastewater characteristics

In order to perform a physico-chemical characterisation of the water samples, the American Public Health Association's *Standard methods for the examination of water and wastewater* were applied (APHA, 1998). In each of the samples, the following water quality parameters were determined:

- total suspended solids (TSS)
- pH
- conductivity
- total dissolved solids (TDS)
- chemical oxygen demand (COD).

3.2.4.3 Total suspended solids (TSS)

A well-mixed sample (500 ml) was filtered through a weighed standard 1.2 μ m glass fibre filter. The filter paper was placed into a filtration apparatus and the vacuum was applied. Then the residue retained on the filter was dried in the oven to a constant weight at 120°C to remove water content. The TSS was calculated by the difference between the mass of filter paper before and after the drying process (APHA, 1998):

g total suspended solids/
$$L = \frac{(A - B) \times 1000}{sample \ volume, mL}$$
 (3.1)

Where:

A = weight of filter + dried residue, g

B = weight of filter, g

3.2.4.4 pH, conductivity and total dissolved solids (TDS)

The pH, conductivity and total dissolved solids of water samples were measured by a combined pH/conductometer (Model HI 9811, Hanna Instruments Ltd., UK). Although this method is rather simple and straightforward, comprehensive care and precautions must be taken during measurement to avoid unnecessary errors. The proper steps include calibrating the meter, placing the electrodes in a well-mixed sample and reading the results directly from the meter. Thus, the device was calibrated with buffers of pH 4.0, 7.0, and 10.0 prior to usage in order to obtain accurate results. Magnetic stir plate and stir bar were used to stir the water samples at a moderate and uniform rate in order to achieve equilibrium between the electrode surface layer and the solution. Moreover, proper cleaning and storing of pH/conductivity probes is important in order to increase the accuracy of the measurements. The acceptable criteria for measured TDS (g/L) to conductivity (mS/cm) ratio are from 0.55 to 0.7 (APHA, 1998). Therefore, if the ratio is outside of these limits, measured TDS or measured conductivity is not accurate and the sample needs to be re-analysed.

3.2.4.5 Chemical oxygen demand (COD)

APHA's Standard methods for the examination of water and wastewater (1998) defines COD as the amount of dissolved oxygen required to oxidise and stabilise the given sample of water (APHA, 1998). Accordingly, cuvette tests were performed for both water samples and were analysed on the day upon receipt. The method for the determination of COD in wastewater samples using cuvette tests is based on the addition of 2 ml of sample to the cuvette, which is heated in LT200 thermostat (Hach Lange, UK) at 148 °C for 2 hours. Based on the concentration, two different cuvettes LCI500 (15 – 150 mgO₂/L) for effluent and LCI400 (150 –

1000 mgO₂/L) for influent water samples were used. After cooling, both cuvettes (sample and deionised water as control) were measured using a DR 2800 spectrophotometer (Hach Lange, UK).

3.3 EXPERIMENTAL METHODOLOGY FOR ADSORBENT PREPARATION

3.3.1 PREPARATION OF CELLULOSE NANOWHISKERS

The cellulose nanowhiskers (CNWs) were prepared in four steps: chopping, hydrolysis, dialysis, and sonication. Pure cotton wool was cut into small cubes $(5 \text{ cm } \times 5 \text{ cm})$ and the required amount of cotton wool was prepared for hydrolysis. Then the cotton wool (80 g) was dispersed in 64 wt% sulphuric acid (8.75 ml H₂SO₄ solution per gram of cotton) and the cotton pieces were gradually added using a glass rod. The suspension was held at 45 °C in the fume cupboard with a magnetic stirrer for 45 minutes to allow cotton hydrolysis. Then the suspension was diluted with cold deionised water and subjected to centrifugation. The suspension was transferred into six plastic containers with equal volume and 3 successive centrifugations were carried out at 10 000 rpm for 15 minutes, using a Sigma 6K15 Refrigerated Centrifuge. After centrifugation, the suspension was dialysed using cellulose membrane tubing (MWCO 12K-14K diameter 48 mm) against running tap water for 48 hours to remove the residual free acid in the dispersion. The suspension was then sonicated using a Branson Digital Sonifier and filtered through a No. 2 fritted glass filter to remove residual aggregates. Next, the filtrate obtained was mixed with Amberlite MB6113 mixed bed ion exchange resin to remove non H₃O⁺ cations, and filtered again before being stored in a fridge prior to use. The process flow for the production of CNWs is shown in Figure 3.1.



Figure 3.1: Process flow for CNWs production

3.3.2 PREPARATION OF SURFACE-CARBOXYLATED CNWS (OXIDISED CNWS) VIA TEMPO-MEDIATED OXIDATION

Preparation involved 400 ml of 0.5 wt% dispersion of CNWs solution (section 3.3.1), which was held at room temperature and stirred mechanically. Then TEMPO free radical (59 mg) and NaBr (645 mg) were added to the solution. The process was commenced by adding NaClO solution (10-15 % available Cl) (5.1 ml) into the CNWs solution and stirring continually at room temperature for 45 minutes. The pH of the solution was maintained at 10 ± 0.1 by adding dropwise 0.5 M NaOH, when necessary. A pH meter (Jenway 350, Staffordshire, UK) was used for monitoring the pH of the reaction. After stirring the solution for a designated time, the oxidation was quenched by adding aqueous methanol (5 ml). The solution was then dialysed with cellulose membrane tubing (MWCO 12K-14K diameter 48 mm) against running tap water for 4 hours to remove any residual free acid in the dispersion. After dialysis, the solution was mixed with Amberlite MB6113 mixed bed ion exchange resin to ensure only H⁺ was present, and filtered through No. 2 fritted-glass filter before freeze-drying using a ScanVac CoolSafe freeze dryer (Labogene, Denmark). The process flow for the production of oxidised CNWs is shown in Figure 3.2.



Figure 3.2: The process flow for production of oxidised CNWs via TEMPOmediated oxidation

3.4 CHARACTERISATION METHODS OF CELLULOSE NANOWHISKERS

The characteristics and analysis of CNWs are crucial in understanding the mechanism on the adsorbent surface. A variety of methods are employed for characterisation in order to obtain a complete description of the structure, morphology and composition of the cellulose. These analytical methods will also provide information on the active sites involved in the binding of pollutants (Park *et al.*, 2010). The methods that have been used in this study are Fourier Transform Infra-Red (FTIR) spectroscopy, zeta potential, Brunauer–Emmett–Teller (BET) method, scanning electron microscopy (SEM), transmission electron microscopy (TEM) and conductometric titration method.

3.4.1 FOURIER TRANSFORM INFRA-RED SPECTROSCOPY

Infrared spectroscopy was used to determine vibration frequency changes in the oxidised CNWs on a Thermo-Nicolet 380 FTIR spectrometer (Thermo Fisher Scientific Inc., USA) in transmission mode. The samples were examined using a spectrometer with a 400 – 4000 cm⁻¹ range, with a resolution of 4 cm⁻¹ for 16 scans. 2 mg of the solid samples (CNWs) were milled with 200 mg potassium bromide (KBr) to form a very fine powder by using an agate pestle and mortar. KBr was used as background material in the analysis. This powder was then compressed to form a thin transparent disk, ready for spectra analysis.

3.4.2 ZETA POTENTIAL

Zeta potential of 0.1 % suspensions of samples in deionised water were measured at 25 °C using a Malvern Instrument Nano-ZS Zetasizer (Malvern Instrument Ltd, Worcestershire, UK). All zeta potential cells should be rinsed and cleaned with deionised water prior to usage in order to avoid any cross contamination of material from one measurement to other. The measurement performed was based on electrophoretic light scattering, and Smoluchowski was used as a model. The measurements were performed 10 times for each sample, and average results were calculated.

3.4.3 DETERMINATION OF CARBOXYLATE CONTENTS

The stability of the oxidised CNWs was tested over time under dry conditions and in the water matrix. For dry stability, the carboxylate content of the oxidised CNWs was determined using conductometric titration to prove that the functional groups were permanent and did not degraded under tested conditions. The titration was carried out on the oxidised CNWs stored under dry conditions for different times (7, 14 and 28 days).

For the wet stability test, freeze-dried oxidised CNWs were mixed with deionised water and left for 30 minutes, 24 hours and 7 days under constant shaking at a rate of 150 rpm in an incubator before titration experiments. Next, the suspensions were freeze-dried and used for the batch experiments. Freeze-drying was carried out because it allowed for accurate dosing of the CNWs. Batch experiments were carried out in conical flasks by adding oxidised CNWs in 20 mL of aqueous copper solution in the same conditions (initial Cu(II) concentration: 50.3 mg/L; sorbent dosage: 2.1 g/L; temperature: 21.3 °C). The initial and final concentrations of Cu(II) solutions were determined using AAS. Freshly prepared oxidised CNWs were used as the control experiment.

3.4.4 BET METHOD

The surface area and average pore diameter analysis of both samples (unmodified CNWs and oxidised CNWs) were performed using a Micromeritics TriStar II

Surface Area and Porosity analyser (Micromeritics, UK). For sample preparations, 20 mg of sample was placed into clean and dry tubes for gas preparation, and degassed in the Pfeiffer Vacuum TMH 071 (Micromeritics, UK) at 105 °C for 22 hours prior to determination of N₂ physisorption isotherm at -196°C (77 K). The weight of the sample was measured before and after degassing. The surface area of the sample was measured by the BET method while the pore size distribution of the sample was calculated from the desorption branch of the nitrogen adsorption-desorption isotherm using the BJH model.

3.4.5 ELECTRON MICROSCOPY (SEM)

Both samples (CNWs and oxidised CNWs) for the SEM study were prepared by direct deposition of samples on an aluminium holder, which was covered by a carbon grid. The sample was then sputter coated (SC 7640, High resolution sputter coater, Quorum Technologies, UK) with a thin layer of platinum (15 nm thick film) under vacuum, which was to provide a homogeneous surface for analysing and imaging. The sample was then examined in the SEM (XL 30 ESEM-FEG, Philips, USA), and operated under high vacuum with accelerating voltage of 20 kV.

3.4.6 ELECTRON MICROSCOPY (TEM)

A drop of 10 µl of both samples (unmodified CNWs and oxidised CNWs suspension) was first dispersed in deionised water for 15 minutes using sonication. Next, the sample was transferred to a copper grid (Carbon Films, 300 Mesh Cu Grids, AGAR) with hydrophilic surface by glow discharge treatment (Pelco, AGAR Scientific), and the excess liquid was absorbed by a filter paper. The specimens were then negatively stained with 2 % uranyl acetate solution to give better contrast between the sample and the carbon film. Excess solution was absorbed with a filter paper and left to dry by evaporation at ambient temperature before commencing the measurements. The surface morphology and particle size of both samples (unmodified CNWs and oxidised CNWs) were observed using a JEOL 2000FXII electron microscope operating at 80 kV accelerating voltage. Image J software was used to process the TEM images.

3.4.7 CONDUCTOMETRIC TITRATION

The content of carboxyl group on the oxidised CNWs was determined by the conductometric titration method (Saito and Isogai, 2004). Approximately 0.1 g of freeze-dried oxidised CNWs was mixed with 49 ml of deionised water and 1 ml of 0.05 M sodium chloride (NaCl), and the mixture was stirred to obtain a well-dispersed solution. The resulting suspension was titrated with 0.05 M sodium hydroxide (NaOH) solution at the rate of 0.1 mL/min while stirring continuously, and a conductivity meter recorded conductivity (Model No: 9811, Hanna Instruments Ltd, UK). The titration was carried out in triplicate for each sample, and the experimental errors were calculated as standard deviation. The carboxyl content of the sample was determined from the conductometric titration curve and was calculated by Equation 3.2. Assuming that each added molecule of NaOH with this range neutralised exactly one carboxyl group and all other weak acidic groups (aldehyde groups) were oxidised to carboxyl groups by the added NaCI:

$$X\left(\frac{mmol}{kg}\right) = \frac{C_t \cdot V_2}{m} \tag{3.2}$$

Where X is the total carboxyl group (mmol/kg), C_t is the concentration of the sodium hydroxide (mol/L), V_2 is the volume (L) of the sodium hydroxide solution consumed at the second intersection point, and *m* is the oven-dry weight of sample after titration (g) (Saito and Isogai, 2004).

3.5 EXPERIMENTAL SET UP FOR BATCH ADSORPTION STUDIES

The first and most important step in the design of experiment (DoE) procedure is to select significant variables that influence the adsorption process such as pH, contact time, temperature, sorbent dosage, and initial concentration (Figure 3.3). In order to do this, scoping studies were undertaken to identify the significant variables and parameter ranges that influence the adsorption process and help set the boundary conditions for the central composite design (CCD).





3.5.1 PREPARATION OF ADSORBATE SOLUTION

3.5.1.1 Clean water matrix

A Cu(II) stock solution with concentration of 1000 mg/L was prepared by dissolving the appropriate amount of CuSO₄.5H₂O in a 1000 ml volumetric flask, followed by dilution up to the mark by the addition of Milli-Q ultrapure water, and the required concentrations were obtained by diluting the stock solution (10 – 60 mg/L). The range for the initial Cu(II) ion solution was chosen as 10 - 60 mg/L, which is environmentally relevant to the actual concentration in municipal and industrial treatments plant (Table 2.4).

3.5.1.2 Wastewater effluent spiked with Cu(II)

The Cu(II) concentration in wastewater effluent was adjusted to the required concentrations (1-5 mg/L) by dissolving the appropriate amount of CuSO₄.5H₂O

in a 200 ml volumetric flask, followed by dilution up to the mark by the addition of filtered effluent.

3.5.2 BATCH ADSORPTION STUDIES USING CLEAN WATER MATRIX

Batch experiments were performed in 100 mL conical flasks in an incubator (Model No: 120, LMS Ltd., Kent, UK), with temperature control and agitation (150 rpm) using a mini table shaker (IKA Vibrax VXR, Germany). The contact time (30 min), and the initial pH (pH 6.0) were selected on the basis of the results obtained from the scoping experiments. The required mass of sorbent was measured separately into the 100 mL conical flask, and then 20 mL of Cu(II) solution with known concentration were added into the flasks. The initial pH of the solution was adjusted with 1 M H₂SO₄ and 1 M NaOH at 6, using the pH meter (Hanna Instruments Ltd., UK), calibrated with buffers of pH 4.0, 7.0, and 10.0 in order to maintain constant pH throughout the experiment. Next, the solution was separated from the sorbent using 0.2 µm surfactant-free cellulose acetate membrane syringe filter, and initial and the final concentration Cu(II) concentration after adsorption process were determined using AAS. All the experiments were conducted in duplicate and the average values are presented. For the control experiment, 20 ml of Cu(II) solution with known concentration were agitated in the absence of adsorbent for each experiment.

3.5.3 BATCH ADSORPTION STUDIES USING WASTEWATER EFFLUENT

The procedure for adsorption experiments, as described in Section 3.5.2, was performed using wastewater effluent spiked with Cu(II). The effluent was previously filtered through a standard 1.2 μ m glass fibre filter. The effect of pH (5–8), sorbent dosage (0.5–10 g/L) and initial concentration of wastewater effluent spiked with Cu(II) (1–5 mg/L) were carried out using effluent while keeping the other conditions the same, as with the clean water matrix. The pH of the solution was adjusted initially with 1M H₂SO₄ and 1M NaOH. In order to avoid any contamination, no efforts were made to maintain the pH throughout the adsorption process. The final pH was recorded. The initial and final solutions

were separated by filtration using $0.2 \,\mu m$ surfactant-free cellulose acetate membrane syringe filter and Cu(II) concentration determined using AAS.

The percentage of the removal Cu (II) ions by the sorbent and the adsorption capacity (mg Cu(II)/g) were expressed by:

$$\% removal = \frac{C_o - C_e}{C_o} x \ 100 \tag{3.3}$$

$$q_e = \left(\frac{(C_o - C_e)V}{W}\right) \tag{3.4}$$

Where C_o (mg/L) is the initial Cu(II) concentration and C_e (mg/L) is the equilibrium Cu (II) concentration in solution, V is the volume of the solution (L), and W is the mass of adsorbent (g) (Ghosh *et al.*, 2013).

3.5.4 BENCHMARK STUDY

A benchmark study was performed in order to understand the complexity and variability of the wastewater matrix on the adsorption performance. For benchmark studies, 20 ml of 4 mg/L Cu(II) wastewater effluent spiked with Cu(II) was agitated with 1 g/L sorbent dosage for 30 minutes at 20 °C at pH 6. This benchmark experiment was performed for each wastewater sample (for every sampling trip). The initial and final samples were separated by filtration using a $0.2 \,\mu\text{m}$ surfactant-free cellulose acetate membrane syringe and the concentration of the samples was determined by ICP-MS.

3.5.5 SELECTION OF INDEPENDENT VARIABLES AND RANGES THROUGH SCOPING EXPERIMENTS

3.5.5.1 Effect of pH

Experiments were carried out to ascertain the effect of pH on the adsorption of Cu(II) from the solution and to find the optimal pH for removing Cu(II) from the water matrix by using oxidised CNWs. Batch experiments were performed with 20 ml of 10 mg/L Cu solutions with pH adjustment using 1 M H_2SO_4 or 1 M NaOH to give a range from 4.0 to 8.0. Then 2 g/L of sorbent dosage was used at

different values of pH, and the flasks were gently agitated (150 rpm) at 20 °C in an incubator for 30 minutes to reach equilibrium. The initial and final solutions were separated by filtration using a 0.2 μ m surfactant-free cellulose acetate membrane syringe, and AAS determined the concentration of Cu(II).

3.5.5.2 Effect of contact time

The scoping studies were carried out to investigate the effect of contact time on the removal of Cu(II) from the water matrix by using oxidised CNWs. The effect of contact time was investigated for 5, 10, 20, 30, 60 and 180 minutes for both clean and wastewater matrix. Experiments were carried out by using 4 g/L of sorbent dosage with 20 ml of 10 mg/L Cu(II) ion solution, and were gently agitated (150 rpm) at 20 °C in an incubator. At each interval, the solution was filtered and the concentration of Cu(II) was determined by AAS.

3.5.5.3 Effect of temperature

The efficiency of the oxidised CNWs as adsorbents at different temperatures from 6 to 25 °C were investigated. This range was chosen because it is a realistic range for wastewater treatment temperatures in the environment. The adsorption experiments were conducted by using 4 g/L of sorbent dosage with 20 ml of 10 mg/L Cu(II) ion solution and were gently agitated (150 rpm) at 10 °C, 15 °C and 25 °C in an incubator for 30 minutes. The initial and final solutions were separated by filtration using a 0.2 μ m surfactant-free cellulose acetate membrane syringe, and AAS determined the concentration of Cu(II).

3.5.5.4 Effect of sorbent dosage

Experiments were carried out to study the effects of sorbent dosage in removing Cu(II) from the water matrix. The experiments were conducted by using sorbent dosage (0.2, 2.0, 4.0, and 10.0 g/L) at pH 6.0 with 10 and 5 mg/L of Cu(II) solution in clean water and wastewater effluent, respectively. After 30 minutes, the initial and final solutions were separated by filtration using a 0.2 μ m surfactant-free cellulose acetate membrane syringe, and AAS determined the concentration of Cu(II).

3.5.6 ADSORPTION ISOTHERM STUDIES

In order to understand the behaviour of adsorbent, two common adsorption isotherms (Langmuir and Freundlich) were used to evaluate the adsorption system design and to describe the adsorption capacities of the modified adsorbent (Shojaeimehr *et al.*, 2014). These isotherms were expressed as:

$$Langmuir: q_e = \frac{q_m K_L C_e}{1 + K_L C_e}$$
(3.5)

Where $q_e (mg/g)$ is the equilibrium amount of Cu(II) adsorbed per unit mass of sorbent, $q_m (mg/g)$ is the maximum Cu(II) ions adsorption capacity to form a complete monolayer on the surface bound and K_L (l/mg) is the Langmuir constant, which is related to the affinity of the binding sites.

Freundlich:
$$q_e = K_f C_e^{\frac{1}{n_f}}$$
 (3.6)

Where K_f and n_f are the Freundlich constants that indicates adsorption capacity and adsorption intensity, respectively.

3.5.7 ADSORPTION KINETICS STUDIES

The prediction of adsorption rate is important in providing the necessary information for the design of the adsorption system. Two kinetic models, which are Lagergren's pseudo-first order and pseudo-second order model, were applied to the experimental data in order to clarify the mechanism of adsorption process such as mass transfer and chemical reaction. The pseudo first-order and pseudo second-order adsorption kinetic rate equation are expressed as follows:

$$\frac{dq_t}{dt} = k_1(q_e - q_t) \tag{3.7}$$

$$\frac{dq_t}{dt} = k_2 (q_e - q_t)^2$$
(3.8)

After integration and applying boundary conditions t=0 to t=t and $q_t=0$ to $q_t=q_t$, the integrated form of Equation 3.9 and 3.10 by Lagergren are given as (Sarı and Tuzen, 2009):

$$\log(q_e - q_t) = \log q_e - \frac{k_1}{2.303}t$$
(3.9)

$$\frac{t}{q_t} = \frac{1}{q_e}t + \frac{1}{k_2 q_e^2}$$
(3.10)

Where q_t and q_e (mg/g) are the amounts of the metal ions adsorbed at time (min) and at equilibrium, respectively. The Lagergren rate constant for first and second-order equation is k_1 and k_2 (min⁻¹), respectively. The adsorption rate (k_1) can be determined by plotting log ($q_e - q_t$) against *t*, while a plot of t/ q_t versus *t* is used for second-order kinetic model and k_2 was found from the slope of the plot.

3.6 INSTRUMENTAL ANALYSIS OF COPPER

3.6.1 ATOMIC ABSORPTION SPECTROMETRY (AAS)

A Perkin-Elmer 272 Atomic Absorption Spectrometry (AAS) (Model No: 272, PerkinElmer Inc., USA) was used to measure the concentration of Cu(II) in aqueous solution. The operating parameters for Cu(II) analysis using AAS are specified in Table 3.2. An air-acetylene flame with a temperature of about 2300 °C was used for all measurements (Perkin Elmer, 1996). The adjustment of the burner with regards to height, lateral and rotational position was required. Fuel/air flows and nebulizer adjustments were also necessary in order to obtain maximum sensitivity.

Parameters	Description	
Flame type	Air/acetylene	
Lamp current	Copper (10 mA)	
Wavelength	324.8 nm	
Slit width	0.7 nm	

Table 3.2: Operating parameters for metal analysis using Atomic Absorption Spectrometry (AAS)

Burner alignment	Straight	Angled
Standard solutions	2, 4, 6, 8, and 10 mg/L	10, 20, 30, 40, 50, 60 and 70 mg/L
Linear regression curve	$y = 0.03313x + 0.00212$ $R^{2} = 0.9998$	y = 0.00488x - 0.00083 $R^2 = 0.9997$
Limit of detection	0.0382 mg/L	0.2966 mg/L
Limit of quantification	0.1273 mg/L	0.9886 mg/L

To set up the AAS instrument, a hollow cathode lamp of copper was installed into the instrument, and both the required wavelength and slit width were then set as listed in standard conditions. The instrument was calibrated with standard Cu(II) solutions. Calibration blank (zero concentration) is always measured first with 'auto-zero' on the calibration blank. Cu(II) determinations were interpolated from calibration curve equations derived from measurements of prepared standard Cu(II) solutions. For this study, two different calibration curves were used based on the concentration of the samples (Appendix A). In order to obtain accurate analysis, the calibrations were checked after every 10 samples by using the standards Cu(II) solutions and 'auto-zero' with the calibration blank. Analysis of each sample was carried out 5 times and an average reading taken.

3.6.2 PREPARATION OF AAS STANDARD SOLUTIONS

A certified 1000 mg/L calibration standard solution of Cu (in 2 % HNO₃) was purchased from ROMIL Ltd. The standard solutions (10–70 mg/L) and (1–10 mg/L) that span the working ranges were prepared by using the provided 1000 mg/L reference standard solution for Cu(II) with Milli-Q ultrapure water. As the dilute standard solutions will degrade with time, all the calibrated standards should be dated and replaced when necessary.

3.6.3 DETERMINATION OF DETECTION LIMITS AND QUANTIFICATION LIMIT

Limit of detection (LoD), or detection limit, is a term used to describe the lowest concentration level that can be reliably determined from a blank (Perkin Elmer, 2004). The instrument detection limit is typically defined as the concentration of

analyte, which is equal to three times the standard deviation of the concentration of 10 replicates of the calibration blank. In some circumstances, using LoD can still result in false positive or false negative data for a sample. Commonly, for most low-level analyses the limit of quantification (LoQ) is now routinely employed. This reduces the risk of quoting false positive or false negative data, as LoQ is greater than LoD by a factor of 3.3. LoQ is defined as 10 times the standard deviation of 10 replicates of calibration blank.

3.7 DESIGN OF EXPERIMENTS

The parameters and their range for Cu(II) removal from the water matrix were studied using response surface methodology (RSM) and artificial neural network (ANN) based on central composite design (CCD) (Figure 3.4). Optimisation studies were carried out by studying the effect of three variables (temperature, initial Cu(II) concentration, sorbent dosage and pH), based on type of water matrices.



Figure 3.4: Design of experiment (selection of experimental design)

3.7.1 CENTRAL COMPOSITE DESIGN (CCD)

A central composite design (CCD) was employed to evaluate the removal of Cu(II) from the water matrix. The factor levels for the experimental design were selected based on the literature and the scoping studies undertaken in this study, as described in Section 5.4. MINITAB 16 Statistical Software was used for statistical analysis. With three variables, a total of 20 experiments were required in order to find the optimum operating condition for the removal of Cu(II) using oxidised CNWs, as calculated from Equation 3.5:

$$N = 2^{n} + 2n + n_{c} = 2^{3} + 2(3) + 6 = 20$$
(3.11)

Where *N* is the total number of experiments required and *n* is the number of factors/independent variables. The centre points (n_c) were used to determine the experimental error and estimate the predictability of the data.

3.7.2 APPLICATION OF CCD FOR ADSORPTION FROM THE CLEAN WATER MATRIX

Three independent variables, temperature (6–25 °C), initial Cu(II) concentration (10–60 mg/L), and sorbent dosage (0.2–10.0 g/L) were optimised for the removal of Cu(II) from the clean water. The 20 CCD experiments were conducted as described in Section 3.5.2 at constant pH 6. The parameters are shown in Table 3.3 with their coded levels (– α , –1, 0, 1, α ; α = 1.633).

Independent variable	Factor code	Range and level				
		-α	-1	0	1	$+\alpha$
Temperature (°C)	X_1	6	9.68	15.5	21.32	25
Initial Cu (II) ion concentration (mg/L)	X_2	10	19.69	35	50.31	60
Sorbent dosage (g/L)	X ₃	0.2	2.09	5.1	8.1	10

Table 3.3: Experimental ranges and levels of the independent variables for clean water

3.7.3 APPLICATION OF CCD FOR ADSORPTION FROM THE WASTEWATER EFFLUENT

Three independent variables, pH (5 – 8), sorbent dosage (0.5 – 10.0 g/L), and initial concentration of wastewater effluent were spiked with Cu(II) (1 – 5 mg/L), were optimised for the removal of Cu(II) from the wastewater effluent. The 20 CCD experiments were conducted as described in Section 3.5.3 at a constant temperature of 20 °C. The parameters are shown in Table 3.4 with their coded levels (– α , –1, 0, 1, α ; α = 1.633).

Table 3.4: Experimental ranges and levels of the independent variables for wastewater effluent

Independent variable	Factor code	Range and level				
		-α	-1	0	1	$+\alpha$
pН	Y1	5	5.6	6.5	7.4	8
Sorbent dosage (g/L)	Y_2	0.5	2.34	5.25	8.16	10
Initial concentration of wastewater effluent spiked with Cu(II)	Y ₃	1	1.78	3	4.23	5

3.8 MATHEMATICAL MODELLING OF CU(II) REMOVAL FROM THE WATER MATRICES

3.8.1 RESPONSE SURFACE METHODOLOGY (RSM)

RSM is an approach that combines mathematical and statistical techniques and can be applied to give a better overall understanding with a minimal number of experiments. The experimental data were processed using MINITAB 16 Statistical Software. The predicted percentage of the removal of Cu(II) ions is explained by the following quadratic equation:

$$Y(\%) = \beta_0 + \sum_{i=1}^k \beta_i x_i + \sum_{i=1}^k \beta_{ii} x_i^2 + \sum_{i< j} \sum \beta_{ij} x_i x_j + \varepsilon$$
(3.12)

Where Y is the predicted response, x_i and x_j are the input variables, β_0 is the intercept term, β_i is the coefficient of linear effect, β_{ii} is the coefficient of squared effect, β_{ij} is the coefficient of interaction effect and ε is the random error.

3.8.2 ARTIFICIAL NEURAL NETWORK (ANN)

ANN is a powerful tool and has been widely used to model the effect of parameters influencing adsorption processes (Shanmugaprakash and Sivakumar, 2013). Although there are many well-known ANN types such as multilayer perceptron, radial basis function networks, linear networks, Bayesian networks, and Kohonen networks, currently the most popular network architecture is multilayer perceptron (MLP) (Savic *et al.*, 2012). This network consists of three or more layers of neurons as the three-layer models with one hidden layer and is commonly applied when predicting the performance of many processes (Pilkington *et al.*, 2014; Witek-Krowiak *et al.*, 2014). In order to use the ANN model for predicting Cu(II) removal from the water matrix, a feed-forward backpropagation was used for modelling the experimental design. In this study, the first layers of neurons representing the independent variables were identical to the factors considered in the RSM approach. Similar to the RSM modelling, the percentage removal of Cu(II) was considered as the output neurons and was developed in MATLAB (R2009b), Mathwork Inc. Software.

Contour plot of response surface



Artificial Neural Network (ANN)

Analysis of variance

(ANOVA)

3D plot of response surface

3.9 DATA ANALYSIS AND STATISTICAL TECHNIQUES

Optimal conditions
Unseen experiments
Absolute average deviation (ADD)
Root mean squared error (RMSE)
Coefficient of determination (R²)

Figure 3.5: Design of experiment (verification of the model)

3.9.1 EVALUATED OF THE MODELS

Fit and diagnose

model, interpret model

Confirm model

The regression analysis, graphical analysis, and analysis of variance (ANOVA) were undertaken using MINITAB 16 Statistical Software (Figure 3.5). The performance of the ANN and RSM model was statistically evaluated in terms of the coefficient of determination (R^2), absolute average deviation (AAD), and the root mean squared error (RMSE). R^2 value ($R^2 = 1$) implies perfect matching between predicted and actual data. This is a measure of its estimation performance for the actual observed values. Moreover, both models and the parameters variation were also determined based on the minimum value of the RMSE and AAD of the training and prediction set, as mentioned in Section 2.7 (Equation 2.6 and 2.7).

3.9.2 TEST AND VALIDATION OF THE MODEL

For purposes of validation and evaluation of RSM and ANN models, additional unseen experiments were conducted in addition to those determined by the CCD, consisting of combinations of experimental parameters not found in the training data set for the models. The prediction abilities of the newly constructed ANN and RSM models were also statistically measured in terms of R^2 , AAD and RMSE. The overall experimental batch adsorption carried out in this study is illustrated following the chart in Figure 3.6.



Figure 3.6: The flowchart of overall batch adsorption experiments
3.10 EXPERIMENTAL SET UP FOR FIXED BED ADSORPTION

In order to study the practical relevance of oxidised CNWs as adsorbent in largescale water treatment, column study by down-flow mode was studied. A continuous flow adsorption study was conducted in a solid phase extractions (SPE) vacuum manifold, with 20 positions (Waters, UK). An empty cartridge (column PD-10, GE Healthcare) made of polypropylene with 1.5 cm inner diameter and 7.4 cm height was packed with adsorbent and set up on the SPE vacuum manifold. A polyethylene frit was placed at the bottom of the cartridge to prevent loss of the adsorbent.

The experiment was performed at room temperature (20 ± 1 °C) by pumping a known concentration of wastewater effluent spiked with Cu(II) in a down-flow mode through the cartridge using a pump (GAST manufacturing, Inc., USA). The wastewater effluent was placed in a polypropylene container and connected with a tube through which the effluent will pass through the cartridge. The treated effluent was collected in a polypropylene container through the exit valve at the base of the glass chamber. Fixed bed sorption studies were performed under optimum conditions (pH 8, sorbent dosage = 6.45 g/L, initial concentration of wastewater effluent spiked with Cu(II) = 4.72 mg/L), obtained from previous experiment performed in a batch system for removal of Cu(II) from the wastewater effluent. For each sorption test, the cartridge was flushed with 5 ml deionised water to ensure compact packing and that the closely packed arrangement of adsorbent had no voids and channels.

The treated effluent (C_t) was collected after every 10 ml and analysed for metal concentration with AAS. The breakthrough curves of C_t/C_o were plotted against volume. The experiments were continued until a constant concentration of Cu(II) was obtained. The adsorption capacity $q_{e,cont}$ (mg/g) can be determined by the equation as in batch studies, but with slight modifications:

$$q_{e,cont} = \left(\frac{(C_o - C_b)}{W}\right) \times V_{ef}$$
(3.13)

Where W is the mass of adsorbent (g), C_o is the initial concentration (mg/L), C_b is the breakthrough concentration (mg/L) and V_{ef} is the volume (L) of effluent that is required to reach the exhaustion of the column.

CHAPTER 4: RESULTS AND DISCUSSION CHARACTERISATION OF CELLULOSE NANOWHISKERS

4.1 INTRODUCTION

This chapter describes the characterisation of cellulose nanowhiskers (CNWs) using several analysis techniques. Several techniques such as Fourier Transform Infrared Spectroscopy (FTIR), zeta potential measurements, Brunauer–Emmett–Teller (BET) analysis, Scanning Electron Microscopy (SEM), Transmission Electron Microscopy (TEM) and conductometric titration were used in this study. The procedures are fully described in Section 3.4.

There are two main sections in this chapter. The first section discusses the characterisation of CNWs using several analysis techniques prior to surface modification. The second section focuses on the stability of oxidised CNWs at different time intervals under dry conditions and in the water matrix.

4.2 HIGHLIGHTS

- Oxidised CNWs were stable at different time intervals under dry conditions and in the water matrix and proved that the functional groups were permanent and did not degrade under the tested conditions.
- Oxidised CNWs were effective in removing Cu(II) from the water matrix, unlike unmodified CNWs.
- Surface morphological, surface area and pore volume of oxidised CNWs were virtually unchanged during TEMPO-mediated oxidation.

4.3 **RESULTS AND DISCUSSION**

4.3.1 FOURIER TRANSFORM INFRA-RED (FTIR) SPECTROSCOPY

The adsorption characteristics of an adsorbent are influenced by the chemical reactivity of surface functional groups (Saito and Isogai, 2005). FTIR is widely used as a spectroscopic tool to examine functionalities present in CNWs. FTIR spectra of the adsorbent were also examined before and after the modification process as described in Section 3.4.1.

The FTIR spectra of CNWs (Figure 4.1) and the assignment of absorption bands are tabulated in Table 4.1, according to work by Marechal and Chanzy (Maréchal and Chanzy, 2000). The FTIR spectra of a typical cellulose show strong absorption characteristics at the OH stretching bands in the range of 3600–3000 cm⁻¹, CH and CH2 stretching vibrations between 3000–2800 cm⁻¹, and CO and CC stretching bands in the range of 1200–900 cm⁻¹ (Horikawa *et al.*, 2006). The strong absorption bands at 1075–1000 cm⁻¹ correspond to the primary alcohols while secondary alcohols absorb at 1150–1075 cm⁻¹ (Langkilde and Svantesson, 1995).



Figure 4.1: The FTIR spectra of CNWs

Wavenumber (cm ⁻¹)	Assignment
3346	νО-Н
2900	vC-H
1643	$\delta H_2 O$
1430	δС-О-Н
1337	δС-О-Н
1317	δС-О-Н
1206	vC-O-C
1163	vC-O-C
1113	vC ₂ -OH
1059	vC ₃ -OH
1034	vC ₆ -OH
706	ωС-О-Н
666	ωС-О-Н

Table 4.1: Assignment of bands in IR spectra of CNWs

The FTIR spectra of CNWs and oxidised CNWs are shown in Figure 4.2. After modification, the presence of a band at 1740 cm⁻¹ corresponds to the C=O stretching frequency of carboxyl groups incorporated into the CNWs (Cao *et al.*, 2012). In this spectrum, the presence of the band near to 1740 cm⁻¹ corresponds to the C=O stretching frequency of carboxyl groups in their acidic form (Habibi *et al.*, 2006).



Figure 4.2: FTIR spectra of the CNWs and oxidised CNWs

4.3.2 ZETA POTENTIAL DETERMINATION

Zeta Potential analysis is a technique utilised to determine the colloidal stability of nanoparticles in solutions. This analysis is important in the understanding of the state of the nanoparticle surface, to predict long-term stability of the nanoparticle and to determine whether a modification process has resulted in successful oxidation of the nanoparticle surface (Saito *et al.*, 2009). A value outside of the range of -25 mV to +25 mV generally indicates good stability of suspension which in the case of unmodified and oxidised CNWs, a stable suspension was observed (Ma *et al.*, 2014; Saito *et al.*, 2009). The procedure is fully described in Section 3.4.2.

The average zeta potentials are -37.6 mV and -74.4 mV for the unmodified and oxidised CNWs respectively. It is clearly apparent that the zeta potential is changed after the modification process. These observations are in agreement with those of Okita *et al.* (2010), who showed that the zeta potential of oxidised cellulose microfibrils dispersed in water are approximately -75 mV for all sources of cellulose. Moreover, the value obtained is acceptable, as it is in agreement with the zeta potential of TEMPO-oxidised cellulose derived from various sources, as reported in other studies (Table 4.2).

Origin of cellulose sample	Zeta potential (mV)	References	
Unmodified cotton	-37.6	This study.	
Modified cotton	-74.4	This study	
Cellulose sludge	-70.6	(Liu <i>et al.</i> , 2016)	
Pulp	-70.0	(Cheng et al., 2016)	
Softwood kraft pulp	-63.5	(Jin et al., 2014)	
Hardwood	-75.0	(Okita et al., 2010)	

Table 4.2: Zeta potential of TEMPO-oxidised cellulose from different sources

The oxidised CNWs show highly negative zeta potentials, caused by the introduction of carboxylate groups to the surface of the CNWs at a high density (Saito et al., 2009). Moreover, the zeta potential value of oxidised CNWs indicates that stable suspensions can be obtained with these modified particles. The greater the value, the higher the electrostatic repulsion between the particles, leading to an increase in stability, and hence less aggregation (Saito et al., 2009).

4.3.3 BET ANALYSIS

The BET surface areas for unmodified and oxidised CNWs are shown in Table 4.3 and BET plot in Appendix B The results obtained for CNWs were within the range of surface area reported for CNWs, e.g. 8–30 m²/g (Hsieh, 2013; Lu and Hsieh, 2010, 2012). The N₂ adsorption-desorption isotherm at 77K of both unmodified and modified CNW are shown in Appendix B.

From the results, the BET surface area of CNWs is approximately 10 times higher than that of the original cotton cellulose, as reported in the literature (Lu and Hsieh, 2010). In addition, the pore volume for CNWs is approximately five times higher than that of the original cotton cellulose, demonstrating that acid hydrolysis has not only introduced surface charges onto the surface of the CNW, but has also improved the surface area and pore volume after freeze-drying.

Samples	BET surface area (m ² /g)	Total pore volume (cm ³ /g)
Original cotton cellulose *	1.55	0.008
Unmodified CNWs	15.72	0.038
Oxidised CNWs	43.11	0.088
*(Lu and Hsieh, 2010)		

Table 4.3: BET surface areas and total pore volume of the samples

However, the surface area and pore volume for oxidised CNWs did not change greatly, being approximately three times higher than for the unmodified CNWs. This is as a result of the TEMPO-mediated oxidation, the main purpose of which is to introduce stable negative electrostatic charges onto the surface of the CNWs, and to obtain better dispersion stability with higher zeta potential, as mentioned in previous section (Habibi *et al.*, 2006). Depending on the morphology of the adsorbent and the modification procedure used, chemical modification may not always improve the surface area and pore volume of the adsorbent.

4.3.4 TEM ANALYSIS

Although SEM was found to be a quick and simple method to examine the CNWs, the resolution was considered insufficient for detailed information such as determination of the size of the whiskers. Therefore, TEM was used to confirm the morphological characteristics and size of individual CNWs. As observed in the literature, TEM images show that unmodified CNWs exist as needle-like structures, as shown in Figure 4.3 (Habibi, 2014; Kvien *et al.*, 2005), while a TEM image of oxidised CNWs is shown in Figure 4.4.

The lengths of CNWs were measured to be in the range of 64.8 to 296.1 nm, with an average of 134.4 ± 51.2 nm. Width was in the range of 4.3 to 13.9 nm with an average of 9.0 ± 2.3 nm. On the other hand, the dimensions of oxidised CNWs were found from TEM images to be in the range of 32.3 to 152.6 nm with an average of 95.0 ± 31.5 nm whereas width was in the range of 3.1 to 24.9 nm with an average of 10.5 ± 4.5 nm. The length and width measurements of both samples

correlate with earlier TEM analysis on CNWs from cotton (Elazzouzi-Hafraoui *et al.*, 2008). Therefore, it can be concluded that TEMPO-mediated oxidation does not compromise the morphological integrity and crystallinity of CNWs.



Figure 4.3: A typical TEM image of negatively stained preparations of unmodified CNW adsorbents



Figure 4.4: A typical image of negatively stained preparations of oxidised CNW adsorbents

4.3.5 DETERMINATION OF CARBOXYLATE CONTENTS

There are several characterisation methods for measuring the carboxylate content of oxidised cellulose, including carbon-13 nuclear magnetic resonance (¹³C NMR) spectroscopy, quantitative infrared spectroscopy (FTIR), conductometric titration and methylene blue adsorption. These methods have been compared by da Silva Perez et al., (2003) who found that the methylene blue adsorption method showed an underestimated result, while ¹³C NMR appeared to overestimate the results. Moreover, FTIR showed low repeatability due to problems associated with quantification, whereas conductometric titration showed reproducible results and was therefore advocated as a reproducible method for measuring the carboxylate content of oxidised cellulose. Saito and Isogai used the conductometric titration method with cellulose nanocrystals further oxidised with NaClO and sodium hypochlorite; can oxidise aldehyde groups into carboxyl groups, increasing the number of carboxyl groups on CNWs (Saito et al., 2005). Therefore, the carboxyl group content on oxidised cellulose nanowhiskers was determined by conductometric titration. A typical conductometric titration curve is shown in Figure 4.5, where the total number of carboxyl groups can be determined from the second intersection point.



Volume, strong base

Figure 4.5: Typical conductometric titration curve (Levlin, 2010)

The conductometric titration curve of oxidised CNW suspensions is shown in Figure 4.6 and is characterised by three phases. First, the decreasing phase corresponds to the decrease of free H⁺ ions through neutralisation of strong acid groups by NaOH. Then, the horizontal phase shows the neutralisation of the weak acids, such as carboxylic acid. In this phase, the added sodium ions are adsorbed as counter ions to the carboxylic acidic groups, and the dissociated protons are neutralised by the added hydroxide ions. After the weak acid is completely neutralised, both cation and anion of the base (NaOH) will contribute to a sharp increase in conductivity of the solution due to the excess of added base. The procedure and calculation for the determination of total number of carboxyl groups are fully described in Section 3.4.4.



Figure 4.6: Determination of the total number of carboxyl groups in oxidised CNW suspension by conductometric titration.

Table 4.4 displays the carboxylate content using different starting material such as cotton linters and hardwood kraft pulp. Although the oxidation conditions applied were similar, the carboxylate content obtained was different for the three samples. Carboxylate content in oxidised CNWs was noticeably higher, indicating that there are more carboxylate groups per gram due to high surface area. Furthermore, work by Okita *et al.*, (2010) indicated that the total carboxylate contents of cellulose were different for plant and non-plant materials such as bacterial, tunicate and algae celluloses.

Cellulose sample	Carboxylate content (mmol/kg)	Reference	
CNWs 49.33		This study	
TEMPO-oxidised CNWs 418.39			
Cotton linters 19.0		(Spite at al. 2005)	
TEMPO-oxidised cotton linters	226.0	(Sano et al., 2005)	
Hardwood kraft pulp 50.0		(Saito and Isogai,	
TEMPO-oxidised hardwood	260.0	2006)	

Table 4.4: Carboxylate contents of cellulose before and after TEMPO-mediated oxidation

The reproducibility of results for oxidised CNWs was tested by performing the TEMPO-mediated oxidation on different batches. The results are shown in Table 4.5. The total carboxylate content obtained for different batches of oxidised CNWs indicate that the method is highly reproducible. In view of the industrial development of a variety of sorbents described in the literature, the reproducibility and consistency between different batches of this method is a highly desirable feature in product design. For example, activated carbon is the most popular adsorbent used for the removal of pollutants from wastewater due to its high adsorption capacity, extensive specific surface area, and reproducibility (Kyzas *et al.*, 2013).

Table 4.5: The amount of carboxylate content for oxidised CNWs for different batches

Oxidised CNWs	Carboxylate content (mmol/kg)
Sample 1	390.88
Sample 2	409.50
Sample 3	418.39
Average ± standard deviation	406.26 ± 14.04

A control experiment was then carried out to compare the ability of unmodified and oxidised CNWs to remove Cu(II) from the water matrix. For the same amount of sorbent dosage under similar conditions, oxidised CNWs were able to remove approximately 66.75 % of Cu(II) while the unmodified CNWs removed only approximately 3.64 % of Cu(II) from the water matrix. The Cu:COOH molar ratio, which is used to compare the removal capability of oxidised CNWs and unmodified CNWs. The Cu:COOH molar ratio for oxidised CNWs is 2.59x10⁻³, while Cu:COOH molar ratio for unmodified CNW is 0.54.

These results show that a chemical modification such as oxidation can alter the surface of CNWs by adding more carboxyl groups, increasing the number of active sites (Thirumavalavan *et al.*, 2010; Zhang *et al.*, 2006). The percentage of Cu(II) removed increases significantly with the presence of a large number of carboxyl groups on the surface of the adsorbent. These observations are in agreement with those of Liu *et al.*, (2009) who used carboxylic acid functionalised *konjac glucomannan* for the adsorption of copper and lead from aqueous solutions.

4.3.6 STABILITY TEST

Stability tests were conducted under two different conditions, under wet and dry conditions, and the stability of the adsorbent was reviewed by measuring the carboxyl content and adsorption capabilities over time. Interestingly, to date, no work has tested the stability of these modified CNW adsorbents, which is really important from both the manufacturing and application perspective. As shown in Table 4.6, the carboxylate content of oxidised CNWs did not change, which demonstrates that oxidised CNWs may be stored under dry conditions for the period time examined (up to 28 days). The procedures are fully described in Section 3.4.3 and 3.4.4. However, the carboxylate contents can be partially degraded due to extreme reaction conditions during chemical or mechanical treatment. This harsh treatment not only decreases the carboxylate content, but also affects the cellulose morphology and crystallinity (Wang *et al.*, 2007).

Days	0	7	14	28
Carboxyl groups (mmol/kg)	406.26	412.31	409.49	413.21

Table 4.6: The carboxylate content of oxidised CNWs under dry conditions

For the wet stability test, freeze-dried CNWs were mixed with deionised water and left for 30 mins, 24 hours and seven days under constant shaking at a rate of 150 rpm in an incubator. Next, the suspensions were freeze-dried before the batch experiments. Freeze-drying was chosen because it allows for accurate dosing of the CNWs. In addition, freeze-drying is a suitable method to preserve the nanoscale dimensions of the cellulose nanowhiskers (Peng *et al.*, 2012). Smoother surfaces and nano-scale dimensions were obtained for freeze-dried samples compared to other drying methods, such as air- and oven-drying (Peng *et al.*, 2012). During the freeze-drying process, capillary forces are minimised and no bulk material is formed (Takaichi *et al.*, 2014).

As shown in Table 4.7, the oxidised CNWs are stable in water for a longer period. The sorbent ability for each sample did not change or reduce, indicating that oxidised CNWs are stable in water for up to 7 days. This showed that the functional groups are permanent under the conditions tested, as they were stable and were not removed or degraded. The Cu(II) removal ability remained the same and was stable in water. This proved that oxidised CNWs are suitable for use as an adsorbent in wastewater treatment. Moreover, numerous studies have demonstrated that the stability of CNW suspension can also be improved through TEMPO-mediated oxidation (Araki *et al.*, 2001; Habibi *et al.*, 2006; Saito and Isogai, 2004).

Table 4.7: The sorbent ability under wet conditions (T = 21.3 °C, C_i = 50.3 mg/L, m = 2.1g/L)

Sample	$C_{\rm f}({\rm mg/L})$	Percentage removal (%)	q (mg/g)	
Fresh	36.40	31.49	0.77	
30 mins	37.69	29.06	0.74	
24 hours	38.08	28.33	0.72	
7 days	37.84	28.78	0.73	
C _i =Initial Cu(II) concentration; C _f =Final Cu(II) concentration; q=adsorption capacity				

CHAPTER 5: RESULTS AND DISCUSSION **PREDICTING THE CAPABILITY OF OXIDISED CNW ADSORBENTS FOR THE REMEDIATION OF COPPER FROM WATER USING RSM AND ANN MODELS**

5.1 INTRODUCTION

Cellulose nanowhisker (CNW) was chemically modified via TEMPO-mediated oxidation to improve its adsorption capacity and enhance its performance under desired conditions in a wastewater treatment plant (WWTP). The preparation of oxidised CNW adsorbents is fully discussed in Chapter 3 (Section 3.3). Following on, the results obtained from the characterisation study, which confirmed the success of modification (Chapter 4), adsorption batch experiments were conducted to evaluate the performance of oxidised CNW adsorbents to adsorb and remediate Cu(II) from the water matrix. Moreover, two models, namely the response surface methodology (RSM) and artificial neural network (ANN) were evaluated for their capability to determine the effectiveness of oxidised CNW adsorbents at removing Cu(II) from the clean water matrix at desired conditions.

There are three main sections in this chapter. The first section discusses the selected variables and parameter ranges for application to the clean water matrix for Cu(II) removal through scoping experiments. Batch experiments were performed according to a CCD matrix of conditions to determine the percentage of Cu(II) removal. Next, both RSM and ANN models were employed to understand the obtained data and evaluate the predictive capability of each model for effective Cu(II) removal from the water matrix. Finally, the third section discusses the ability of both models at predicting data not found in the training data set for the models, as described in Section 3.5.

5.2 HIGHLIGHTS

- This study focused on real wastewater treatment plant conditions, unlike CNW and other adsorbents studies.
- Both RSM and ANN models were used as predictive, not only descriptive models.
- Model suitability was tested outside of the model descriptive range, unlike CNW and other adsorbent works.
- ANN had superior prediction capability to RSM for Cu(II) removal on oxidised CNW adsorbents.

5.3 RESULTS AND DISCUSSION

5.3.1 CONTROL EXPERIMENTS FOR THE EFFECT OF FILTER

Control experiments were undertaken to determine a suitable filter, as this is a necessary stage prior to the separation of adsorbent from the solution (as described in Section 3.5.3). Several studies have investigated the possible factors that can significantly affect metal concentrations, e.g., the type of filter, membrane materials and filtration method (Matoug, 2013). Moreover, the existing literature addresses the use of filter paper in filtration causing systematic errors on the analysed solution, particularly at lower metal concentrations (Engin *et al.*, 2010; Ileri *et al.*, 2014). Therefore, in order to ascertain that the filter membrane material used did not retain metal ions in the solution, control experiments were carried out to identify the most appropriate filter. Surfactant-free cellulose acetate filters were identified as most suitable, with an average retention of 0.8 ± 0.5 %; subsequent results corrected for this. Additionally, the normal cellulose acetate filters evaluated demonstrated adsorption to the filter with an average adsorption of 7.4 ± 1.0 % (Table 5.1).

Type of filter	C _i (mg/L)	C _f (mg/L)	Average retention (%)	Standard deviation	
Cellulose		55.59			
acetate	59.44	53.87	7.4	1.0	
(0.25 µm)		55.63			
Surfactant-		59.15			
free cellulose acetate	59.44	59.36	0.8	0.5	
(0.45 µm)		58.38			
C_i = Initial Cu(II) concentration; C_f = Final Cu(II) concentration.					

Table 5.1: Data for the control experiments to determine appropriate filter.

5.3.2 CONTROL EXPERIMENTS TO DETERMINE PARAMETER RANGES FOR THE CENTRAL COMPOSITE DESIGN (CCD)

To help set the boundary conditions for the CCD, scoping studies were undertaken to identify the variables and parameter ranges that influence the adsorption process. Furthermore, parameters and their ranges were also selected based on the existing literature, since in many studies, the range of investigated parameters for the adsorption process is often not representative of the actual conditions in a WWTP (Cao *et al.*, 2014; Thirumavalavan *et al.*, 2010). Therefore, prior to implementing the CCD, it is important to confirm the variation of ranges for each parameter. Scoping studies adopted a one-parameter at a time approach to determine the ranges to be used for each parameter.

5.3.2.1 Initial Cu(II) concentration

The range for the initial Cu(II) ion solution was chosen as 10 - 60 mg/L, which is environmentally relevant to the actual concentration in industrial treatment plants (as described in Chapter 2; see Table 2.4). Wang and Chen (2009) reported that chemical precipitation and electrochemical treatment were ineffective, especially when the range for metal ion concentration in aqueous solution is 1 - 100 mg/L(Wang and Chen, 2009). Moreover, Isobe *et al.* (2013) performed the adsorption process with a high initial metal ion concentration (1000 mg/L of a Cu(II) solution), which is unrealistic for actual commercial adsorption processes, as they are generally applied to low concentration streams. A focus on finding new adsorbents highlights that treatment technologies such as chemical precipitation and electrochemical treatment are unable to treat low copper concentrations.

5.3.2.2 Contact time of the adsorption process

The purpose of this experiment was to determine the necessary contact time needed for the adsorption process to reach adsorption equilibrium. The equilibrium time (time required to reach this state of equilibrium) is a significant operational factor for economical wastewater treatment processes (Amuda *et al.*, 2009). A set of experiments at predetermined time intervals (5, 10, 20, 30, 60, 120 and 180 min) were carried out at different sorbent dosages and initial Cu(II) concentrations (as shown in Figure 5.1 and 5.2); the procedure is described in Section 3.5.4.2.

Figure 5.1 shows that adsorption increased sharply with contact time during the first 5 min, contributing more than 91 % of Cu(II) removal. The experimental results showed that the adsorption process in this study occurred in two phases, i.e. a rapid initial sorption rate (first 5 min) followed by a slower rate in later stages (30 min). It is noted that almost 90% removal was achieved shortly, at the first 5 min, of the adsorption process. From the Figure 5.2, it is clear that the adsorption process is very rapid during the first 5 min and it is noted that the equilibrium is reached already after 30 min. This rapid adsorption indicates that the active adsorbent sites are readily occupied by the Cu(II) ions as soon as the adsorbent is introduced into the system. Hence, the maximum removal of Cu(II) was attained during this period.

After reaching equilibrium, the removal efficiency varies about 1-3% with increase in contact time. This may be due to the process of adsorption and desorption that take place adsorbent surface saturation by Cu(II) ions. As such, contact time will be a constant parameter and was set at 30 min for all adsorption studies in this current research, since further increases in contact time did not indicate a significant change in percentage removal.

The similar results were obtained by other studies (Sengil and Oezacar, 2008; Srivastava and Hasan, 2011; Wankasi *et al.*, 2006a). However, there were few

papers that investigate on how the adsorption process on the potential adsorbent at very short contact time (Mehdinia *et al.*, 2015). Most of the studies in literature performed the batch experiments at 5 min intervals (5, 10, 20, 30, 60 min). Wågberg and Hägglund (2001) has utilised a new method in order to be able to perform the adsorption of different polyelectrolyte adsorption on bleached cellulosic fibers process at short contact time. An equipment known as "The Jar" has been used to collect the final samples for after every short contact time (10 seconds and upwards). Thus, for further study in adsorption process, a more detailed study of the implementation of a new method for short contact time would be necessary in order to understand the mechanism of the adsorption process.

Moreover, the oxidised CNW adsorbents reached a much shorter equilibrium time for the removal of Cu(II), compared to other conventional adsorbents such as activated carbon and zeolites. This indicates that the modification of CNW via TEMPO-mediated oxidation improves adsorption performance. Due to its high density of carboxyl groups and its nanoscale dimensions, the performance of oxidised CNW adsorbents was efficient when compared to other conventional adsorbents. Table 5.2 summarises the equilibrium time needed for the removal of Cu(II) ions with the application of common adsorbents and oxidised adsorbents from previously published studies. According to Wankasi *et al.* (2006b), a short contact time for reaching equilibrium indicates that the predominant mechanism is chemical adsorption (chemisorption), while a longer contact time indicates physical adsorption, as described in Section 2.3.1.



Figure 5.1: Variations in the removal efficiency of Cu(II) with contact time for different sorbent dosages: Initial Cu(II) concentration = 60 mg/L; T = 25 °C; pH = 6.



Figure 5.2: Variations in the removal efficiency of Cu(II) with contact time for different initial concentrations of Cu(II) solution: sorbent dosage = 10 g/L; T = 25 °C; pH = 6.0 C

Modified/functionalised adsorbents	Equilibrium time (hrs)	References	
Oxidised CNW	0.5	This study	
Nitrogen containing functional group activated carbon	8 – 10	(Kasnejad et al., 2012)	
Carboxylated functionalised bentonites	5	(Anirudhan <i>et al.</i> , 2012)	
Alginate-immobilised bentonite clay	4 – 6	(Tan and Ting, 2014)	
Mimosa tannin gel	3	(Sengil and Oezacar, 2008)	
Nano-NaX zeolites	2	(Ansari et al., 2015)	
Carboxylated cellulose nanocrystals (with succinic anhydride)	2.5	(Yu <i>et al.</i> , 2013b)	

Table 5.2: The equilibrium time for Cu(II) removal by common adsorbents.

5.3.2.3 Cu(II) solution pH in the adsorption process

One of the most important factors that not only influences the surface charge of the adsorbent, but also the solution chemistry of heavy metal, is pH (Reddy, 2012). As discussed in Section 2.1, the pH range in municipal and industrial treatment plants was found to be in the range of 6 - 7 (Saiano *et al.*, 2005). However, the optimum pH range (4 - 6) for the adsorption process published in the literature was considered too acidic, since a pH below 6 will increase the competition between protons and metal ions for active sites (Reddy, 2012). Thus, it is important to note that the removal of Cu(II) may decrease when pH is acidic, rather than having an optimum pH. Moreover, for copper, adsorption studies should not be carried out experimentally at a pH higher than 6, due to the limited solubility of Cu(II) leading to precipitation in this pH range (Tunay *et al.*, 1992). According to Thirumavalavan *et al.* (2010), establishing a pH range without considering metal hydroxide precipitation will also affect the removal of Cu(II) from the water matrix. The theoretical minimum solubility for different metals occurs at different pH values. As such, it is important to investigate a range of pH

values via scoping experiments for the adsorption process, as metal hydroxide precipitation may complicate the sorption process between adsorbent and heavy metals (Reddy, 2012).

Control experiments were therefore carried out to check the eventual precipitation of metals at a given pH in order to prove that the theoretical minimum solubility for different metals occurs at different pH values. The effect of pH on the adsorption of Cu(II) by oxidised CNW adsorbents is shown in Table 5.3. From the control experiment, the optimum pH value for maximum removal of Cu(II) was observed at pH 6, with 91.37 % (2.04 mg/g) of Cu(II) removed, while at pH 4, only 85.56 % (1.88 mg/g) Cu(II) removal was achieved. It can therefore be concluded that at a pH below 6, the removal percentage was found to decrease, due to electrostatic repulsion between positively charged metal ions and negative adsorption sites. At low pH (pH 4), the amino acid is protonated at both the amine and carboxyl functions. It was reported that at the acidic region, there would be a strong competition between the H⁺ and Cu²⁺, which will reduce the removal percentage of Cu(II) by the carboxyl group (Reddy, 2012).

However, as pH increased to 6, there was an observed decrease in H^+ ions in the solution and hence, more negatively charged sites were available for adsorption (Kadirvelu *et al.*, 2001). As such, it was decided that the optimum pH for the removal of Cu(II) ions using oxidised CNW adsorbents was pH 6, which is also a realistic condition in the wastewater treatment environment (Chand *et al.*, 2015; Saiano *et al.*, 2005).

Sample No.	рН	Final concentration $C_f(mg/L)$	Removal efficiency (%)	Adsorption capacity q (mg/g)	
1	4.0	1.29	85.56	1.89	
2	6.0	0.77	91.37	2.04	
Co= initial concentration; m=sorbent dosage; T=temperature; t=time					

Table 5.3: The effect of Cu(II) solution pH on the removal of Cu(II); $C_o = 9$ mg/L, m = 4 g/L, T = 25°C, t = 30 min.

5.3.2.4 Determination of sorbent dosage (Oxidised CNW adsorbents)

Adsorbent dosage is an important parameter when evaluating the adsorption process, as it will provide the amount of sorbent dosage for a given initial concentration, separation cost and the total water treatment cost (Anupam *et al.*, 2011). The amount of sorbent used for a volume of water (sorbent dosage, g/L) must therefore be determined to ensure maximum Cu(II) removal for the wastewater condition being evaluated. The effect of sorbent dosage on the removal percentage of Cu(II) is shown in Figure 5.3.



Figure 5.3: Variations in Cu(II) removal efficiency according to sorbent dosage (Oxidised CNW): initial Cu(II) concentration; $C_i=10 \text{ mg/L}$; T=25 °C; pH=6.

As shown in Figure 5.3, it is clear that the percentage of Cu(II) removal increased from 59.3 % to 92.4 %, alongside sorbent dosage increasing from 0.2 to 10.0 g/L at 25°C for the conditions of pH 6 and an initial concentration of Cu(II) of 10 mg/L. These results correspond well with previous studies using other biomass adsorbents (Rao *et al.*, 2012; Sahan *et al.*, 2010; Singh *et al.*, 2010). The amount of proton exchange between the oxidised CNW adsorbents and the Cu(II) solution escalated alongside an increasing amount of sorbent dosage; this may be attributed to the increased availability of exchangeable sites (Geyikci *et al.*, 2012). Thus, increasing the amount of sorbent dosage in the initial concentration is favourable for improving Cu(II) removal during the adsorption process. Hence, the sorbent dosage range investigated in this study was chosen as 0.2-10 g/L; at

10 g/L, the highest dosage, more than 90 % of Cu(II) could be removed from the initial concentration of Cu(II) solution (60 mg/L).

5.3.2.5 Temperature of the adsorption medium

Several researchers have noted adsorption medium temperature to be an important factor when the mechanism of the metal-binding processes is energy dependent (Bulut and Tez, 2007; Reddy, 2012; Sahan et al., 2010). This temperature should be a realistic range in the wastewater treatment environment, where the mean annual temperature of wastewater varies from 6 - 25 °C (Burton et al., 2013). Another example is a study conducted by Hanaki (2008), who showed wastewater temperature range to be 15 – 25 °C (Hanaki, 2008). However, most of the reported studies were conducted under conditions that were not realistic to wastewater treatment environments. Moreover, these studies, which were performed at ranges not relevant to the real world, may have reduced the capabilities of the adsorbents. For example, Sahan et al. (2010) reported a decrease in the removal of Cu(II) from aqueous solutions at temperatures up to 45 °C, as too high a temperature can accelerate the decomposition of chelating efficiency, leading to a decrease in the adsorption ratio. Zhao (2006) investigated the influence of temperature on the adsorption of Cu(II) by cellulosic-adsorbent resin and showed that adsorption ratio gradually decreased alongside an increase in solution temperature from 35 - 60 °C (Zhao, 2006). The investigated temperature range in this current research was chosen to be 6 - 25 °C in order to study the efficiency of oxidised CNW as adsorbents at different temperatures, as justified in Table 2.9 (Section 2.3.3). Therefore, the temperature (6 - 25 °C), initial Cu(II) ion concentration (10 - 60 mg/L) and sorbent dosage (0.2 - 10 g/L)were used as independent (input) variables, and were investigated for their impact on the removal efficiency of Cu(II) from the water matrix (Table 5.4).

Parameters	Range
Initial concentration (mg/L)	10 - 60
Temperature (°C)	6 – 25
Sorbent dosage (g/l)	0.2 - 10

Table 5.4: Ranges of the investigated parameters.

5.4 MODELLING OF THE ADSORPTION PROCESS

5.4.1 CENTRAL COMPOSITE DESIGN (CCD)

CCD is the most frequently used design method and was employed in this study because it has better predictive capabilities, and has been extensively applied in adsorption studies. Therefore, in the following experiments, CCD is used to develop a correlation between three independent variables and one output (response). The temperature, initial Cu(II) ion concentration and sorbent dosage were used as independent (input) variables and were studied for their impact on the removal of Cu(II) from the water matrix. Using these three variables, a total of 20 experiments were required in order to locate the optimum operating conditions for the adsorption of Cu(II) using oxidised CNW adsorbents. The range of independent variables, alongside their coded levels (- α , -1, 0, 1, α ; α = 1.633, respectively) is presented in Table 5.5. The value of α , which depended on the number of factors, was chosen to maintain rotatability, which refers to the uniformity of prediction error (Carroll, 2003). In order to obtain a good estimate of experimental error (pure error), the central point was repeated six times. The obtained quadratic equation in terms of coded factors for Cu(II) removal percentage in the form of Y is:

$$Y (\%) = 77.51 + 0.94X_1 - 10.92X_2 + 17.34X_3 + 0.79X_1^2 + 0.50X_2^2 - 8.06X_3^2 + 0.69X_1X_2 + 0.92X_1X_3 + 3.84X_2X_3$$
(5.1)

Where Y is the percentage removal of Cu(II), X_1 is temperature, X_2 is the initial concentration and X_3 is the sorbent dosage.

Independent variable	Factor code	Range and level				
		-α	-1	0	1	$+\alpha$
Temperature (°C)	X_1	6	9.68	15.5	21.32	25
Initial Cu(II) ion concentration (mg/L)	X_2	10	19.69	35	50.31	60
Sorbent dosage (g/L)	X ₃	0.2	2.09	5.1	8.1	10

Table 5.5: CCD Experimental ranges and levels of the independent variables.

5.4.2 RSM MODEL

Results for the percentage of Cu(II) removal were obtained by performing batch experiments according to the CCD matrix of conditions. Table 5.6 shows the experimental results obtained from the experimental runs and the predicted values by the build RSM model. The percentage error (% error) was calculated as the ratio of difference in experimental and predicted value to experimental value (Equation 3.11).

One of the 20 experiments, with 0.20 g/L sorbent dosage, showed a large residual error with 10.9 % for the RSM model, which influenced the value of R^2 . Similar problems occurred but not to the same degree of error when less than 2.10 g/L (0.042 g of oxidised CNW adsorbents) sorbent dosage was used to remove Cu(II) from the water matrix. This problem is believed to be due to the presence of adsorbed species at the surface of the cellulose nanowhiskers blocking reactive sites (Labet and Thielemans, 2011). For the higher sorbent dosage, this will not have an impact on the adsorption process, due to greater availability of reactive sites on the oxidised CNW. Similar results have been reported for other heavy instances of metal adsorption onto biomass (Sahan *et al.*, 2010; Sarkar and Majumdar, 2011; Sedighi *et al.*, 2012; Sugashini and Begum, 2013). However, from Figure 5.4, it still can be observed that the predicted values by the RSM model and the actual experimental data are in good agreement, with a coefficient of determination ($R^2 = 0.9541$).

Run Number	Coded values			Actual values				Sorption capacity			
	\mathbf{X}_1	X_2	X ₃	\mathbf{X}_1	X_2	X ₃	Experimental	Predicted	Residual	Absolute Error (%)	q (mg/g)
1	1	1	-1	21.3	50.31	2.10	44.50	39.35	5.15	11.57	10.63
2	1	-1	1	21.3	19.69	8.10	95.72	96.33	-0.61	0.64	2.47
3	-1	-1	-1	9.7	19.69	2.10	74.81	68.83	5.98	7.99	6.53
4	0	0	0	15.5	35.00	5.10	76.95	77.51	-0.56	0.72	5.05
5	0	0	0	15.5	35.00	5.10	76.35	77.51	-1.16	1.53	4.95
6	-1	1	1	9.7	50.31	8.10	77.80	78.46	-0.66	0.84	4.56
7	1	-1	-1	21.3	19.69	2.10	72.54	67.49	5.05	6.96	5.75
8	0	0	0	15.5	35.00	5.10	78.97	77.51	1.46	1.85	5.14
9	0	0	0	15.5	35.00	5.10	75.68	77.51	-1.83	2.42	4.78
10	-1	-1	1	9.7	19.69	8.10	93.24	94.00	-0.76	0.82	2.09
11	-1	1	-1	9.7	50.31	2.10	42.93	37.93	5.00	11.64	9.63
12	1	1	1	21.3	50.31	8.10	81.94	83.54	-1.60	1.95	5.30
13	-α	0	0	6.0	35.00	5.10	74.33	78.09	-3.76	5.05	4.77
14	0	0	α	15.5	35.00	10.00	88.65	84.34	4.31	4.87	2.99
15	0	0	0	15.5	35.00	5.10	78.08	77.51	0.57	0.73	4.72
16	0	α	0	15.5	60.00	5.10	58.26	61.00	-2.74	4.71	6.50
17	α	0	0	25.0	35.00	5.10	78.33	81.14	-2.81	3.59	5.07
18	0	-α	0	15.5	10.00	5.10	92.85	96.68	-3.83	4.12	1.58
19	0	0	0	15.5	35.00	5.10	81.20	77.51	3.69	4.55	5.07
20	0	0	-α	15.5	35.00	0.20	16.81	27.70	-10.89	64.78	20.67
X_1 =temperature (°C); X_2 =initial Cu(II) ion concentration (mg/L); X_3 =sorbent dosage (g/L).											

Table 5.6: The experimentally obtained percentage of Cu(II) removal compared to that predicted by the central composite design (CCD) models



Figure 5.4: The experimentally obtained Cu(II) removal compared to that predicted by the response surface methodology (RSM).

Analysis of variance (ANOVA) was applied to evaluate the quality of the fitted model (Bezerra *et al.*, 2008). The ANOVA of this model is presented in Table 5.7 and the model was assessed for its suitability by examining the lack of fit through ANOVA. The lack of fit obtained is significant, due to low probability (P = 0.005) and a higher F-test value of 14.73. These results indicate that the RSM model is unable to effectively predict the removal of Cu(II) from the water matrix.

The significance of each term in the equation regarding the percentage of adsorbed Cu(II) ions was validated by this statistical test. Remarkably, as can be seen from the results, most of the terms in the quadratic model were statistically insignificant (P>0.05) in terms of their effect on Cu(II) percentage removal with a model F-value of 23.09.

Source	Sum of squares	DF	Mean square	F value	P-value Prob > F	alue $b > F$ Coefficient		
Model	6635.87	9	737.32	23.09	< 0.0001	77.5063		
X_1	11.63	1	11.63	0.360	0.560	0.9339		
X_2	1590.5	1	1590.50	49.81	< 0.0001	-10.9219		
X ₃	4010.2	1	4010.2	125.60	< 0.0001	17.3425		
X_1^2	22.39	1	8.29	0.26	0.621	0.7922		
X_2^2	15.4	1	3.32	0.10	0.754	0.5016		
X_3^2	857.45	1	857.45	26.85	0.000	-8.0577		
X_1X_2	3.78	1	3.78	0.12	0.738	0.6875		
X_1X_3	6.70	1	6.70	0.21	0.657	0.9150		
X_2X_3	117.81	1	117.81	3.69	0.084	0.3838		
Residual	319.29	10	31.93					
Lack of fit	299	5	59.8	14.73	0.005			
Pure error	20.3	5	4.06					
Total	6955.16	19						
*DF = degree of freedom								

Table 5.7: Analysis of variance (ANOVA) for Cu(II) removal prediction using the RSM model.

The initial Cu(II) concentration and sorbent dosage, X_2 and X_3 , both had a significant effect, while the second-order effects of sorbent dosage (X_3^2) on Cu(II) percentage removal has the highest significant effect among the remaining second-order effects. The negative value of the main effect coefficient, initial Cu(II) concentration, demonstrates that Cu(II) percentage removal decreased alongside an increasing initial Cu(II) concentration. Plus, the negative coefficient of the second order parameters, shows a maximum value in response within selected range of the parameters, which showed that large amount of sorbent dosage will give higher removal of Cu(II) from the water matrix (Shojaeimehr *et al.*, 2014). Source contribution in removal percentage was evaluated based on the adjusted sum of square and the plot for this is shown in Figure 5.5. The plot shows that sorbent dosage and initial Cu(II) concentration contributed more than 99 % to influencing the removal percentage, while temperature had a less than 1

% contribution on influencing Cu(II) removal. This result may differ from other reported studies due to the reported temperature range having been much higher and at an unrealistic range in the WWTP. The removal of copper and lead ions from a clean water matrix by wood sawdust, for example, has been studied via batch experiments at a temperature range of 26 - 56 °C (Ofomaja *et al.*, 2010b).

5.4.2.1 Thermodynamic study

In order to confirm the effect of temperature on the removal of Cu(II) from the water matrix, thermodynamic analysis were applied to study the relationship between rate constant (k) and temperature. The relationship between rate constant and temperature is represented by the Arrhenius equation:

$$\ln k = \ln A - \frac{E_a}{RT} \tag{5.2}$$

Where k is the rate constant obtained at different temperatures with the best-fit kinetic model, E_a (kJ mol-1) is the activation energy, A (g/mg.min), is the Arrhenius factor, R, is the gas constant (R=8.314 J/mol.K) and T(K) is the solution temperature.

According to the Arrhenius equation, the value of activation energy (E_a) calculated from the slope of the line is 4.99 kJ/mol ($R^2 = 0.6948$). This indicated that the temperature had a non-significant effect on the rate constant and the adsorption has a low potential barrier. Moreover, it is also implying that adsorption occurs more readily compared to typical chemical reactions in which E_a is in the range 65–250 kJ/mol.

The results of these experiments showed that copper adsorption was found to increase with temperature, indicating that the adsorption process is endothermic. The first order effects of initial Cu(II) concentration and sorbent dosage, X_2 and X_3 , were highly significant compared to the first order effect of temperature (X₁). A possible explanation for this result is that maximum percentage Cu(II) removal is only achieved with high sorbent dosage and low initial Cu(II) concentration.

The results of these experiments showed that copper adsorption was found to increase alongside an increasing temperature, indicating that the adsorption process was endothermic. The first order effects of initial Cu(II) concentration and sorbent dosage, X_2 and X_3 , were highly significant compared to the first order effect of temperature (X₁). A possible explanation for this result is that maximum percentage Cu(II) removal is only achieved with high sorbent dosage and low initial Cu(II) concentrations.



Figure 5.5: Source distribution in the removal percentage (adjusted sum of squares) versus source.

5.4.3 ANN MODEL

Since RSM showed a large residual error (Figure 5.4), which influenced the value of R², an ANN-based model was also developed for describing the removal of Cu(II) by oxidised CNW adsorbents. Similar to RSM modelling, the data generated through CCD were used to determine optimal architecture for the ANN model. A total of 20 experiments was divided into three subsets comprising training (12 data points), validation (4 data points) and testing (4 data points) points. The aim of splitting data into three subsets was to measure the capability of the model for the prediction of unseen experiments that were not used for training. Thus, the overview performance of ANN model can be assessed.

The data for the comparison between the observed and predicted values of Cu(II) removal are presented in Table 5.8. The actual and predicted percentage removal of Cu(II) by the ANN model is presented in Figure 5.6 and the coefficient of determination R^2 was found to be 0.9925, showing good agreement between the two sets of results.



Figure 5.6: The experimentally obtained result for Cu(II) removal compared to that predicted by the artificial neural network (ANN).

Run Number -	Coded values			Uncoded values				Sorption capacity			
	X_1	X ₂	X ₃	X_1	X ₂	X ₃	Experimental	Predicted	Residual	Error (%)	q (mg/g)
1	1	1	-1	21.3	50.31	2.10	44.50	44.75	-0.25	-0.56	10.63
2	1	-1	1	21.3	19.69	8.10	95.72	96.06	-0.34	-0.35	2.47
3	-1	-1	-1	9.7	19.69	2.10	74.81	74.77	0.04	0.05	6.53
4	0	0	0	15.5	35.00	5.10	76.95	77.45	-0.49	-0.64	5.05
5	0	0	0	15.5	35.00	5.10	76.35	77.45	-1.10	-1.44	4.95
6	-1	1	1	9.7	50.31	8.10	77.80	77.66	0.14	0.18	4.56
7	1	-1	-1	21.3	19.69	2.10	72.54	70.00	2.54	3.50	5.75
8	0	0	0	15.5	35.00	5.10	78.97	77.45	1.52	1.93	5.14
9	0	0	0	15.5	35.00	5.10	75.68	77.45	-1.77	-2.34	4.78
10	-1	-1	1	9.7	19.69	8.10	93.24	98.27	-5.03	-5.40	2.09
11	-1	1	-1	9.7	50.31	2.10	42.93	42.89	0.03	0.08	9.63
12	1	1	1	21.3	50.31	8.10	81.94	81.59	0.35	0.43	5.30
13	-α	0	0	6.0	35.00	5.10	74.33	75.18	-0.84	-1.13	4.77
14	0	0	α	15.5	35.00	10.00	88.65	88.84	-0.19	-0.21	2.99
15	0	0	0	15.5	35.00	5.10	78.08	77.45	0.63	0.81	4.72
16	0	α	0	15.5	60.00	5.10	58.26	58.03	0.23	0.39	6.50
17	α	0	0	25.0	35.00	5.10	78.33	78.15	0.17	0.22	5.07
18	0	-α	0	15.5	10.00	5.10	92.85	93.82	-0.97	-1.04	1.58
19	0	0	0	15.5	35.00	5.10	81.20	77.45	3.75	4.62	5.07
20	0	0	-α	15.5	35.00	0.20	16.81	16.86	-0.05	-0.28	20.67
X ₁ =temperature (°C); X ₂ =initial Cu(II) ion concentration (mg/L); X ₃ =sorbent dosage (g/L)											

Table 5.8: The experimentally obtained percentage Cu(II) removal compared to that predicted by the artificial neural network (ANN)

For better graphical interpretation of the Cu(II) adsorption process, threedimensional response surface plots were generated. Figure 5.7 shows the effect of the investigated parameters on the removal of Cu(II), with one of the three parameters held constant at its intermediate value (15.5 °C, 35 mg/L, or 5.10 g/L).



Figure 5.7: Surface plots (left) and corresponding contour plots (right) showing the effects of adsorption parameters on the Cu(II) removal as predicted by the ANN model, with temperature held constant at 15.5°C (A), initial Cu(II) concentration held constant 35 mg/L (B) and sorbent dosage held constant at 5.10 g/L (C).

Figure 5.7(A) shows the response of the Cu(II) removal when varying the initial Cu(II) concentration and sorbent dosage. Generally, ANN-based data analysis indicated that a high amount of sorbent dosage and a low initial Cu(II) ion concentration increased the percentage removal of Cu(II) from the water matrix, due to the availability of adsorption sites. As can been seen from Figure 5.7 (A), the percentage removal of Cu(II) increased with raising the sorbent dosage up to 8 g/L, and then remained constant with further increases. These results are consistent with findings reported in the literature (Geyikci *et al.*, 2012; Turan *et al.*, 2011). The increase in the percentage removal when the sorbent dosage increased is due to the concentration gradient acting as a driving force between the solute concentration and that adsorbed onto the surface of the oxidised CNW adsorbents.

Figure 5.7 (B) shows that variations of temperature only had a slight effect on Cu(II) removal. This indicates that higher temperature values did not damage the active sites in the sorbent or weaken the adsorptive force between the active sites of the adsorbent and Cu(II) ions significantly, at least not in the temperature range studied in this work.

Figure 5.7 (C) shows that the percentage removal of Cu(II) decreases at a higher initial Cu(II) concentration, since there will be a relative decrease in the available active metal binding sites per Cu(II) ion for adsorption. At a fixed sorbent dosage, there was a decrease in the percentage removal with further increments of the initial Cu(II) concentration, due to saturation of the CNW surface with Cu(II) ions. On the other hand, change in temperature had no significant effect on the response, due to the narrow range of temperatures investigated in this study. This result differs from other studies and can likely be explained by the fact that experiments in the literature had been conducted under temperatures that are not realistic for a wastewater treatment environment (Bingol *et al.*, 2012; Shojaeimehr *et al.*, 2014). Cao *et al.* (2014) presented the effects of temperature (10 – 50 °C) and showed that adsorption of Cr (VI) increased with increasing temperatures even when the range was higher and unrealistic conditions to a wastewater treatment.

5.5 STATISTICAL COMPARISON AND PERFORMANCE OF RSM AND ANN MODELS

In order to test the predictability of the RSM and ANN models, the distribution of residuals and errors of the two models were compared. The residuals and error distribution obtained from the prediction of RSM and ANN are presented in Figure 5.8 and 5.9, respectively. From the results, it can be seen that fluctuations in the residuals and errors were relatively small for the ANN model compared to RSM. In one of the 20 experiments (for the RSM model), point A (0.2 g/L sorbent dosage) is the farthest from the centre of the system. As noted previously in Section 5.4.2, a lower sorbent dosage (0.2 g/L) will impact the adsorption process and lead to a lower R² value, which in turn will affect the estimation capabilities of the RSM model. These results (for both RSM and ANN models) correspond well with the work of Geyikci *et al.* (2012), who indicated the RSM model (R² = 0.898) prediction as having greater deviation than the prediction by the ANN model (R² = 0.672).



Experimental runs

Figure 5.8: Plot representing the residuals obtained from the predictions of RSM and ANN modelling.


Figure 5.9: Plot representing error distribution obtained from the predictions of RSM and ANN modelling.

In addition to examining the poorness of fit using ANOVA for the RSM model (as determined in Section 5.4.1), the abilities of the ANN and RSM models in predicting Cu(II) removal from the water matrix were statistically evaluated in terms of the coefficient of determination (\mathbb{R}^2), absolute average deviation (AAD), and the root mean squared error (RMSE) (as described in Section 3.9.1). The AAD and RMSE for the RSM model were calculated to be 7.07 % and 3.99, whilst that of the ANN model was 1.15 % and 1.66. Moreover, the \mathbb{R}^2 for both models ($\mathbb{R}^2 = 0.9541$ for RSM, $\mathbb{R}^2 = 0.9925$ for ANN) showed that the ANN model made more accurate predictions than the RSM model. The statistical data obtained in this study matched well with other reported studies in which the ANN model showed a clear advantage over the RSM model for both data fitting and prediction capabilities (Geyikci *et al.*, 2012; Shojaeimehr *et al.*, 2014).

The main purpose of CCD is to provide high quality predictions for linear and quadratic interaction effects on the part of parameters that affect the adsorption process (Witek-Krowiak *et al.*, 2014). For further validation, an additional 14 unseen experiments were conducted in addition to those determined by the CCD; these consisted of combinations of experimental parameters not found in the training data set applied to the models. 3D scatter plots for the unseen experiments are shown in Figure 5.10, which includes a comparison of 3D scatter

plots derived from data contained in two other studies. In general, the limited number of studies that do conduct unseen experiments tend to use a limited number of experiments and the chosen unseen experiments do not represent the system of conditions, making it difficult to appropriately evaluate the predictive capability of the models. Figure 5.10 (B) indicates research by Ghosh *et al.* (2015), who conducted eight unseen experiments that did not represent an entire system, in order to study the validity of the RSM and ANN models. Bingol *et al.* (2012), indicated in Figure 5.10 (C), tested the validity of the models by conducting 11 new trials, all of which focused on one side of the system.

The 14 unseen experiments undertaken in this present study and illustrated in Figure 5.10 (A) were chosen to represent parameter space both inside and outside the tested parameter system, in order to provide a better understanding for testing the validity of the models. The actual and predicted values of the responses, along with their residual values for both models, are given in Table 5.9. The prediction abilities of the newly constructed ANN and RSM models were statistically measured in terms of R^2 , AAD and RMSE. Table 5.10 shows a statistical comparison of both models, based on the 20 CCD and unseen experiments that represent both inside and outside of the system. From the results, it is confirmed that the ANN model predicts more accurately than the RSM model, both the original 20 CCD and 14 unseen experiments.



Figure 5.10: 3D scatter plots showing a comparison of the CCD with unseen experiments within the systems (A) for this work; (B) for Ghosh et al., (2015); (C) Bingol *et al.* (2012)

Data		Temperature	Initial Cu(II)	Sorbent	Sorption	Cu(II)	AN	JN	RSM	
index	Run	n (°C)	concentration (mg/L)	dosage (g/L)	capacity (mg/g)	removal (%)	Predicted	Residual	Predicted	Residual
	1	21.3	50.31	5.1	5.27	62.21	66.82	-4.61	69.35	-7.14
В	2	9.7	19.69	5.1	3.17	88.42	92.57	-8.51	89.45	-5.39
yste	3	15.5	35.00	8.1	4.23	82.39	89.09	-6.7	86.72	-4.33
the s	4	18	55.00	8.1	4.76	74.19	77.83	-3.64	79.56	-5.37
Within t	5	10	55.00	4.0	5.75	60.41	53.96	6.45	53.97	6.44
	6	20	35.00	9.5	1.23	82.64	86.87	-4.23	87.77	-5.13
	7	20	15.00	5.1	3.10	88.31	93.27	-4.96	92.67	-4.36
	8	6	10.00	8.1	1.13	91.37	100	-8.63	100	-8.63
В	9	25	60.00	8.1	5.57	70.43	79.34	-8.91	83.33	-12.9
syste	10	25	60.00	4.0	9.26	57.2	52.72	4.48	56.01	1.19
the s	11	10	10.00	4.0	1.84	92.23	100	-7.77	92.73	-0.50
Outside	12	6	35.00	2.1	5.58	55.75	63.49	-7.74	54.14	1.61
	13	6	20.00	2.1	3.21	86.24	88.62	-2.38	70.22	16.02
	14	10	10.00	2.1	2.32	90.02	94.3	-4.28	79.3	10.72

Table 5.9: Validation data for 14 unseen experiments (CCD).

Data inday	Coefficient of d	AAI	D (%)	RMSE		
Data muex	ANN	RSM	ANN	RSM	ANN	RSM
20 CCD	0.9925	0.9541	1.15	7.07	1.66	4.00
14 Unseen	0.9374	0.7409	7.98	8.28	6.29	7.70
7 Inside	0.9530	0.9162	7.43	7.46	5.81	9.39
7 Outside	0.9395	0.6783	8.54	9.11	6.74	9.39

Table 5.10: Comparison of the predictive abilities of the RSM and ANN models.

Although both the RSM and ANN models provided good quality predictions (\mathbb{R}^2) for the parameters within the design range, the ANN model showed clear superiority over the RSM model for both data fitting and estimation capabilities for the 14 additional experiments (Figure 5.11). Moreover, the ANN model has the advantage of considering a wider range of adsorption conditions within a single experimental design (Witek-Krowiak *et al.*, 2014)



Figure 5.11: Comparison of the experimental and predicted results for unseen experiments between the RSM and ANN models.

The results found in the current study were similar to findings in other research, which have also shown the ANN approach to be more appropriate for the modelling of sorption by cellulose adsorbent than the RSM model, based on the CCD (Bingol *et al.*, 2012; Shojaeimehr *et al.*, 2014). As such, the ANN model is more flexible and predictable, which allows for the addition of a new set of

experiments in order to build a new dependable model. The reason for this is that the RSM model is limited in terms of only assuming quadratic non-linear correlation; the ANN model overcomes this limitation, since it can inherently capture almost any complex and non-linear process (Bingol *et al.*, 2012; Geyikci *et al.*, 2012).

5.6 ADSORPTION ISOTHERM

In order to understand the behaviour of adsorbent, two common adsorption isotherms, Langmuir and Freundlich isotherm models (referred to in Equation 3.5 and 3.6 in Section 3.5.4.4) were used to evaluate the adsorption system design and to describe the adsorption capacity of the oxidised adsorbent (Shojaeimehr *et al.*, 2014). Batch adsorption was carried out at a constant sorbent dosage by varying the concentration of Cu(II).

From the results (Appendix C), the coefficient of determination (R^2) showed that the Langmuir isotherm ($R^2 = 0.9998$) had a better correlation than the Freundlich isotherm ($R^2 = 0.9461$), which in turn indicated the homogeneous distribution of active sites on the adsorbent surface. The maximum monolayer adsorption capacity, as obtained from the Langmuir isotherm, was found to be 14.65 mg/g, and the Langmuir constant (K_L) was 1.4025 L/mg. Langmuir constant (K_L) is referred to the bonding energy of sorption, where the higher the magnitude of Langmuir constant (K_L), the stronger the bond formed. The equilibrium parameter (R_L), was calculated at different initial concentration of Cu(II) solution using the Equation 2.3. The calculated R_L were found to be in the range of (0-1) at all initial Cu(II) concentrations which confirms the favourable uptake of Cu(II) process according to Langmuir isotherm, as discussed in Section 2.3.2.1.

These results are in agreement with work using macrofungus (*Amanita rubescens*) for the adsorption of Pb(II) and Cd(II) ions from aqueous solution (Sarı and Tuzen, 2009). In addition, copper ion sorption by wood sawdust from synthetic wastewater was also better described by the Langmuir isotherm (Ofomaja *et al.*, 2010b).

The value of the Freundlich constant ($K_F = 2.3329$) is an indicator of adsorption capacity, which indicates affinity of Cu(II) species. Furthermore, the value of

Freundlich parameters also reflects the number of adsorption sites. Moreover, the n value from the Freundlich isotherm was 5.19 ($1 \le n \le 10$), indicating adsorption is favourable for the studied concentration range (Singh *et al.*, 2010). These observations correspond well with those of Sarkar and Majumdar (2011), who used oxidised chitosan beads for the removal of Cu(II) from a clean water matrix (Sarkar and Majumdar, 2011).

In addition, the comparison of Cu(II) sorption performance is better when based on a complete Cu(II) sorption isotherm curve (Calero *et al.*, 2011). Therefore, the q_m value, which was obtained from the Langmuir isotherm, was compared with other sorbents reported in the literature, based on their maximum adsorption capacity of Cu(II) ions (Table 5.11). It is important to emphasise that a direct comparison of the q_m from this study with q_m of other sorbents is challenging, due to experimental conditions not being comparable.

Although the reported Cu(II) adsorption capacity of oxidised CNW adsorbents was relatively smaller than for a number of adsorbents in other studies, the experimental conditions in this study were informed by actual conditions in the wastewater treatment process. For example, one of the important factors affecting adsorption capacities is the pH of the Cu(II) in water. Other studies have commonly conducted evaluations in a pH range of 4.0 - 5.0, which, as an acidic environment, is not representative of the actual conditions in a WWTP but suitable for certain heavy metal For example, the optimum pH for the adsorption of Cr(VI) by rice husk was at pH 4 (Rao *et al.*, 2012). This is because at acidic pHs, the surface of adsorbent is highly protonated. The protonated form of adsorbent can form bond with chromate and dichromate anions by electrostatic attraction for high removal of Cr(VI) to occur.

5.7 ADSORPTION KINETICS

In order to study the controlling mechanisms of adsorption processes such as mass transfer and chemical reaction, Lagergren's first and second order models were used to test the experimental data, as described earlier in Section 3.5.6. The kinetics parameters obtained from both models are presented in Table 5.12 and Appendix C. The linear graphs obtained for first and second order equations are shown in Figure 5.12 and 5.13, respectively. From the results, it can be concluded from the coefficients of determination ($R^2 = 1.000$) that the adsorption mechanism of Cu(II) onto oxidised CNW adsorbents follow the pseudo second order kinetic model. Moreover, the calculated q_e value was in good agreement with experimental q_e for the pseudo second order kinetic model.

		Experimenta	al conditions		a			
Adsorbents	рН	T (°C)	Ci (mg/L)	Time (min)	q_m (mg/g)	References		
Oxidised CNW	6	10	10-60	30	14.65	This study		
Cellulose graft polymers	4	20	200	300	17.16	(Guclu et al., 2003)		
Commercial resins (Duolite GT-73)	5	*	*	*	61.63	(Vaughan et al., 2001)		
Cotton stalks	3	25	20-200	30	4.0	(Nada et al., 2006)		
Granular activated carbon	5	30	*	*	5.08	(An et al., 2001)		
Olive pomace	6.5	20	50-200	60	1.0-5.0	(Pagnanelli et al., 2003)		
Starch-graft-acrylic acid copolymers	4	20	200	300	16.52	(Guclu et al., 2003)		
*: not reported; C_i = Initial concentration; q_m = maximum adsorption capacity.								

Table 5.11: Comparison of maximum adsorption capacities of Cu(II) ions by different adsorbents.

Kinetic equation and parameter	Cu(II)	
Experimental		
Initial Cu(II) concentration, Co (mg/L)	10.0	60.0
Adsorption capacity, $q_e (mg/g)$	1.91	9.19
First-order kinetic equation		
$k_1 (min^{-1})$	0.0207	0.0230
q1(mg/g)	0.3502	4.1468
\mathbb{R}^2	0.7863	0.8872
Second-order kinetic equation		
k ₂ (g/mg min)	0.2993	0.0191
q ₂ (mg/g)	1.91	9.35
R ²	1.0000	0.9996

Table 5.12: Adsorption kinetic parameters of Cu(II) onto oxidised CNW adsorbents.

 k_1 and $k_2 = 1^{st}$ and 2^{nd} order kinetics constant; q_1 and q_2 = adsorption capacity for 1^{st} and 2^{nd} order kinetics.

This model predicts adsorbent behaviour over the entire range of adsorption and is in agreement with chemical sorption being the rate-controlling step. In chemisorption (chemical adsorption), the process involves valence forces through sharing or exchange of electrons between the metal ions and the adsorbent (Lu *et al.*, 2009). The chemical bonding between divalent metal ions (copper) and polar functional groups (carboxyl groups) are responsible for the cation-exchange capacity of the adsorbent (CNWs). Similar results, where the adsorption mechanism follows the pseudo second-order kinetic model, were reported by other researchers in the context of the adsorption of Cu(II) onto the different adsorbents (Mata *et al.*, 2008; Ofomaja *et al.*, 2010b).



Figure 5.12: Linear plot, first-order rate equation, using oxidised CNW adsorbents (sorbent dosage = 4 g/L, temperature = 25 °C, pH = 6)



Figure 5.13: Linear plot, second-order rate equation, using oxidised CNW adsorbents (sorbent dosage = 4 g/L, temperature = 25 °C, pH = 6).

CHAPTER 6: RESULTS AND DISCUSSION UNDERSTANDING THE WASTEWATER MATRIX AND CHARACTERISATION

6.1 INTRODUCTION

The previous chapter described the removal of Cu(II) from a clean water matrix (Milli-Q ultrapure water) by oxidised CNW adsorbents using mathematical models. However, in a real-world wastewater treatment plant (WWTP), the water matrix will be far more complex. The composition of wastewater is typically a complex water matrix, often containing a variety of organic and inorganic compounds. In addition, wastewater varies in composition both temporally and spatially (within a WWTP and between different wastewater streams). The variation in wastewater composition may contribute to the inaccuracies and decreased predictive capabilities of models for the optimisation of remediation of copper from the wastewater matrix. Therefore, the main objective for this chapter is to study and determine not only the concentrations of heavy metals and their variations in wastewater samples, but also the physicochemical properties of the wastewater as characterised by Water Quality Practices.

This chapter is divided into four sections. The first section is focused essentially on the correct laboratory practice required, prior to ICP-MS instrumental analysis, to ensure robust analytical data. The second section provides the results of the analysis of heavy metal concentrations and their variability in the influent and effluent samples. The Water Quality Practices, including the physicochemical parameters, are described in third section. The final section describes benchmark experiments that basically inform the effects of wastewater matrix characteristics (wastewater composition and its variability) on the adsorbent's capability to remediate Cu(II) from this matrix.

6.2 HIGHLIGHTS

- Proper laboratory practice to ensure robust analytical data
- Report of physicochemical data analysis of influent and effluent wastewater samples.
- Presentation of heavy metal concentrations and their variations in influent and effluent wastewaters.
- Benchmark experiments to provide better understanding of the impacts of wastewater matrix on the remediation of Cu(II) from the wastewater matrix.

6.3 RESULTS AND DISCUSSION

6.3.1 SOURCES OF CONTAMINATION OF ICP-MS INSTRUMENTAL ANALYSIS

The analysis of metals with low detection limits involving concentrations of parts per billion or parts per trillion is a difficult and complicated process, requiring extra care during sample preparation (Zougagh *et al.*, 2002). Table 6.1 shows the limit of detection (LoD) and limit of quantification (LoQ) for each element, determined by ICP-MS (Thermo-Fisher iCAP-Q) equipped with CCTED (collision cell technology with energy discrimination). LoD and LoQ for ICP-MS were determined in a similar way as the employed for these limits by atomic absorption spectrometry (AAS) where LoD is defined as three times the standard deviation of the calibration blank (Section 3.6.3).

Element	Unit	LoD	LoQ
Al	μg/L	0.036	0.121
Ca	mg/L	0.011	0.036
Cd	μg/L	0.002	0.008
Cu	μg/L	0.001	0.003
Fe	μg/L	0.153	0.509
K	mg/L	0.003	0.012
Mg	mg/L	0.001	0.004
Mn	μg/L	0.002	0.008
Na	mg/L	0.001	0.003
Ni	μg/L	0.017	0.056
Pb	μg/L	0.002	0.005
Zn	μg/L	0.058	0.193

Table 6.1: Detection capabilities of ICP-MS for wastewater analysis

In order to ensure an accurate determination of low concentration metals, it is necessary to minimise errors by avoiding any sources of contamination during all stages of experimental procedure, from sample transport to sample preparation and analysis. The potential sources of contamination are listed below:

- Sampling equipment
- Storage containers
- Preservation reagents
- Laboratory environment
- Instrumentation

In order to evaluate the efficiency of the washing procedure and the impurity levels in Milli-Q ultrapure water (18.2 M Ω -cm resistivity) and any chemical reagents used, several control experiments were conducted, the samples being analysed by ICP-MS. Results presented in Table 6.2 show changes in concentration of 12 selected elements in Milli-Q water during handling in the laboratory. The purity of reagents and chemicals used in this study were evaluated and the results summarised in Table 6.3. Since Milli-Q water has been used as a reagent blank, the purity of the Milli-Q water is important, in order to

avoid contamination during sample preparation and analysis. The impurity levels in Milli-Q water were evaluated and the results show that Milli-Q water acidified by HNO₃ (2% by volume) has similar purity to commercial high-purity water. However, the concentration of Zn was detected at a higher level compared to other elements in Milli-Q ultrapure water. This may be attributed to contamination from the paper towels used during the sample preparation. Besides Zn, paper towels generally contain trace levels of transition metals such as Pb, Cr and Co (Thomas, 2013). Moreover, powder gloves should be avoided during the sample preparation as powder in these gloves contain high concentration of Zn (Kay, 2004). Therefore, powder-free or non-latex gloves are recommended when handling the equipment, samples, blank and standard solutions.

Moreover, the quality and selection of acids are important when dealing with analysis at low detection limits (ng/g and pg/g) (Rodushkin *et al.*, 2010). The main purpose of acidification is to avoid any metals being adsorbed onto the container walls (Balaram, 2005). Therefore, HNO₃ was used for sample preparation and acid washing, as other acids such as H₂SO₄ and H₃PO₄ are not suitable for analytical analysis by ICP-MS. This is because these acids do not completely decompose in the plasma and can adhere to the interface components and ion lenses causing signal instability (Rodushkin *et al.*, 2010). Also, insoluble metal sulphates or phosphate may form (depending on the acid used) and thus change the dissolved relevant metal ion concentration. Furthermore, these acids have low purity levels compared to HNO₃ and HCl, causing further contamination of the samples (Balaram, 2005).

According to Rodushkin *et al.*, (2010), sample containers and glassware are potential sources of contamination that require appropriate acid washing with dilute nitric acid prior to use. The procedure is fully described in Section 3.2.3. The acid-washed laboratory glassware showed a significant decrease in the contaminations and concentrations of selected elements. Also, minimising the number of laboratory items used during the experimental process, and keeping those washed items filled with acid, helps to reduce contamination.

The results (Table 6.2) show that acid-washed polypropylene sample tubes show lower levels of contamination compared to acid-washed glassware. These results

are in agreement with the work of Rodushkin *et al.*, (2010) who suggested that disposable tubes made of polystyrene or polypropylene are much more suitable for ICP-MS sample preparation. Moreover, glass sample containers should be avoided, as trace levels of some metals in water commonly adsorb onto the glass wall, thus releasing measureable concentrations of metals when acidic solutions are added. Moreover, a study assessing metal contaminations leaching from a series of plastic recycling bottles during various treatments has been reported (Cheng *et al.*, 2010). The results revealed that heating and microwaving significantly enhance antimony leaching from polyethylene terephthalate (PET) bottles. Therefore, polypropylene and other plastic materials are more suitable for sample storage. Similar sampling procedures were observed in other studies, where the collected wastewater samples were stored in polypropylene containers prior to metal analysis (Christophe *et al.*, 2011).

		Lab-ware and glassware							
Elements	Unit	Unwashed glassware	Acid-washed glassware	Unwashed sample tube	Acid-washed sample tube	Syringe filter	Pipette tips		
Al	μg/L	4.9 ± 1.8	0.4 ± 0.6	9.8 ± 1.0	1.3 ± 0.1	0.43 ± 0.04	0.02 ± 0.02		
Ca	mg/L	0.05 ± 0.02	0.02 ± 0.01	0.031 ± 0.002	0.007 ± 0.004	0.004 ± 0.008	0.003 ± 0.002		
Cd	μg/L	0.04 ± 0.03	0.03 ± 0.02	0.002 ± 0.001	0.002 ± 0.001	0.003 ± 0.001	0.01 ± 0.01		
Cu	μg/L	7.5 ± 1.0	0.07 ± 0.04	0.05 ± 0.01	0.01 ± 0.03	0.051 ± 0.001	0.04 ± 0.05		
Fe	μg/L	0.4 ± 0.1	0.7 ± 0.6	0.36 ± 0.05	0.27 ± 0.03	0.05 ± 0.01	0.059 ± 0.003		
Κ	mg/L	0.03 ± 0.02	0.004 ± 0.005	0.004 ± 0.001	0.005 ± 0.001	0.007 ± 0.001	0.005 ± 0.001		
Mg	mg/L	0.01 ± 0.02	0.001 ± 0.001	0.003 ± 0.001	0.001 ± 0.001	0.002 ± 0.003	0.001 ± 0.001		
Mn	µg/L	0.04 ± 0.02	0.03 ± 0.01	0.02 ± 0.01	0.003 ± 0.001	0.02 ± 0.01	0.04 ± 0.02		
Na	mg/L	0.09 ± 0.01	0.003 ± 0.007	0.003 ± 0.001	0.003 ± 0.001	0.83 ± 0.02	0.001 ± 0.001		
Ni	μg/L	0.14 ± 0.04	0.36 ± 0.02	0.6 ± 0.1	0.02 ± 0.02	0.012 ± 0.003	0.213 ± 0.003		
Pb	µg/L	0.06 ± 0.01	0.06 ± 0.02	0.039 ± 0.001	0.028 ± 0.002	0.013 ± 0.001	0.038 ± 0.002		
Zn	μg/L	5.6 ± 1.3	2.3 ± 0.3	1.3 ± 0.2	0.7 ± 0.3	0.3 ± 0.5	0.38 ± 0.02		

Table 6.2: Concentration of selected elements in Milli-Q water from different control experiments

Values represent mean of three replicates \pm standard deviation

				Reagents and chemicals					
Elements	Unit	Milli-Q water	Milli-Q water, 2% HNO ₃	High purity water	50 µg/L Cu standard	50 μg/L Cu standard with 2% HNO ₃			
Al	μg/L	0.22 ± 0.04	0.1 ± 0.1	0.4 ± 0.4	1.2 ± 1.0	0.6 ± 0.9			
Ca	mg/L	0.004 ± 0.001	0.005 ± 0.003	0.003 ± 0.003	0.08 ± 0.02	0.02 ± 0.01			
Cd	μg/L	0.017 ± 0.014	0.002 ± 0.001	0.002 ± 0.001	0.007 ± 0.002	0.039 ± 0.002			
Cu	μg/L	0.05 ± 0.05	0.04 ± 0.02	0.095 ± 0.003	36.9 ± 1.2	0.8 ± 0.1			
Fe	μg/L	0.606 ± 0.004	0.40 ± 0.04	0.2 ± 0.2	0.463 ± 0.002	0.02 ± 0.02			
Κ	mg/L	0.007 ± 0.001	0.005 ± 0.001	0.005 ± 0.001	0.009 ± 0.002	0.009 ± 0.003			
Mg	mg/L	0.002 ± 0.001	0.001 ± 0.001	0.001 ± 0.001	0.001 ± 0.001	0.002 ± 0.001			
Mn	μg/L	0.006 ± 0.003	0.005 ± 0.006	0.005 ± 0.002	0.26 ± 0.05	0.03 ± 0.01			
Na	mg/L	0.003 ± 0.001	0.001 ± 0.001	0.03 ± 0.01	0.05 ± 0.01	0.03 ± 0.02			
Ni	μg/L	0.13 ± 0.02	0.19 ± 0.04	0.03 ± 0.03	0.6 ± 0.3	0.4 ± 0.2			
Pb	μg/L	0.009 ± 0.003	0.028 ± 0.005	0.02 ± 0.01	0.04 ± 0.01	0.04 ± 0.01			
Zn	μg/L	0.33 ± 0.01	0.3 ± 0.8	0.297 ± 0.004	3.4 ± 0.8	2.3 ± 1.6			
Values repres	Values represent mean of three replicates ± standard deviation								

Table 6.3: Concentration of selected elements in reagents and chemicals

6.3.2 THE INFLUENCE OF STORAGE TIME

The influence of storage time on heavy metal concentration is important, as ICP-MS analysis can take several days (up to two days) due to instrument accessibility. The stability of wastewater effluent was tested at different time intervals, prior to analysis by ICP-MS. As described in Section 3.2.4.1, monitoring the influence of storage time on the effluent was necessary, as it allowed for the observation of any changes in wastewater quality during storage, normally between 0 and 120 hours (APHA, 1998). However, the impact of wastewater storage time on water quality parameters is not necessary, as the adsorption process to remediate Cu(II) from the wastewater matrix is performed using fresh effluent wastewater.

As shown in Table 6.4, the effluent samples do not show any significant changes in heavy metal concentration during storage at 4 °C for 72 or 120 hours. This demonstrates that storage under these conditions does not affect the stability of heavy metals. Besides proper storage, suitable sample containers, and acidification to a pH below 2.0, have been used in this study, to minimise the precipitation and adsorption of heavy metals on container walls.

Heavy motals $(u \sigma/I)$	Storage time (hours)						
The avy metals ($\mu g/L$)	0	72	120				
Al	3.83 ± 0.11	3.83 ± 0.01	3.56 ± 0.03				
Cd	0.03 ± 0.01	0.03 ± 0.01	0.03 ± 0.01				
Cu	3.65 ± 0.05	3.58 ± 0.07	3.59 ± 0.04				
Fe	40.74 ± 0.73	41.05 ± 0.02	40.58 ± 0.78				
Ni	3.46 ± 0.06	3.46 ± 0.05	3.47 ± 0.02				
Pb	0.26 ± 0.01	0.26 ± 0.01	0.26 ± 0.02				
Viles we we set the set of the se							

Table 6.4: Comparison of selected heavy metal concentrations after 0, 72 and 120 hours storage of effluent samples

Values represent mean of three replicates \pm standard deviation

6.3.3 PHYSICOCHEMICAL DATA ANALYSIS OF WATER SAMPLES

Wastewater characterisation by various water quality parameters (e.g. pH and conductivity) is required in order to meet legislation requirements. As described in Section 3.2.4, influent and effluent wastewater samples were collected from Severn Trent Water Stoke Bardolph wastewater treatment plant (WWTP), for a period of three months (January to March 2016). Samples were characterised for pH, total dissolved (TDS) and total suspended solids (TSS), conductivity, and chemical oxygen demand (COD) in the laboratory, while dissolved oxygen (DO) and temperature were analysed at the sampling points. The analytical methods used for the water quality analysis were in accordance with "Standard Methods for the Examination of Water and Wastewater" in order to meet legislation requirements (APHA, 1998). The results of the physicochemical parameters for influent and effluent water samples are shown in Table 6.5. These data represent a total of 10 wastewater samples for both influent and effluent at different sampling periods, with hydraulic retention time (HRT) of 17 hours. The effect of HRT on pollutant removal performance is important due to its impact on the reduction of organic matter and nutrients (Merino-Solís et al., 2015).

Dissolved oxygen (DO) is a measure of the content of molecular oxygen present in water. The concentration of dissolved oxygen is a necessary control factor in wastewater, as it favours the organisms desired during the aerobic process (Wiese *et al.*, 1997). Low dissolved oxygen will decrease the activity of aerobic organisms and may cause sludge bulking, while an increase in dissolved oxygen may cause flocculation, with unsettled particles remaining in the wastewater (Singh *et al.*, 2012). The work by Chapman (1997) reported that the minimum DO value for supporting aquatic life is 4 - 5 mg/L; concentration below this value may adversely affect aquatic biological life, while concentrations below 2 mg/L may lead to death for most aquatic life (Chapman, 1997). The DO level for effluent samples (4.9 - 5.8 mg/L) were within the range for sustaining aquatic life, while the DO level for influent samples (1.2 - 5.3 mg/L) were found to be below the standard. Observations of temperature of wastewater are important as the rate of biological activity and solubility of oxygen depends on the temperature (Spellman, 2013). Figure 6.1 shows that the temperature of influent is lower than the effluent, which suggests the intrusion of storm water (Hwang *et al.*, 2015). As mentioned in Section 5.3.2.5, the realistic range for temperature in the wastewater treatment environment varies from 6 - 25 °C (Burton *et al.*, 2013; Hanaki, 2008). Therefore, the temperature obtained for both influent and effluent were within the expected temperature range.



Figure 6.1: Variation in temperature for sampled wastewater influent and effluent (February to March 2016) from Stoke Bardolph WWTP.

Besides temperature, pH, a measure of the concentration of hydrogen ions in a solution, is an important parameter in the operation of biological treatment units. An increase in pH is due not only to industrial or other non-domestic discharges, but also to the denitrification process (WEF, 2007). As shown in Table 6.5 and Figure 6.2, the pH of the samples were within the permissible limit for wastewater, ranging from 6.0 to 9.0 (CEC, 2006). Moreover, all pH values were also within the range set by the WHO for wastewater, ranging from 7.0 to 7.4 (WHO, 2006).

Daramatara	Unit	Discharge limit	Influer	nt	Effluent					
Falaineters	Ullit	Discharge mint –	Range	Mean	Range	Mean				
DO	$mg/L O_2$	5 – 9 ^a	1.2 – 5.3	3.8	4.9 - 5.8	5.4				
Temperature	°C	$6 - 25^{a}$	7.6 – 11.9	10.4	10.3 – 16.2	12.1				
рН		$6 - 9^{d}$	7.0 - 7.8	7.6	6.9 -7.4	7.0				
COD	mg/L	125 ^a	61.3 - 285	120	13.1 – 26.2	21.3				
TSS	mg/L	< 35 ^d	38.9 - 177.4	89.8	3.9 - 14.2	6.4				
TDS	g/L	0.3 -0.9 ^c	0.3 – 0.9	0.5	0.3 - 0.4	0.4				
Conductivity	mS/cm	$0.05 - 1.5^{a}$	0.7 - 2.0	1.0	0.6 – 0.9	0.8				
Sodium	mg/L	$40 - 70^{\circ}$	54.8 - 81.7	80.9	62.6 - 89.9	74.4				
Magnesium	mg/L	$4 - 10^{b}$	32.07 - 40.27	30.9	22.2 - 30.7	27.1				
Potassium	mg/L	$7 - 15^{b}$	12.5 - 21.0	19.7	14.4 - 28.2	18.4				
Calcium	Calcium mg/L 6-16 ^b 85.0-113.7 85.8 61.9-81.9 74.8									
^a (CEC, 2006), ^b (Tcl	^a (CEC, 2006), ^b (Tchobanoglous <i>et al.</i> , 1991), ^c (WHO, 2006), ^d (CEC, 1991)									
DO=dissolved oxyge	n; COD=chemica	al oxygen demand; TSS= total	suspended solid; TDS=tot	al dissolved solid						

Table 6.5: Characteristics of influent and effluent wastewater



Figure 6.2: Variation in pH for influent and effluent (February to March 2016)

The TSS values of the influent and effluent were found to be in the range of 38.9 - 177.4 and 3.9 - 14.2 mg/L, respectively. According to literature, the wastewater can be classified as follows: TSS < 100 mg/L as weak, 100 < TSS < 220 mg/L as medium, and TSS > 220 mg/L as strong wastewater (Osobamiro and Atewolara-Odule, 2015; Singh *et al.*, 2012). Therefore, TSS levels in influent and effluent samples classified them as medium and weak wastewater, respectively. Moreover, the permissible standard for TSS (for discharge) is 35 mg/L (CEC, 1991). Therefore, the effluent is safe to be discharged into the river, while the high TSS level in the influent, due to the suspended particles from the waste, may affect aquatic life. However, the mean values (0.5 g/L) obtained for TDS in both sampling points were within the permissible limits stipulated by the WHO for wastewater (WHO, 2006).

Conductivity measurements indicate the level of dissolved inorganic material present. The biological nitrogen removal in wastewater treatment is the main cause of the significant reduction in the conductivity of wastewater (Levlin, 2010). However, the increase in the conductivity measurement of the influent indicates an unusual discharge, probably from an industrial source. Moreover, the measurements obtained (Figure 6.3) were in agreement with other studies reported in the literature (Alonso *et al.*, 2004; Bhuiyan *et al.*, 2010). The conductivity of wastewater generally ranges from 0.05 - 1.50 mS/cm, while some

industrial wastewater is reported to have higher conductivity measurements, up to 10.0 mS/cm (WEF, 2007).



Figure 6.3: Variation in conductivity for influent and effluent (February to March 2016).

6.3.4 CHEMICAL OXYGEN DEMAND (COD)

Chemical oxygen demand (COD) is one of the standard parameters, and provides information on the level of organic contamination of wastewater (Bansode *et al.*, 2004). The procedure for the determination of COD using the cuvette test is fully described in Section 3.2.4.5. Figure 6.4 shows COD concentrations of influent and effluent samples collected at 10 different sampling periods.

The COD of influent published in literature was reported to be in the range of 200 to 600 mg/L (WEF, 2007). Although the results for influent were within this range, the COD concentrations of the influent during the sampling periods were highly variable. However, it was observed that the measurements of effluent samples consistently produced lower COD values than for influent samples.

COD levels of the effluent were stable for all sampling periods, at lower than 40 mg/L. This suggests that the high level of organic pollution due to anthropogenic activity may have resulted in high values of COD in the influent (Himanshu and Vashi, 2014). These results were in agreement with the work of Gardner *et al.*

(2012) who measured the characteristics of final effluent from wastewater treatment plants around the UK.



Figure 6.4: Variation in COD for influent and effluent (February to March 2016).

To summarise, variations in water quality parameters were observed in both influent and effluent wastewater samples. Table 6.6 summarises these water quality parameters for both influent and effluent wastewater samples.

Table 6.6: Data for water quality parameters for influent and effluent wastewater samples during February-March 2016

Parameters	Influent (average ± SD)	Effluent (average ± SD)
Temperature (°C)	10.4 ± 0.9	12.1 ± 1.4
Dissolved oxygen (DO) (mg/L)	3.8 ± 4.2	5.3 ± 4.6
рН	7.6 ± 0.2	7.0 ± 0.1
Conductivity (mS/cm)	1.0 ± 0.3	0.8 ± 0.1
Chemical oxygen demand (COD) (mg/L)	120.4 ± 51.6	21.3 ± 4.1
Total suspended solid (TSS) (mg/L)	89.8 ± 37.3	6.4 ± 2.8
Total dissolved solid (TDS) (g/L)	0.5 ± 0.1	0.4 ± 0.1
*SD=standard deviation		

6.3.5 HEAVY METAL CONCENTRATION AND ITS VARIATION IN INFLUENT AND EFFLUENT WASTEWATER SAMPLES

The concentrations of heavy metals such as copper (Cu), cadmium (Cd), chromium (Cr), iron (Fe), and lead (Pb) were determined using ICP-MS analysis of influent and effluent samples taken from 10 different sampling periods at the Severn Trent Water Stoke Bardolph WWTP.

The concentration of heavy metals in influent and effluent ranged from $0.01 - 439.05 \ \mu g/L$ and $0.006 - 99.82 \ \mu g/L$, respectively. As seen from Figures 6.5 to 6.9, variations in the concentrations of heavy metals were observed in both influent and effluent samples. These observations are consistent with the trend observed in other studies (Malakootian *et al.*, 2011; Sadrzadeh *et al.*, 2009). The highest concentration of Cu content in influent wastewater sample was found to be 12.34 μ g/L. However, this concentration is within the range for wastewater discharge contained in the Water Framework Directive (2000/60/EC). From the results obtained for copper concentration in influent and effluent wastewater samples, the selected range for initial Cu(II) concentration (10 – 60 mg/L) in the clean water matrix is not environmentally relevant (Chapter 5). However, the range chosen is closer to the copper concentration observed in some industrial wastewater, as shown in Table 2.4.



Figure 6.5: The concentrations of Cu in influent and effluent



Figure 6.6: The concentrations of Cr in influent and effluent.



Figure 6.7: The concentrations of Cd in influent and effluent



Figure 6.8: The concentrations of Fe in influent and effluent



Figure 6.9: The concentrations of Pb in influent and effluent

*Values represents mean of duplicates, where dotted lines represent duplicate samples

6.3.6 BENCHMARKING THE WASTEWATER MATRIX FROM THE POLLUTANT AND ADSORBENT PERSPECTIVE

New wastewater quality parameter (benchmark study) has been developed to quantify the impact of wastewater composition on the efficiency of Cu(II) removal by oxidised CNW adsorbents. The issue with previous studies is the assumption that the actual wastewater composition is the same for each experiment, or has no influence on the removal capability of that process. No work to date has performed benchmark experiments on each fresh wastewater sample to challenge that assumption.

Moreover, the impacts of the variability and composition of the wastewater matrix on the adsorbent's capability to remediate Cu(II) have not yet been reported by other studies. Thus, throughout these benchmark experiments, the complexity and variability of the wastewater to the treatment was controlled. The procedure followed in the benchmark experiments is described in Section 3.5.4. Benchmark experiments for each effluent wastewater sample were carried out under similar conditions and the percentage of Cu(II) removal by oxidised CNW adsorbents are reported in Table 6.7.

Table 6.7: Data over time of Cu(II) removal as evaluated by benchmark studies (Initial concentration of wastewater effluent spiked with Cu(II)=4.0 mg/L, sorbent dosage=1.0 g/L, pH=6.0)

Benchmark experiment for Cu(II) removal (%)
80.11
68.22
81.82
74.41
77.39
78.76
74.15
81.27
80.11
77.27

For these benchmark experiments, a sorbent dosage of 1 g/L was selected in order to study the impacts of the variability and compositions of the wastewater matrix on the adsorbent capability. If higher sorbent dosage was selected, the results for each wastewater sample will gave 100% Cu(II) removal, since effluent contained less contamination than influent wastewater. As the main aim of this benchmark experiment is to study the effect of wastewater matrix on the efficiency of Cu(II) removal by oxidised CNW adsorbents, it is important that the independent variable (benchmark experiment) varies for ANN modelling, in order to detect the variation of wastewater and the influence of this variable on the removal capability.

The results reported in Table 6.7 show that the percentage Cu(II) removal by oxidised CNW adsorbents varies for each wastewater sample on different sampling dates. Results over six weeks give an average $77.35 \pm 4.15\%$. Although the difference was low (around 4%), it is significant, and is likely to influence the efficiency and capability of oxidised CNW adsorbents. Moreover, the results obtained only support and covered only three months of sampling trips and focused on the effluent from only one treatment plant. According to the United State EPA, the complexity of the wastewater matrix depends on the number and volume of wastewater streams generated. The amount and type of contamination found in wastewater will depend on the different industrial activities realised in the area, the degree of urban development, and the type of treatment facilities found around it (USEPA, 1998).

The composition of wastewater in WWTPs depends on several factors, such as compound physico-chemical properties, the climate conditions (temperature and sunlight intensity), the type of treatment process employed, and the operational conditions of the treatment process (temperature of operation and hydraulic retention time) (Gracia-Lor *et al.*, 2012; Jelic *et al.*, 2011). Thus, variations in water quality parameters were observed in both influent and effluent wastewater samples, as shown in Table 6.6. For example, in the case of COD in influent and effluent, the COD concentrations of the influent during the sampling periods were highly variable. This indicated that the influent contained higher levels of organic pollutants than did the effluent wastewater, which may have influenced the capability of the adsorption process. Removal of Cu(II) from wastewater by an

adsorption process is dependent on the organic compound's biodegradability, its volatility, and its ability to be adsorbed onto adsorbents (Kushwaha *et al.*, 2010). Therefore, the composition of wastewater can vary significantly from plant to plant, between different wastewater streams, and within a plant at different times.

This demonstrates that the complexity of wastewater, in terms of its composition and variability, affect the capability of the adsorbent to remediate Cu(II) from the wastewater matrix. Moreover, the complexity and variation of wastewater composition may also affect the accuracy and efficiency of mathematical modelling in predicting the capability of this adsorbent to remediate spiked copper from the wastewater effluent. Therefore, this will be discussed further in next chapter, which will focus more on the remediation of Cu(II) from the wastewater matrix.

CHAPTER 7: RESULTS AND DISCUSSION PREDICTING THE CAPABILITY OF OXIDISED CNW ADSORBENTS FOR REMEDIATION OF SPIKED COPPER FROM WASTEWATER EFFLUENT USING RSM AND ANN MODELS

7.1 INTRODUCTION

Chapter 6 determined the complexity of wastewater and its variability in terms of heavy metal and water quality indicators both temporally and spatially. A benchmark study reported in the previous chapter (Section 6.3.6) determined that the wastewater matrix, which is variable and complex, has an impact on adsorbent capability and performance. With the results obtained in Chapter 6, a better understanding has been gained of the selection of variables and their ranges in the experimental design. In this chapter, the capability of oxidised cellulose nanowhisker (CNW) adsorbents for remediation of spiked Cu(II) from wastewater effluent is described. Similar to the design modelling approach applied to the clean water matrix in Chapter 5, RSM and ANN were used to develop an approach for the remediation of spiked Cu(II) from wastewater effluent. As remediation processes from wastewater are often complicated due to the variation in wastewater compositions, results obtained for ANN modelling, unlike in other optimisation studies.

7.2 HIGHLIGHTS

- This chapter focuses on remediation of spiked Cu(II) from wastewater effluent obtained from a real wastewater treatment plants (WWTPs) using oxidised CNW adsorbents.
- The variations in wastewater composition are included as independent variables for ANN modelling, unlike in other optimisation studies.
- ANN including variability has a superior prediction capability to RSM and ANN without variability for Cu(II) removal from wastewater effluent.

7.3 SCOPING EXPERIMENTS FOR EVALUATING REMOVAL PERFORMANCE IN THE WASTEWATER MATRIX

The purpose of the scoping studies is to identify the variables and parameter ranges that influence the adsorption process and help set the boundary conditions for the central composite design (CCD). The variables and parameter ranges used in this chapter are different from the clean water matrix, due to the variation in wastewater composition. The pH, sorbent dosage, and initial concentration of wastewater effluent spiked with Cu(II) are used as independent (input) variables and are studied for their impact on the removal of spiked Cu(II) from the wastewater effluent. These parameters are selected due to their effects in real WWTPs. Moreover, as the composition of wastewater varies with time, benchmark experiments were conducted for each sampling trip to study the effect of variation in wastewater composition (as mentioned in Section 3.5.3). For the same amount of sorbent dosage under similar conditions, different percentage removal was obtained for each benchmark experiment. This demonstrated that the variation in wastewater composition influences the efficiency of the adsorption process.

The contact time for the adsorption process from the effluent was chosen as 30 mins, in accordance with results obtained from scoping studies. This is because a further increase in the contact time did not show a significant change in percentage removal. Similar observations were observed for the remediation of Cu(II) from the clean water matrix, where adsorption equilibrium was achieved within 30 mins. It was found that adsorption increased sharply with contact time during the first 5 mins, contributing to more than 78% of Cu(II) removal. It then decreased slowly to reach a plateau and it was observed that 30 mins was sufficient to reach adsorption equilibrium.

In the case of the clean water matrix, the results showed that temperature has a less than 1 % contribution towards Cu(II) removal, while sorbent dosage and initial Cu(II) concentration contribute more than 99 % towards the removal percentage (Figure 5.5). Therefore, it was decided that the optimum temperature for the removal of Cu(II) from effluent using oxidised CNW adsorbents is 20°C.

This has been confirmed by new set of control experiments, where the maximum removal of Cu(II) was observed at 20 °C with 90 % of Cu(II) removed, while at 10 °C only 80 % Cu(II) removal was achieved.

As discussed in Chapter 5, it was decided that the optimum pH for the removal of Cu(II) ions from clean water matrix using oxidised CNW adsorbents is pH 6. However, in order to study the effect of pH of the wastewater effluent, a pH range (5 - 8) realistic for a wastewater treatment environment was applied. The pH range applied is similar to those of Baltpurvins *et al.*, (1996), who reported that higher initial metal concentrations (1000 mg/L) result in broader pH range (4.5 – 9.0) for metal precipitation. Furthermore, pH 5 is a realistic pH value for industrial wastewater, since the pH for electroplating wastewater is around this level (Chand *et al.*, 2015; Malakootian *et al.*, 2011; Rajemahadik *et al.*, 2013). For municipal wastewater (Stoke Bardolph WWTP), as reported in Chapter 6, the pH range was found to be in the range of 7 - 7.8. In addition, Fish Directive (2006/44/EC) also reported that the pH range for discharge of treated effluent water is in the range of 6 - 9 (CEC, 2006).

The sorbent dosage for the adsorption process was chosen to be in the range of 0.5 - 10.0g/L, in accordance with results obtained from the clean water matrix (Section 5.4.2). Therefore, pH 5 – 8, sorbent dosage 0.5 - 10 g/L, and initial concentration of wastewater effluent spiked with Cu(II) of 1 - 5 mg/L, were investigated for their effect on the efficiency of Cu(II) removal from wastewater effluent.

7.4 MATHEMATICAL MODELLING OF SPIKED CU(II) REMOVAL FROM WASTEWATER EFFLUENT

7.4.1 CENTRAL COMPOSITE DESIGN (CCD)

The pH, sorbent dosage, and initial concentration of wastewater effluent spiked with Cu(II) were used as independent (input) variables and were studied for their impact on the removal of spiked Cu(II) from wastewater effluent. The range of independent variables, with the levels of the experimental factors, is given in Table 7.1. A second-order polynomial equation was used to determine the relationships between variables and response, as the first-order model suffers from lack of fit due to interaction between variables and surface curvature. The final equation in terms of coded factors obtained by the application of RSM is given by:

$$Z (\%) = 81.4753 + 1.5759Y_1 + 4.1651Y_2 + 7.1588Y_3 + 0.9451Y_1^2$$

- 2.3417Y_2^2 - 2.3848Y_3^2 - 0.3212Y_1Y_2 - 0.3087Y_1Y_2 (7.1)
- 1.2187Y_2Y_3

where Z is the response variable (percentage removal of copper from wastewater effluent) and Y_1-Y_3 are the uncoded values of the independent variables (pH, sorbent dosage, and initial concentration of wastewater effluent spiked with Cu(II)).

Independent variable	Unit	Factor code	Range and level				
			-α	-1	0	+1	$+\alpha$
pН		Y_1	5.0	5.6	6.5	7.4	8.0
Sorbent dosage	g/L	Y_2	0.5	2.34	5.25	8.16	10.0
Initial concentration of wastewater effluent spiked with Cu(II)	mg/L	Y ₃	1.00	1.78	3.00	4.23	5.00

Table 7.1: CCD Experimental ranges and levels of independent variables

7.4.2 RESPONSE SURFACE METHODOLOGY (RSM)

Results for the percentage of Cu(II) removal from wastewater effluent spiked with Cu(II) were obtained by performing batch experiments according to the CCD matrix of conditions. Table 7.2 shows the experimental results obtained from the various runs, together with the values predicted by the built RSM model, with residual values in the range of 0.21 to 4.8, which influenced the value of R^2 . The percentage error (% error) was calculated as the ratio of the difference between experimental and predicted value to experimental value (Equation 3.11).

In order to test the suitability of the model, the predicted and actual experimental values were plotted (Figure 7.1) to provide the coefficient of determination ($R^2 = 0.9409$). The R^2 value in this study was low compared to other studies that used a clean water matrix for the adsorption process. The study of a cellulose-based adsorbent for chromium removal from a clean water matrix, for instance, showed a high coefficient of determination ($R^2 = 0.9959$) (Liu *et al.*, 2011). In addition, a study by Singh *et al.*, (2010) showed that the remediation of Cu(II) from a synthetic solution was successfully predicted by the RSM model. The experimental data fit well with the predicted values derived from this model, with a coefficient of determination R^2 of 0.9982.
Run	Co	ded val	ues	Un	coded val	ues					
Number	Y1	Y ₂	Y ₃	Y_1	Y ₂	Y ₃	Experimental	Predicted	Residual	Error (%)	Absolute error (%)
1	-1	+1	+1	5.6	8.16	4.23	86.05	86.64	-0.59	-0.69	0.69
2	-1	+1	-1	5.6	8.16	1.78	71.26	74.15	-2.89	-4.05	4.05
3	0	0	0	6.5	5.25	3.00	80.65	81.27	-0.62	-0.76	0.76
4	0	0	0	6.5	5.25	3.00	82.62	81.27	1.35	1.64	1.64
5	-1	-1	+1	5.6	2.34	4.23	80.32	80.11	0.21	0.26	0.26
6	+1	+1	+1	7.4	8.16	4.23	88.2	88.54	-0.34	-0.38	0.38
7	+1	-1	+1	7.4	2.34	4.23	83.62	83.29	0.33	0.40	0.40
8	-1	-1	-1	5.6	2.34	1.78	60.52	62.74	-2.22	-3.66	3.66
9	0	0	0	6.5	5.25	3.00	80.43	81.27	-0.84	-1.04	1.04
10	+1	+1	-1	7.4	8.16	1.78	74.51	77.27	-2.76	-3.71	3.71
11	+1	-1	-1	7.4	2.34	1.78	65.19	67.15	-1.96	-3.00	3.00
12	0	0	0	6.5	5.25	3.00	81.52	81.27	0.25	0.31	0.31
13	0	$+\alpha$	0	6.5	10	3.00	84.64	81.82	2.82	3.33	3.33
14	0	0	$+\alpha$	6.5	5.25	5.00	85.62	86.60	-0.98	-1.14	1.14
15	-α	0	0	5	5.25	3.00	83.36	81.21	2.15	2.58	2.58
16	0	0	-α	6.5	5.25	1.00	68.02	63.22	4.80	7.06	7.06
17	0	0	0	6.5	5.25	3.00	80.01	81.27	-1.26	-1.57	1.57
18	0	0	0	6.5	5.25	3.00	81.09	81.27	-0.18	-0.22	0.22
19	$+\alpha$	0	0	8	5.25	3.00	88.04	86.36	1.68	1.91	1.91
20	0	-α	0	6.5	0.5	3.00	69.23	68.22	1.01	1.46	1.46
Y1=pH; Y2	2=sorbe	ent dosa	ige; Y ₃ =	initial co	oncentrati	on of was	stewater effluent sp	iked with Cu(II)			

Table 7.2: The experimentally obtained for spiked Cu(II) removal from the wastewater effluent compared to that predicted by the response surface methodology (RSM)



Figure 7.1: The experimentally obtained remediation of spiked Cu(II) from wastewater effluent compared to that predicted by the response surface methodology (RSM)

Analysis of variance (ANOVA) for Cu(II) removal from the effluent was applied to evaluate the quality of fit of the model. The significance of each term in the equation to the percentage of the adsorbed Cu(II) ions was validated by this statistical test. The results of the second-order response surface model fitting in the form of ANOVA are shown in Table 7.3.

Generally, it can be considered that higher Fisher's F-test values and lower P values indicate the significance of the coefficients of the parameters. Values of P greater than 0.10 indicate that the model terms are not significant (Cao *et al.*, 2014). As seen from Table 7.3, all the first-order main effects in the quadratic model are statistically significant (P<0.05) for their effect on the Cu(II) percentage removal from the effluent. The positive first order coefficient indicates that adsorption process increased with increasing the variable.

However, the second-order effect of pH (Y_1^2) on the Cu(II) percentage removal are not significant among the other second-order effects. The negative coefficient of the second order parameters, shows a maximum value in response within selected range of the parameters, which showed that large amount of sorbent dosage and high initial concentration of wastewater effluent spiked with Cu(II), will result in higher removal of Cu(II) from the wastewater. However, it was observed from Table 7.3 that the coefficient for the first and second order effects were significant when compared with interactive effects (Y_1Y_2 , Y_1Y_3 , and Y_2Y_3) with (P>0.10).

Source	Sum of squares	DF	Mean square	F value	P-value Prob > F	Coefficient
Model	1120.08	9	124.45	22.72	< 0.0001	81.4753
Y_1	33.11	1	33.11	6.05	0.036	1.5759
Y ₂	231.31	1	231.30	42.23	< 0.0001	4.1651
Y ₃	683.31	1	683.31	124.76	< 0.0001	7.1588
Y_1^2	21.18	1	11.80	2.15	0.176	0.9451
Y_2^2	62.59	1	72.42	13.22	0.005	-2.3414
Y_3^2	75.11	1	75.11	13.71	0.005	-2.3848
Y_1Y_2	0.83	1	0.83	0.15	0.707	-0.3212
Y_1Y_3	0.76	1	0.76	0.14	0.718	-0.3087
Y ₂ Y ₃	11.88	1	11.88	2.17	0.175	-1.2187
Residual	49.29	9	5.447			
Lack of fit	45.74	5	9.148	10.30	0.021	Significant
Pure error	3.55	4	0.888			
Total	1190.41	19				
*DF- degree	e of freedom	ı				

Table 7.3: Analysis of variance (ANOVA) of Cu(II) removal prediction by using RSM model

The source contribution in Cu(II) removal percentage is evaluated based on adjusted sum of squares, and the plot is shown in Figure 7.2. The plot shows that sorbent dosage and initial concentration of wastewater effluent spiked with Cu(II) contribute more than 96% towards the removal percentage, while pH has a less than 5% contribution towards Cu(II) removal. A possible explanation for this result is the narrow pH range (5 - 8) compared to other studies which focused on a wider range (2 - 10), which would be an unrealistic condition in a real WWTP (Mohan *et al.*, 2015). The small value for pH (Y₁) in the first order coefficient

term (Table 7.3) illustrates the non-significant, negative effect of the variable on the adsorption process. In this study, Cu(II) removal is maximum around pH 7.0, would be expected to interact more strongly with negatively charged binding sites on the modified CNW adsorbents.

In order to improve the accuracy of the model, the insignificant terms were removed from the quadratic equation. However, there was no improvement to the accuracy of the model, even after eliminating the insignificant terms. The significance of lack of fit indicates that the RSM model is invalid for the present work when it has a value of less than 0.05 (Hamsaveni *et al.*, 2001; Zulkali *et al.*, 2006). Therefore, from the results, the lack of fit obtained is significant due to low probability (P=0.005) and a higher F-test value of 15.31, which is reinforced by the relatively low coefficient of determination ($R^2 = 0.9409$) for the overall model. Therefore, this result shows that the RSM model is unable to effectively predict the removal of spiked Cu(II) from wastewater effluent, and is similar to that obtained for the clean water matrix (Chapter 5).





A possible explanation for the poor fit of the quadratic equation in the RSM model is the variability of wastewater composition, which is likely to influence the efficiency of the adsorption process. Similar observations were found in studies on lead removal from industrial sludge leachate using red mud adsorbent,

and in the case of solid-phase extraction of gold from industrial wastewater using modified mesoporous silica. They showed that the variability in real wastewater samples cannot be efficiently predicted by the RSM model.

7.4.3 ARTIFICIAL NEURAL NETWORK (ANN)

An ANN-based model was also built for predicting the removal of Cu(II) from the effluent by oxidised CNWs adsorbents. In a similar way to RSM modelling, the data generated through CCD were used to determine the optimal architecture of the ANN model. As with RSM, three factors, namely pH, sorbent dosage, and initial concentration of wastewater effluent spiked with Cu(II) were used as input variables, and were used to predict the percentage Cu(II) removal from wastewater effluent through the adsorption process. A similar procedure was applied for the remediation of Cu(II) from the clean water matrix (Section 5.4.3). The total of 20 experiments were divided into three subsets comprising training (12 data points), validation (4 data points) and testing (4 data points). The trained network was used to estimate the response of 36 experimental points. R² between actual and estimated responses was determined as 0.999 (Appendix D).

Although ANN and RSM models in combination with experimental design have been increasingly applied in the area of water and wastewater treatment, the operation of a WWTP is often complicated because of the complexity of the wastewater matrix; this varies both temporally and spatially. A study by Ebrahimzadeh *et al.* (2012) showed good agreement between ANN predictions and experimental data, with a correlation coefficient of 0.9945, whereas this amount decreases to 0.8857 for an RSM model. However, there was no obvious reported improvement in the determination of metal ions from an industrial sample, even after applying the optimum conditions suggested by both models. As mentioned in Chapter 6, the concentrations of heavy metals in influent and effluent samples varied with time (Figure 6.5). Therefore, wastewater comprising of many contaminants including organic pollutants, natural organic matter, pathogenic microorganisms and various colloidal particulates may also reduce the prediction capability of the RSM and ANN models. Of these two models, ANN is found to be more efficient and more suitable for modelling such WWTP processes due to its accuracy and adequacy, and is promising in engineering applications (Witek-Krowiak *et al.*, 2014). ANN is more appropriate in the case of complex processes (i.e. WWTP processes) as the model allows for predictions of the output on the basis of input data without the need to define the relationship between them (Witek-Krowiak *et al.*, 2014).

A number of high quality reviews have appeared in the literature dealing with the application of ANN-based models in the field of water treatment (Antonopoulou *et al.*, 2012; Hamed *et al.*, 2004). For example, the study reported by Antonopoulou and his co-workers demonstrated that the ANN model gave better estimation capabilities than did the RSM model, throughout the range of variables, in the optimisation of photocatalytic degradation of phenolic compounds in treated wastewater. In addition, ANN has been successfully used as a tool for estimating the performance of filtration processes in wastewater treatment plants (Vijayabhanu and Radha, 2013).

An investigation by Nasr *et al.* (2012) focused on applying the ANN model to predict the performance of WWTPs in terms of chemical oxygen demand (COD), biological oxygen demand (BOD) and total suspended solid (TSS). The authors found that ANN models provided good quality predictions, with R^2 of greater than 0.9 between the observed and predicted output variables. In addition, ANN was employed to predict percentage separation of lead ions from real wastewater using electrodialysis. The results showed that ANN successfully traced the non-linear behaviour of separation percentage and current efficiency versus temperature, voltage, concentration and flow rate with a standard deviation of not more than 1% (Sadrzadeh *et al.*, 2009).

However, no work to date has included the variation in wastewater composition as one of the independent variables (ANN input) for remediation of Cu(II) from wastewater. For example, lead removal from industrial sludge leachate using red mud was studied by Geyikci *et al.* (2012). In this study, the effects of the sorbent dosage, contact time and pH on removal percentage were investigated using RSM and ANN modelling approaches without considering variation in wastewater compositions (Geyikci *et al.*, 2012). A study by Aber *et al.*, (2009) observed that the performance of electrocoagulation processes in removal of Cr(VI) from synthetic and real wastewater was successfully predicted by an ANN model. They found that an electrocoagulation process can be used for removal of total chromium and other pollutants from real wastewater (Aber *et al.*, 2009). Even though this study included effective parameters as independent variables, the process is often complicated due to the variety of contamination present in raw wastewater.

The issue with these studies is the assumption that the actual wastewater composition is the same for each experiment, or that its composition has no influence on the removal capability of that process. As mentioned in Chapter 6 (Understanding the wastewater matrix), it can be clearly shown that the composition of wastewater varies with time. Therefore, in order to study the effect of variation in wastewater composition through removal of spiked Cu(II) from wastewater effluent, benchmark experiments were conducted for each sampling trip, as mentioned in Section 3.5.3. The results obtained from the benchmark experiments (Table 7.4) showed that the percentage removals (74.41 – 78.76%) were different for each sampling trip, due to complexity of the wastewater matrix and its variability. Thus, it can be summarised that the variation of actual wastewater composition affects adsorption performance. The results obtained from benchmark experiments will therefore be included as the fourth independent variable in ANN modelling. The data for the comparison between the experimental values and those predicted by the ANN model, with and without the variability, are shown in Table 7.4. Moreover, a regression analysis between the experimental data and the values predicted by both ANN models is shown in Figure 7.3. The coefficient of determination ($R^2 = 0.9963$) for the ANN model with variability indicates good agreement between experimental and predicted results.



Figure 7.3: Comparison between the Cu(II) removal predicted by the artificial neural network (ANN) model and the experimentally determined removal; (WW) wastewater

			Inputs		Cu(II) removal (%)					
Run Number	V.	V.	V.	Benchmark	Exporimental	ANI	N	ANN (including W	W variability)	
	Y 1	¥ 2	Y 3	experiments (%)	Experimental –	Predicted	Residual	Predicted	Residual	
1	5.6	8.16	4.23	74.41	86.05	86.45	-0.40	86.08	-0.03	
2	5.6	8.16	1.78	74.41	71.26	71.76	-0.50	71.07	0.19	
3	6.5	5.25	3.00	77.39	80.65	81.61	-0.96	81.62	-0.97	
4	6.5	5.25	3.00	77.39	82.62	81.61	1.01	81.62	1.00	
5	5.6	2.34	4.23	78.76	80.32	80.13	0.19	80.36	-0.04	
7	7.4	8.16	4.23	74.41	88.2	88.35	-0.15	88.26	-0.06	
8	7.4	2.34	4.23	78.76	83.62	83.87	-0.25	83.09	0.53	
9	5.6	2.34	1.78	78.76	60.52	60.73	-0.21	60.58	-0.06	
10	6.5	5.25	3.00	74.41	80.43	81.61	-1.18	80.88	-0.45	
11	7.4	8.16	1.78	74.41	74.51	74.53	-0.02	74.53	-0.02	
12	7.4	2.34	1.78	78.76	65.19	65.58	-0.39	65.96	-0.77	
13	6.5	5.25	3.00	74.15	81.52	81.61	-0.09	81.17	0.35	
14	6.5	10	3.00	74.41	84.64	84.96	-0.32	84.52	0.12	
15	6.5	5.25	5.00	77.39	85.62	86.25	-0.63	85.79	-0.17	
16	5.0	5.25	3.00	78.76	83.36	83.24	0.12	83.35	0.01	
17	6.5	5.25	1.00	74.41	68.02	68.49	-0.47	68.01	0.01	
18	6.5	5.25	3.00	74.15	80.01	81.61	-1.60	81.17	-1.16	
19	6.5	5.25	3.00	74.15	81.09	81.61	-0.52	81.17	-0.08	
20	8.0	5.25	3.00	78.76	88.04	87.93	0.11	88.07	-0.03	
$Y_1 = pH; Y_2 = sc$	orbent d	losage; Y ₃	=initial o	concentration of wa	stewater effluent s	spiked with Cu	(II)			

Table 7.4: The experimentally obtained removal of Cu(II) compared to that predicted by ANN, and ANN including wastewater (WW) variability, as defined by benchmark experiments conducted for each sampling trip

For a better graphical interpretation of the process of Cu(II) adsorption from wastewater, three-dimensional response surface plots were generated. These plots are important in order to study the relationship between the response and the independent variables. The mutual interactive behaviour between two independent variables, while the third variable is held constant at its intermediate value (pH 6.5, 5.25 g/L, 3 mg/L), is shown in Figure 7.4.

As shown in Figure 7.4(A), maximum removal of Cu(II) is observed at a sorbent dosage of 8 g/L and pH 8. The percentage Cu(II) removal increased with the increase of pH due to the negative surface charge of oxidised CNW at alkaline pH values. At pH 8 and above, carboxyl group, –COOH, was changed into –COO⁻, hence the ion exchange between Cu(II) and potential functional groups increased (Reddy, 2012). The low adsorption that takes place in acidic solutions can be due to the competition between hydrogen and copper ions for active sites on the oxidised CNW surface (Rajemahadik *et al.*, 2013). However, pH did not significantly affect the adsorption removal, contrary to the findings of other studies, as the pH range studied in this work is narrow (pH 5.0-8.0). The effect of pH on the adsorption of Cr(VI) was investigated by Mohan *et al.* (2015) by varying pH from 2.0 to 10.0. Due to the wide pH range, pH was found to be one of the main parameters affecting the adsorption process.

Figure 7.4(B) shows the interaction effect of sorbent dosage and initial concentration of wastewater effluent spiked with Cu(II) on the removal of Cu(II) from the effluent, with pH held constant at pH 6.5. The Cu(II) removal increases with increasing sorbent dosage, which may be due to the increase in total surface active sites on the adsorbent surface. The amount of proton exchange between the adsorbent and the solutions increases with increasing sorbent dosage. Similar observations were found in studies on Cu(II) removal using alkali-modified spent tea, and in the case of Cr(VI) removal by modified silica. It was reported that higher removal at higher sorbent dosage may be due to the availability of more active functional groups in the greater mass of adsorbent (Cao *et al.*, 2014).



Figure 7.4: Surface plots (left) and corresponding contour plots (right) showing the effects of adsorption parameters on the spiked Cu (II) removal from wastewater effluent as predicted by the ANN model with initial concentration of wastewater effluent spiked with Cu(II) held constant 3 mg/L (A), pH held constant at 6.5 (B) and sorbent dosage held constant at 5.25 g/L (C).

Percentage removal of Cu(II) increases when pH and initial concentration of wastewater effluent spiked with Cu(II) increase, as shown in Figure 7.4(C). The increasing initial concentration of wastewater effluent spiked with Cu(II) provides the driving force to overcome the mass transfer resistance of Cu(II) ions between the aqueous and solid phases. Similar observations were reported in the literature, where the maximum Cu(II) removal by *Trametes versicolor* fungi was observed when initial Cu(II) concentration increased from 37 to 60 mg/L at pH of 5.51 (Sahan *et al.*, 2010).

7.5 MULTIPLE LINEAR REGRESSION (MLR)

Multiple linear regression (MLR) is a linear statistical analysis that is applicable for predicting the relationship between a dependent variable and two or more independent variables (Tiryaki and Aydın, 2014). In MLR, the dependent variable is known as the predictand, while the independent variables are the predictors (Arulsudar *et al.*, 2005). MLR models are used in the prediction of Cu(II) removal from the wastewater matrix, being represented by the relationship between the percentage removal and a set of predictor variables. MLR is based on least squares fit, where the model is adjusted such that the sum of squares of differences of actual and predicted values is minimised. The general MLR equation can be formulated by:

$$\ddot{\mathbf{Y}} = \beta_0 + \beta_1 X_1 + \dots + \beta_n X_n + \varepsilon \tag{7.1}$$

Where \ddot{Y} is the dependent variable, X_n the independent variables, β_n the predicted parameters, and ε is the error term.

Multiple linear regression (MLR) and artificial neural networks (ANNs) were used to predict the removal of Cu(II) from the effluent by oxidised CNW adsorbents. The data used in the MLR and ANN models were obtained from 20 CCD experiments. The MLR model (with and without accounting for the variability of the wastewater matrix) gives the mathematical expression of the output of the MLR analysis:

$$Z(\%) = 17.0722 + 1.7352Y_1 + 1.6348Y_2 + 5.7487Y_3 + 0.3218Y_4$$
(7.2)

$$Z(\%) = 42.4169 + 1.7352Y_1 + 1.4315Y_2 + 5.8445Y_3$$
(7.3)

Where Z is the dependent variable (percentage removal of copper from wastewater effluent) and Y_i the independent variables (pH, sorbent dosage, initial concentration of wastewater effluent spiked with Cu(II) and benchmark experiment, respectively). The data for the comparison between the experimental values and those predicted by the MLR model, with and without variability, are shown in Table 7.5.

			Inputs		Cu(II) removal (%)				
Run				Benchmark		ML	R	MLR (including W	W variability)
Number	Y_1	Y ₂	Y ₃	experiments (%)	Experimental	Predicted	Residual	Predicted	Residual
1	5.6	8.16	4.23	74.41	86.05	88.51	2.46	88.36	2.31
2	5.6	8.16	1.78	74.41	71.26	74.19	2.93	74.28	3.02
3	6.5	5.25	3.00	77.39	80.65	78.75	-1.91	79.08	-1.57
4	6.5	5.25	3.00	77.39	82.62	78.75	-3.88	79.08	-3.54
5	5.6	2.34	4.23	78.76	80.32	80.18	-0.14	80.25	-0.07
7	7.4	8.16	4.23	74.41	88.2	91.63	3.43	91.48	3.28
8	7.4	2.34	4.23	78.76	83.62	83.30	-0.32	83.37	-0.25
9	5.6	2.34	1.78	78.76	60.52	65.86	5.34	66.16	5.64
10	6.5	5.25	3.00	74.41	80.43	78.75	-1.69	78.12	-2.31
11	7.4	8.16	1.78	74.41	74.51	77.31	2.80	77.40	2.89
12	7.4	2.34	1.78	78.76	65.19	68.98	3.79	69.29	4.10
13	6.5	5.25	3.00	74.15	81.52	78.75	-2.78	78.04	-3.48
14	6.5	10	3.00	74.41	84.64	85.54	0.90	85.89	1.25
15	6.5	5.25	5.00	77.39	85.62	90.43	4.81	90.58	4.96
16	5.0	5.25	3.00	78.76	83.36	76.14	-7.22	76.92	-6.44
17	6.5	5.25	1.00	74.41	68.02	67.06	-0.96	66.63	-1.39
18	6.5	5.25	3.00	74.15	80.01	78.75	-1.27	78.04	-1.97
19	6.5	5.25	3.00	74.15	81.09	78.75	-2.35	78.04	-3.05
20	8.0	5.25	3.00	78.76	88.04	81.35	-6.69	82.13	-5.91

Table 7.5: The experimentally obtained removal of Cu(II) compared to that predicted by ANN, and ANN including wastewater (WW) variability, as defined by benchmark experiments conducted for each sampling trip

Y₁=pH; Y₂=sorbent dosage; Y₃=initial concentration of wastewater effluent spiked with Cu(II)

7.6 STATISTICAL COMPARISON AND PERFORMANCE OF MODELS FOR WASTEWATER EFFLUENT

The performance of the built MLR and ANN models (with and without accounting for the variability of the wastewater matrix), were compared and statistically measured by the coefficient of determination (R^2), absolute average deviation (AAD), and root mean squared error (RMSE). The relationships between the experimental (actual) values and calculated (predicted) values obtained using the ANN and MLR prediction models are shown in Figure 7.5. From the results, it can be seen that ANN models (with and without accounting for the variability of the wastewater matrix) show good agreement between predicted and actual results, compared to the MLR model.

The ANN including variability of wastewater matrix model fitted the experimental data with excellent accuracy and with a better prediction ($R^2 = 0.9963$) than the ANN not including variability ($R^2 = 0.9945$), and than the MLR model including ($R^2 = 0.7994$) and not including variability ($R^2 = 0.7961$). The AAD and RMSE for the ANN model including variability of wastewater matrix were calculated to be 0.30 % and 0.48 respectively, whilst those of the ANN model were 0.63 % and 0.69 respectively. In addition to the coefficients of determination for ANNs, the AAD and RMSE confirm that the ANN model including the variability of the wastewater matrix as the fourth independent variable is superior in predicting the removal of spiked Cu(II) from wastewater effluent. The calculations are fully described in Section 3.9.1. The comparison of the actual results, the ANN and MLR prediction models (accounting for the variability of the wastewater matrix) is presented graphically in Figures 7.6 and 7.7, for 20 CCD experiments.

A possible explanation for this result is the complexity of the wastewater in term of composition and its variability, which can affect the capability of the adsorbent to remediate Cu(II) from the wastewater matrix. A variety of organic and inorganic compounds can be found in the composition of wastewater, and its variability, both temporally and spatially (within a WWTP and in different wastewater streams), is likely to influence the efficiency and capability of oxidised CNW adsorbents. This explanation is reinforced when examining the results obtained in Chapter 6, where it is stated that in a real WWTP, the water matrix will be far more complex than clean water. Thus, with the results obtained from benchmark experiments, it is demonstrated that the complexity and variability in wastewater composition affects the adsorption performance.

Such interactions between percentage of Cu(II) removal and variations in wastewater sample (benchmark experiments) would be difficult to ascertain from the MLR model. Therefore, by supplying the ANN model with the data obtained from the benchmark experiments as the fourth independent variable, it is possible to improve the prediction ability of the ANN model. As can be seen from Table 7.4, it is found that the ANN model with the variability of the wastewater matrix is superior to the ANN model without the variability of the wastewater matrix when predicting Cu(II) removal from the wastewater matrix.

This finding has also been supported by other studies that have not included the variations in real wastewater composition as one of the independent variables (ANN input). For instance, although Geyikci and his co-workers reported that the results of ANN were found to be more reliable than RSM ($R^2 = 0.672$), a low coefficient of determination ($R^2 = 0.898$) from the ANN model indicated that the variation in industrial sludge leachate composition had an influence on the removal capability of the adsorbent. The major issue with this study is the assumption that real wastewater composition is the same for each sample, or has no impact on the removal capability of the adsorbent. Hence, it can be concluded that including wastewater variability as one of the input variables will lead to improvements in the predictability of the ANN model.



Figure 7.5: Comparison of the actual and predicted values by MLR (A and B) and ANN models (C and D) (including WW variability)



Figure 7.6: Comparison of the actual and predicted values by MLR and ANN models



Figure 7.7: Comparison of the actual and predicted values by MLR and ANN models (including WW variability)

7.6.1 MODEL VALIDATION USING UNSEEN EXPERIMENTS

For the validation and evaluation of the MLR and ANN models, an additional 10 unseen experiments were conducted, consisting of combinations of experimental factors that were not considered in the 20 CCD experiments. This was a necessary procedure, since no work has been reported to date in the choice of additional experiments to represent the whole system of remediating Cu(II) from a wastewater matrix. 3D scatter plots for the unseen experiments are displayed in Figure 7.7, and include a comparison of 3D scatter plots derived from data contained in two other studies which involved real wastewater samples. The 10 unseen experiments undertaken in this study, illustrated in Figure 7.8(A), were chosen to represent parameter space both inside and outside the system, to better understand and test the validity of the models.

However, as illustrated in Figure 7.8(B), Ebrahimzadeh *et al.*, (2012) designed 10 random experiments, using MATLAB programming, that did not represent the whole system, in order to study the ability of the RSM and ANN models to predict the solid-phase extraction of gold ions from industrial wastewater. Furthermore, in order to test the validity of RSM and ANN results, Geyikci *et al.* (2012) conducted 10 extra experiments that were all concentrated inside the system, as shown in Figure 7.8(C).

The actual and predicted values of the responses for the 10 unseen experiments, along with their residual values for the models, are summarised in Table 7.6. Moreover, the R^2 for both models ($R^2 = 0.9644$ for ANN including wastewater variability, $R^2 = 0.8991$ for ANN without), show that the ANN model predicts more accurately when variation in wastewater composition is included as the fourth independent variable. As shown in Table 7.7, the predictive abilities of the newly constructed MLR and ANN models, with and without wastewater variability, were statistically measured in terms of R^2 , AAD and RMSE. From the results, it is confirmed that the ANN model including wastewater variability predicts more accurately the remediation of spiked Cu(II) from wastewater effluent, in both the original 20 CCD and the 10 unseen experiments. This is because the ANN model allow for predicting the response (percentage of Cu(II) removal) without the need to justify the relationship between them, which is

particularly important in the case of real-world WWTP, where the water matrix will be more complex (Witek-Krowiak *et al.*, 2014).

This finding has also been supported by others who have used MLR and ANN in prediction studies. For instance, Tiryaki and his co-workers used ANN and MLR for predicting the compression strength of heat-treated woods. The results indicated that an ANN model provided better prediction results compared to an MLR model. Moreover, ANN models save time and decrease the experimental costs (Tiryaki and Aydın, 2014). In a pharmaceutical study, it was found that the ANN model provides predictions that are more accurate, and is more useful in the optimisation of pharmaceutical formulations compared to an MLR model. Another advantage of the ANN model is the flexibility to work with more input variables, which is helpful when involving large number of experiments; for MLR, a large number of input variables leads to a polynomial with many coefficients that involves tedious computation (Arulsudar *et al.*, 2005).



Figure 7.8: 3D scatter plots showing: (A) comparison of the CCD with unseen experiments within the systems for this work; (B) solid-phase extraction for determination of gold from industrial wastewater (Ebrahimzadeh *et al.*, 2012); (C) Lead adsorption from industrial sludge leachate (Geyikci *et al.*, 2012).

	Inputs							Cu(II) removal (%)								
Data index	Run	Y ₁	1 Y ₂	Y ₃	Benchmarks (%)	Actual	MI	MLR		MLR (WW variation)		ANN		ANN (WW variability)		
							Predicted	Residual	Predicted	Residual	Predicted	Residual	Predicted	Residual		
Je	1	7	1	3	71.96	74.49	73.53	-0.96	71.26	-3.23	72.98	1.51	77.61	-3.12		
of th n	2	6	1	2	71.96	65.81	65.95	0.14	63.77	-2.04	60.68	5.13	64.98	0.83		
itside c syster	3	5	8	3	77.39	76.88	80.08	3.20	77.52	0.64	84.14	-7.26	81.92	-5.04		
	4	7	10	4	77.39	78.72	92.26	13.54	90.01	11.29	87.54	-8.82	86.28	-7.56		
Õ	5	5	5	2	78.76	68.8	69.94	1.14	67.41	-1.39	71.25	-2.45	72.81	-4.01		
e	6	6	4	4	78.76	77.64	71.67	-4.72	70.04	-6.35	69.93	6.46	74.31	2.08		
f th m	7	6	5	2	74.41	72.50	83.99	6.09	81.91	4.01	84.92	-7.02	77.07	0.83		
e o stei	8	5.5	4	4.5	74.41	77.9	75.77	1.38	74.15	-0.24	80.19	-5.80	80.01	-5.62		
lsid sy	9	7.5	4	2.5	78.76	74.39	84.98	6.05	82.62	3.69	84.85	-5.92	86.01	-7.08		
Ir	10	7	9	3	74.15	78.93	79.82	1.36	77.39	-1.07	82.29	-3.83	82.04	-3.58		

Table 7.6: Validation data for 10 unseen experiments

Y₁=pH; Y₂=sorbent dosage; Y₃=initial concentration of wastewater effluent spiked with Cu(II); WW=wastewater

Table 7.7: Comparison of the predictive abilities of RSM and ANN model

		AAI	D (%)		RMSE					
Data index	MLR	MLR (WW variability)	ANN	ANN (WW variability)	MLR	MLR (WW variability)	ANN	ANN (WW variability)		
20 CCD	3.23	3.84	0.63	0.30	3.48	3.46	0.69	0.48		
10 unseen	4.98	4.38	7.17	5.23	5.43	4.64	5.83	4.57		
5 Inside	5.05	3.84	7.53	4.98	4.46	4.46	5.90	4.46		
5 Outside	4.90	4.93	6.81	5.49	6.26	5.37	5.75	4.67		
AAD= absolu	AAD= absolute average deviation; RMSE= root mean squared error; WW=wastewater									

7.7 PROCESS OPTIMISATION AND OPTIMUM PARAMETERS

As discussed in previous section, the aim of this work is to better understand the complexity and variability of the wastewater matrix, including evaluating the impact of the wastewater matrix on adsorbent performance to remediate copper pollutant from a real-world wastewater matrix. RSM and ANN models were developed in order to optimise the system and to create a predictive model to evaluate the Cu(II) removal performance by the oxidised CNW adsorbents. Additionally, a benchmark study was applied and incorporated into the ANN model, which was utilised to account for wastewater matrix variability and impact on adsorbent performance. In order for these results to be implemented in industrial applications, the optimum operating conditions will be applied in continuous flow experiments for the removal of spiked Cu(II) from wastewater effluent.

Process optimisation is a function of maximising the removal of Cu(II) from the wastewater matrix via a combination of different studied factors. There are two options for finding the optimal operating conditions for spiked Cu(II) removal from wastewater effluent: the graphical optimisation function and the desirability function.

Graphical representation of the model is the simplest approach for determining optimal operating conditions, particularly when the optimisation procedure involves two factors and one response. Vera Candioti *et al.* (2014) illustrated a suitable method for determining optimal operating conditions that involves one response via the graphical representation of the model, either by 3D space or contour graphs. In these graphs, the response is represented as a function of two factors. When more than two factors are studied, the other factors that are not plotted must be set at a constant value. Therefore, only a limited part of the experimental domain is shown, which leads to the difficult establishment of optimal operating conditions (Vera Candioti *et al.*, 2014).

Desirability is an objective function that ranges from zero outside of the limits, to one at the goal. In 1980, Derringer and Suich (1980) developed the desirability function, which has been widely used in industry to find optimal operating conditions. The main aim of this function is not only to find a good set of operating conditions that meet all the relevant criteria, but also to give the best desirability value. Moreover, the desirability function has been successfully applied in several studies to determine the desired parameters for maximum heavy metals removal from the water matrix (Amini *et al.*, 2008; Anupam *et al.*, 2011; Rao *et al.*, 2012; Zolgharnein *et al.*, 2013a). Therefore, the appropriate way to find the optimal operating conditions for this study is by applying the desirability function.

In this study, the optimal operating conditions for the spiked Cu(II) removal from the wastewater effluent were determined using the desirability functions available in MINITAB 16 statistical software. The optimum operating conditions suggested by the design of experiment (DoE) model for the three variables, i.e., pH, sorbent dosage and initial Cu(II) concentration studied in this experiment, were pH 8.0, 6.45 g/L and 4.72 mg/L, respectively. Benchmark experiments were performed to account for wastewater matrix variability and impact on adsorbent performance, prior to determining optimal operating conditions. As the value of desirability obtained for Cu(II) removal was 1, it has been proven that the estimated function may represent the experimental model and the desired conditions (Anupam *et al.*, 2011).

In order to confirm the model's adequacy, batch experiments were conducted in triplicate at optimum conditions to obtain maximum spiked Cu(II) removal experimentally. The procedure is fully described in Section 3.5.3. The predicted and experimental optimum conditions of the process variables for the maximum percentage spiked Cu(II) removal from the wastewater effluent is shown in Table 7.8. The removal percentages obtained were lower than predicted removal efficiency in optimal conditions. This was because the composition and concentration of substances in wastewater varies significantly over time (Henze *et al.*, 2008).

	Optima	al operating condit	tions		Exper	imental operat	ing conditions	Cu(II) removal (%)			
Run	рН	Sorbent dosage (g/L)	Initial concentrat (mg/L)	Cu(II)	рН	Sorbent dosage (g/L)	Initial Cu(II) concentration (mg/L)	Benchmark experiments (%)	Actual	Predicted	% Error
1	8.0	6.45	4.72		7.8	6.50	4.61	77.27	91.36	92.11	-0.75
2	8.0	6.45	4.72		7.9	6.50	4.61	80.11	91.36	92.11	-0.75
3	8.0	6.45	4.72		8.0	6.15	4.31	78.75	90.54	92.11	-1.57

Table 7.8: Optimised operating conditions for spiked Cu(II) removal from wastewater effluent.

7.7.1 PERFORMANCE OF CONTINUOUS FLOW EXPERIMENT UNDER OPTIMAL OPERATING CONDITIONS

Continuous flow experiments were carried out using oxidised adsorbent for the removal of spiked Cu(II) from wastewater effluent. For the continuous flow experiments, each experiment was conducted under optimal conditions, which was determined from the desirability functions (as discussed in Section 7.6). Continuous flow experiments were performed in a solid phase extraction (SPE) vacuum manifold, with the adsorbent continuously in contact with wastewater effluent spiked with Cu(II). The procedure is discussed in detail in Section 3.10.

Continuous flow experiments were operated at two different pressures (P), 10 and 15 mmHg in a column filled with oxidised CNW adsorbents. The final Cu(II) concentration in the effluent was plotted against the volume of treated effluent, the profile for which is shown in Figure 7.9. As the pressure increased, the final concentration of Cu(II) in the effluent also increased, thereby decreasing removal efficiency (Table 7.9).



Figure 7.9: Effect of pressure on Cu(II) removal from wastewater effluent.

The results suggest that at a high pressure, the adsorbent in the cartridge may compact as a result of the pressure, thereby reducing the availability of sorption sites for adsorption. This result corresponds well with those of Maheshwari and Gupta (2016), who suggested that with an increase in pressure, there is a decrease

in contact time between metal ions and adsorbent, which may lead to a reduction in the overall percentage removal of heavy metals.

Run	Volume (ml)	Final Cu(II) (m	concentration ng/L)	Removal efficiency (%)		
		10 mmHg	15 mmHg	10 mmHg	15 mmHg	
1	0	C _i =4.98	C _i =4.98	0	0	
2	5	0.43	0.40	91.48	91.99	
3	10	0.43	0.42	91.48	91.53	
4	15	0.44	0.54	91.16	89.21	
5	20	0.49	0.62	90.14	87.59	
6	25	0.56	0.75	88.75	85.05	
7	30	0.60	0.80	88.06	83.89	
8	35	0.65	0.84	87.04	83.20	
9	40	0.66	0.84	86.76	83.20	
10	45	0.69	0.85	86.02	82.97	
11	50	0.70	0.85	86.02	82.97	
$C_i = Ini$	itial Cu(II) conce	ntration (mg/I	L)			

Table 7.9: Effect of pressure on spiked Cu(II) removal efficiency from wastewater effluent by oxidised CNW adsorbents.

As noted in Section 2.5.2, chemically modified adsorbents improve removal efficiency and adsorption capacity Although oxidised CNW adsorbents are able to remove approximately 90% of spiked Cu(II) from wastewater effluent, without reinforcement and granulation, it is not a suitable structure for use in continuous flow column operations. This is because the physical structure of oxidised CNWs is extremely soft and in this study, easily caused column clogging when wet (Volesky, 1995). These observations are in agreement with those of Mason (2007), who stated that natural biomass, including cotton wool, is extremely soft and not suitable for column operation (Mason, 2007). Therefore, the mechanical properties of adsorbents must be improved in order to provide a more stable structure, where the adsorbent can be used directly in a standard operation process.

CHAPTER 8: CONCLUSIONS AND RECOMMENDATION FOR FUTURE RESEARCH

8.1 CONCLUSIONS

The main aim of this study is to predict the capability of oxidised cellulose nanowhisker (CNW) adsorbents for the remediation of copper from water matrices under realistic conditions using RSM and ANN modelling. This thesis is divided into five parts. The first part focused on the modification of the adsorbent, and the success of modification (surface area, dimensions, surface charge, and stability) was confirmed using several analytical methods. The second part of the study observed the capability of oxidised CNW adsorbents in the remediation of copper from a clean water matrix using response surface methodology (RSM) and artificial neural network (ANN) models. In the third part, the complexity of wastewater in term of composition and composition variability, both temporally and spatially, was presented. This gave a better understanding of the selection of variables and their ranges in the experimental design. Benchmark study was reported to inform how the variable and complex wastewater matrix affects the adsorption process. The fourth part of the study examined the potential of oxidised CNW adsorbents to remediate Cu(II) from the wastewater matrix. The data obtained from benchmark studies was used as an additional independent variable for ANN modelling to account for matrix composition. Finally, the application of oxidised CNW adsorbents was tested in a continuous flow experiment. This chapter will summarise the most important results obtained in this thesis.

8.1.1 CHARACTERISATION OF CNW ADSORBENTS

CNWs have been successfully prepared via sulphuric acid hydrolysis under controlled conditions. TEMPO-mediated oxidation then introduced stable negative electrostatic charges on the surface of the CNWs. The success of this modification was confirmed by functionalities, surface charge, morphology (surface area and porosity), and stability using FTIR, zeta potential, Brunauer-Emmett and Teller (BET) analysis, scanning electron microscopy (SEM),

transmission electron microscopy (TEM) and conductometric titration. FTIR analysis demonstrated that the CNWs were successfully modified, with the appearance of a C=O stretching band at 1740 cm⁻¹, indicating the carboxyl content. The zeta potential of this oxidised CNW in water was -74.4 mV, indicating the introduction of carboxylate groups to the surface of the CNWs at high density. The BET method showed that the surface areas of unmodified CNW and oxidised CNW adsorbents were 15.72 m²/g and 43.11 m²/g respectively, within the reported range of surface area for CNWs. The three times higher surface area for oxidised, i.e. carboxymethyl, cellulose however indicates a slightly higher colloidal stability during the freezing process that precedes freeze drying, resulting in a less aggregated state after freeze drying. Since the resolution of the SEM was considered insufficient for detailed information, TEM was used to identify individual whiskers, which enabled the determination of their size and shape. The whiskers were measured to be 134.4 ± 51.2 nm and 9.0 ± 2.3 nm in length and width respectively. SEM and TEM images confirmed that the oxidised CNW maintain their initial morphological integrity after the oxidation process. Conductometric titration of sample suspensions showed a carboxylate content of 54 and 410 mmol/kg for the unmodified CNW and oxidised CNW, respectively. For the same amount of sorbent dosage under similar conditions, oxidised CNW adsorbents were able to remove 66.75 % of Cu(II), while the unmodified CNW adsorbents removed only 3.64 % of Cu(II) from the water matrix. The adsorption capacity of these adsorbents was 14.65 mg/g and 0.59 mg/g respectively. These results indicated that there are more carboxylate groups per gram due to high surface area. From a manufacturing and application perspective, the stability of the oxidised CNW adsorbents was tested under dry and wet conditions. This demonstrated for the first time that the functional groups were permanent and did not degraded under tested conditions.

8.1.2 PREDICTING THE CAPABILITY OF OXIDISED CNW ADSORBENTS FOR THE REMEDIATION OF COPPER FROM CLEAN WATER USING RSM AND ANN MODELS

Oxidised CNW adsorbents were effective in removing Cu(II) from the water matrix, offering the potential of an abundant, inexpensive, and available adsorbent; a suitable alternative to expensive adsorbents. RSM and ANN models were used to understand the optimum conditions for the removal of Cu(II) using oxidised CNW adsorbents. To further test the predictive capability of both models, an additional 14 unseen experiments not used in the development of the RSM and ANN models. These unseen experiments were chosen to best represent the system of conditions both inside and outside the system, the first time this has been undertaken for evaluating cellulose-based adsorbents. The performance of both models, which were statistically evaluated, indicated that ANN has superior predictability than RSM. From isotherm and kinetics studies, it was observed that the adsorption process followed the Langmuir adsorption isotherm and pseudo-second-order kinetics.

8.1.3 UNDERSTANDING THE WASTEWATER MATRIX AND CHARACTERISATION

The data presented for the wastewater matrix showed that wastewater composition is complex and varies in its composition, both temporally and spatially. In addition, the physical and chemical (physicochemical) parameters discussed in this chapter also vary with time, which make the wastewater matrix far more complex than a clean water matrix for evaluating the performance of the adsorption process. Moreover, the benchmark study developed in this study also demonstrates that the variable and complex wastewater matrix composition affects the capability of oxidised CNW adsorbents to remediate Cu(II) from the matrix. Therefore, it is important to take these issues into account when predicting the capability of oxidised CNW adsorbents for remediation of Cu(II) from wastewater effluent. Few studies utilise the wastewater matrix when evaluating adsorbent performance and to date, none consider and account for the temporal and/or spatial variability of that matrix.

8.1.4 PREDICTING THE CAPABILITY OF OXIDISED CNW ADSORBENTS FOR REMEDIATION OF SPIKED COPPER FROM WASTEWATER EFFLUENT USING RSM AND ANN MODELS

Oxidised CNW adsorbents are capable of removing spiked Cu(II) ions from wastewater effluent. The RSM and ANN models were employed to optimise the system and to create a good predictive model. No work in the reviewed literature included matrix complexity and the variability of the wastewater as one of the independent variables in ANN modelling. Evidently this novel approach and the outcomes were employed in this study for the first time, as most studies do not consider matrix variability and its impact when evaluating the efficiency of an adsorbent. To test the predictive capability of these models, an additional 10 unseen experiments, not used in developing both models, were chosen to represent the system of conditions both inside and outside the system. This study (20 CCD and 10 unseen experiments) found that the ANN model accounting for wastewater variability was superior to the RSM model and to the ANN model not including wastewater variability, in terms of the coefficient of determination (\mathbb{R}^2) , the absolute average deviation (AAD) and root mean squared error (RMSE) when predicting the efficiency of Cu(II) removal from the wastewater matrix. The optimum adsorption conditions were determined as an initial pH value of 8.0, a sorbent dosage of 6.45 g/L and initial Cu(II) concentration of 4.72 mg/L. At optimum adsorption conditions, the percentage removal of spiked Cu(II) from the wastewater effluent was found to be 92.11%. Although oxidised CNW adsorbents were able to remove approximately 90% of spiked Cu(II) from wastewater effluent, the physical structure of oxidised CNW adsorbents is not suitable for use in continuous flow column operations.

8.2 **RECOMMENDATION FOR FUTURE RESEARCH**

8.2.1 MANUFACTURING OXIDISED CNW ADSORBENTS

This work recommends an equal focus not only on evaluating the capability of an adsorbent (e.g. oxidised CNW) to remove Cu, but also on how the adsorbent can be manufactured in a cost effective manner at scale to enable application in WWTPs. It is important to appreciate the matrix and the fact that it is variable; and to appreciate the need to evaluate a technology in response to that matrix variability, as the wastewater matrix has influence on the technology's performance. This thesis determines that applying the effects of matrix complexity and the variability of the wastewater (benchmark study) into the ANN model (which is not yet done in other studies) can predict adsorbent's performance in WWTPs. Therefore, these findings may be used in future to evaluate not only adsorbents but also treatment technologies; particularly those technologies that need a water quality parameter (benchmark study) to evaluate treatment capability in the face of the variable wastewater matrix. Further, the results obtained from the continuous flow experiment (Chapter 8) confirm that the physical structure of oxidised CNWs renders them unsuitable for use in a column operation. Future research should pay attention to prepare the adsorbent for use for instance by pelleting, or by improving the physical structure of oxidised CNWs. Cellulose aerogels are known to be highly porous materials that exhibit excellent adsorption properties as well as remarkable reusability. Therefore, oxidised CNW aerogels could be used effectively in a continuous flow experiment for the removal and recovery of Cu(II) from water matrices.

8.2.2 SUITABILITY OF ADSORBENTS FOR CONTINUOUS FLOW COLUMN APPLICATION

The physical and chemical structure of adsorbent plays an important part in the continuous flow column application. Reports in the literature have shown that attempts have been made to improve the mechanical strength of the adsorbents by chemical crosslinking, immobilisation and granulation processes. Therefore, for further application in wastewater treatment, a more detailed study of the

mechanical properties of CNW adsorbents would be necessary in order to improve the strength and stability of the adsorbent.

8.2.3 REGENERATION OF ADSORBENT

Raw material conservation and problems of discharge disposal lead inherently to the issues of regeneration and life cycle management. In most adsorption processes, regeneration of the adsorbent is necessary to ensure it is cost effective and environmentally friendly. Although one-time use may be more economical for small systems, regeneration techniques for CNWs may be necessary to produce an economically viable process for large systems. Therefore, further study on the regeneration process should be performed to test the stability of CNWs after multiple batch sorption-desorption cycles.

8.2.4 RECOVERY OF HEAVY METALS FROM THE ADSORBENT

In the application of oxidised CNW adsorbents for the recovery of Cu(II) from the water matrix, an assessment is necessary to making the adsorption process more economical. The study and application of adsorbents for WWTP has focussed on removing pollutant from the water matrix to meet ever increasingly stringent legislation aimed at reducing the pollutant load of treated water entering the receiving rivers and water bodies. However, a shift in attitude has meant that pollutants such as copper when separated from the water matrix are also valuable products. There is an appreciation that adsorption-based treatment technologies therefore offer recovery of copper for reuse, subject to downstream clean up. Hence adsorbents for metal removal form the water matrix followed by recovery and reuse are an attractive approach for future study. A limited number of reported studies have focused on the recovery of heavy metals from saturated adsorbents and desorbing agents. Future studies would focus not only on the recovery process, but also on the choice of suitable desorbing/regenerating agent that did not adversely affect the structure of the CNW.

8.2.5 PREPARATION OF DUAL AND MULTI-FUNCTIONAL GROUPS ON THE SURFACE OF CNWS

In this study, an oxidised CNW adsorbent was successfully modified and used as an adsorbent for Cu(II) removal. The experiments showed that this oxidised CNW adsorbent was able to remediate Cu(II) from clean water and wastewater effluent. This study shows that the adsorption of heavy metals from the aqueous phase onto a solid adsorbent depends on the affinity between the metal and functional group. This should be further investigated in order to ensure the best functional group for heavy metal pollutants. Therefore, dual or multi-functional groups could be attached to the adsorbent surface to enhance the selectivity of CNW adsorbents towards target metal ions or other constituents in wastewater. As a note, the production cost for developing this multi-functional material in a one- or multiple-step process should be taken into account.

REFERENCES

Abdel-Sabour, M.F., Rabie, F.H., Mostafa, T., Hassan, S.A., 2001. Impact of industrial wastewater disposal on surface water bodies in Mostord area, North Greater Cairo. *Journal of Environmental Sciences-China* 13, 485-490.

Aber, S., Amani-Ghadim, A.R., Mirzajani, V., 2009. Removal of Cr(VI) from polluted solutions by electrocoagulation: modeling of experimental results using artificial neural network. *Journal of Hazardous materials* 171, 484-490.

Act, 1991. Water Industry Act. The National Archives. 1991 Chapter 56, 25 July 1991

Aghav, R.M., Kumar, S., Mukherjee, S.N., 2011. Artificial neural network modeling in competitive adsorption of phenol and resorcinol from water environment using some carbonaceous adsorbents. *Journal of Hazardous materials* 188, 67-77.

Agilent Technologies, I., 2010. The Principles of ICP-MS. Agilent Technologies, Inc.

Agilent Technologies, I., 2015. Flame Atomic Absorption Spectrometry: Analytical methods Agilent Technologies Inc.

Aksu, Z., Isoglu, I.A., 2005. Removal of copper(II) ions from aqueous solution by biosorption onto agricultural waste sugar beet pulp. *Process Biochemistry* 40, 3031-3044.

Aleboyeh, A., Kasiri, M.B., Olya, M.E., Aleboyeh, H., 2008. Prediction of azo dye decolorization by UV/H2O2 using artificial neural networks. *Dyes and Pigments* 77, 288-294.

Ali, I., Gupta, V.K., 2006. Advances in water treatment by adsorption technology. *Nature Protocols* 1, 2661-2667.

Alila, S., Boufi, S., 2009. Removal of organic pollutants from water by modified cellulose fibres. *Industrial Crops and Products* 30, 93-104.

Alonso, E., Santos, A., Callejón, M., Jiménez, J.C., 2004. Speciation as a screening tool for the determination of heavy metal surface water pollution in the Guadiamar river basin. *Chemosphere* 56, 561-570.

Alves, G., Leandro, V., Gil, L.F., 2009. Adsorption of Cu(II), Cd(II), and Pb(II) from aqueous single metal solutions by succinylated mercerized cellulose modified with triethylenetetramine. *Carbohydrate Polymers* 77, 142-149.

Amarasinghe, B.M.W.P.K., Williams, R.A., 2007. Tea waste as a low cost adsorbent for the removal of Cu and Pb from wastewater. *Chemical Engineering Journal* 132, 299-309.

Amini, M., Younesi, H., Bahramifar, N., Lorestani, A.A.Z., Ghorbani, F., Daneshi, A., Sharifzadeh, M., 2008. Application of response surface methodology for optimization of lead biosorption in an aqueous solution by Aspergillus niger. *Journal of Hazardous materials* 154, 694-702.

Amuda, O.S., Adelowo, F.E., Ologunde, M.O., 2009. Kinetics and equilibrium studies of adsorption of chromium(VI) ion from industrial wastewater using

Chrysophyllum albidum (Sapotaceae) seed shells. *Colloids and Surfaces B: Biointerfaces* 68, 184-192.

An, H.K., Park, B.Y., Kim, D.S., 2001. Crab shell for the removal of heavy metals from aqueous solution. *Water Research* 35, 3551-3556.

Anirudhan, T.S., Jalajamony, S., Sreekumari, S.S., 2012. Adsorption of heavy metal ions from aqueous solutions by amine and carboxylate functionalised bentonites. *Applied Clay Science* 65-66, 67-71.

Ansari, M., Raisi, A., Aroujalian, A., Dabir, B., Irani, M., 2015. Synthesis of nano-NaX zeolite by microwave heating method for removal of lead, copper, and cobalt ions from aqueous solution. *Journal of Environmental Engineering* 141.

Antonopoulou, M., Papadopoulos, V., Konstantinou, I., 2012. Photocatalytic oxidation of treated municipal wastewaters for the removal of phenolic compounds: optimization and modeling using response surface methodology (RSM) and artificial neural networks (ANNs). *Journal of Chemical Technology & Biotechnology* 87, 1385-1395.

Anupam, K., Dutta, S., Bhattacharjee, C., Datta, S., 2011. Adsorptive removal of chromium (VI) from aqueous solution over powdered activated carbon: optimisation through response surface methodology. *Chemical Engineering Journal* 173, 135-143.

APHA, 1998. Standard Methods for the Examination of Water and Wastewater, 20th ed. American Public Health Association, American Public Health Association: Washington, D.C.

Araki, J., Wada, M., Kuga, S., 2001. Steric stabilization of a cellulose microcrystal suspension by poly(ethylene glycol) grafting. *Langmuir* 17, 21-27.

Argun, M.E., Dursun, S., Karatas, M., 2009. Removal of Cd(II), Pb(II), Cu(II) and Ni(II) from water using modified pine bark. *Desalination* 249, 519-527.

Arulsudar, N., Subramanian, N., Muthy, R.S.R., 2005. Comparison of Artificial Neural Network and Multiple Linear Regression in the Optimization of Formulation Parameters of Leuprolide Acetate Loaded Liposomes. *Journal of Pharmacy and Pharmaceutical Sciences* 8, 243-258.

Ashraf, M.A., Mahmood, K., Wajid, A., Maah, M.J., Yusoff, I., 2011. Study of low cost biosorbent for biosorption of heavy metals, in: Wu, K.J. (Ed.), Food Engineering and Biotechnology, pp. 60-68.

Atalla, R., Vanderhart, D., 1984. Native cellulose: A composite of two distinct crystalline forms. *Science* 223, 283-285.

Azila, Y.Y., Mashitah, M.D., Bhatia, S., 2008. Process optimization studies of lead (Pb(II)) biosorption onto immobilized cells of Pycnoporus sanguineus using response surface methodology. *Bioresource Technology* 99, 8549-8552.

Balaram, V., 2005. Recent developments in analytical techniques for characterization of ultra pure materials - An overview. *Bulletin of Materials Science* 28, 345-348.

Baltpurvins, K.A., Burns, R.C., Lawrance, G.A., Stuart, A.D., 1996. Use of the solubility domain approach for the modeling of the hydroxide precipitation of
heavy metals from wastewater. *Environmental Science & Technology* 30, 1493-1499.

Bansode, R.R., Losso, J.N., Marshall, W.E., Rao, R.M., Portier, R.J., 2004. Pecan shell-based granular activated carbon for treatment of chemical oxygen demand (COD) in municipal wastewater. *Bioresource Technology* 94, 129-135.

Barakat, M.A., 2011. New trends in removing heavy metals from industrial wastewater. *Arabian Journal of Chemistry* 4, 361-377.

Barker, T.B., Milivojevich, A., 2016. Quality by Experimental Design, Fourth Edition. CRC Press.

Barrett, E.P., Joyner, L.G., Halenda, P.P., 1951. The determination of pore volume and area distributions in porous substances. I. Computations from Nitrogen Isotherms. *Journal of the American Chemical Society* 73, 373-380.

Bashir, M.J., Amr, S.S.A., Aziz, S.Q., Aun, N.C., Sethupathi, S., 2015. Wastewater treatment processes optimization using response surface methodology (RSM) compared with conventional methods: review and comparative study. *Middle-East J Sci Res* 23, 244-252.

Bashir, M.J.K., Aziz, H.A., Yusoff, M.S., Adlan, M.N., 2010a. Application of response surface methodology (RSM) for optimization of ammoniacal nitrogen removal from semi-aerobic landfill leachate using ion exchange resin. *Desalination* 254, 154-161.

Bashir, M.J.K., Aziz, H.A., Yusoff, M.S., Aziz, S.Q., Mohajeri, S., 2010b. Stabilized sanitary landfill leachate treatment using anionic resin: Treatment optimization by response surface methodology. *Journal of Hazardous materials* 182, 115-122.

Beaty, R.D., Kerber, J.D., 1978. Concepts, instrumentation and techniques in atomic absorption spectrophotometry. Perkin-Elmer USA.

Beck-Candanedo, S., Roman, M., Gray, D., 2005. Effect of reaction conditions on the properties and behavior of wood cellulose nanocrystal suspensions. *Biomacromolecules* 6, 1048-1054.

Bezerra, M.A., Santelli, R.E., Oliveira, E.P., Villar, L.S., Escaleira, L.A., 2008. Response surface methodology (RSM) as a tool for optimization in analytical chemistry. *Talanta* 76, 965-977.

Bhattacharya, D., Germinario, L.T., Winter, W.T., 2008. Isolation, preparation and characterization of cellulose microfibers obtained from bagasse. *Carbohydrate Polymers* 73, 371-377.

Bhuiyan, M.A.H., Parvez, L., Islam, M.A., Dampare, S.B., Suzuki, S., 2010. Heavy metal pollution of coal mine-affected agricultural soils in the northern part of Bangladesh. *Journal of Hazardous materials* 173, 384-392.

Bingol, D., Hercan, M., Elevli, S., Kilic, E., 2012. Comparison of the results of response surface methodology and artificial neural network for the biosorption of lead using black cumin. *Bioresource Technology* 112, 111-115.

Bini, C., Bech, J., 2014. PHEs, Environment and Human Health: Potentially harmful elements in the environment and the impact on human health. Springer Netherlands.

Bondeson, D., Mathew, A., Oksman, K., 2006. Optimization of the isolation of nanocrystals from microcrystalline cellulose by acid hydrolysis. *Cellulose* 13, 171-180.

Box, G.E.P., Wilson, K.B., 1951. On the experimental attainment of optimum conditions. *Journal of the Royal Statistical Society. Series B (Methodological)* 13, 1-45.

Brown, R.M., 2004. Cellulose structure and biosynthesis: What is in store for the 21st century? *Journal of Polymer Science Part a-Polymer Chemistry* 42, 487-495.

Brunauer, S., Emmett, P.H., Teller, E., 1938. Adsorption of gases in multimolecular layers. *Journal of the American Chemical Society* 60, 309-319.

Bulut, Y., Tez, Z., 2007. Removal of heavy metals from aqueous solution by sawdust adsorption. *Journal of Environmental Sciences-China* 19, 160-166.

Burton, F., Tchobanoglous, G., Tsuchihashi, R., Stensel, H.D., Metcalf, Eddy, I., 2013. Wastewater Engineering: Treatment and Resource Recovery. McGraw-Hill Education.

Calero, M., Blazquez, G., Martin-Lara, M.A., 2011. Kinetic modeling of the biosorption of Lead(II) from aqueous solutions by solid waste resulting from the olive oil production. *Journal of Chemical and Engineering Data* 56, 3053-3060.

Cao, J., Wu, Y., Jin, Y., Yilihan, P., Huang, W., 2014. Response surface methodology approach for optimization of the removal of chromium(VI) by NH2-MCM-41. *Journal of the Taiwan Institute of Chemical Engineers* 45, 860-868.

Cao, X., Ding, B., Yu, J., Al-Deyab, S.S., 2012. Cellulose nanowhiskers extracted from TEMPO-oxidized jute fibers. *Carbohydrate Polymers* 90, 1075-1080.

Carlson, M.A., Lohse, K.A., McIntosh, J.C., McLain, J.E.T., 2011. Impacts of urbanization on groundwater quality and recharge in a semi-arid alluvial basin. *Journal of Hydrology* 409, 196-211.

Carroll, C., 2003. Engineering Statistics Handbook, in: Paul, T., Ledi, T. (Eds.), Central Composite Design

Cavas, L., Karabay, Z., Alyuruk, H., Doğan, H., Demir, G.K., 2011. Thomas and artificial neural network models for the fixed-bed adsorption of methylene blue by a beach waste Posidonia oceanica (L.) dead leaves. *Chemical Engineering Journal* 171, 557-562.

CEC, 1976. Council directive on pollution caused by certain dangerous substances discharged into the aquatic environment of the Community. Council of the European Communities. *Official Journal*, L 129/23, 4 May 1976

CEC, 1986. The Sewage Sludge Directive (86/278/EEC). Council of the European Communities (CEC). *Official Journal*, L 181/6, 4 July 1986

CEC, 1991. Directive concerning urban wastewater treatment (91/271/EEC). Council of the European Communities. *Official Journal*, L135, 30 May 1991

CEC, 2000. Directive 2000/60/EC of the European Parliament and of the Council on establishing a framework for Community action in the field of water policy.

Council of the European Communities. Official journal, L 327/1, 22 December 2000

CEC, 2006. Directive of the European Parliament and of the Council on the quality of fresh waters needing protection or improvement in order to support fish life (2006/44/EC). Council of the European Communities (CEC). *Official Journal*, L 264/20, 25 September 2006

CEC, 2008. Directive of the European Parliament and of the Council on environmental quality standards in the field of water policy, amending and subsequently repealing Council Directives 82/176/EEC, 83/513/EEC, 84/156/EEC, 84/491/EEC, 86/280/EEC and amending Directive 2000/60/EC of the European Parliament and of the Council. Council of the European Communities. *Official Journal*, L 348/84, 24 December 2008

Chand, P., Bafana, A., Pakade, Y.B., 2015. Xanthate modified apple pomace as an adsorbent for removal of Cd (II), Ni (II) and Pb (II), and its application to real industrial wastewater. *International Biodeterioration & Biodegradation* 97, 60-66.

Chapman, D., 1997. Water Quality Assessment. A Guide to the use of Biota, Sediments and water in Environmental Monitoring, 2 ed. Taylor & Francis, E& FN Spon, London.

Chen, S., Shen, W., Yu, F., Hu, W., Wang, H., 2010. Preparation of amidoximated bacterial cellulose and its adsorption mechanism for Cu2+ and Pb2+. *Journal of Applied Polymer Science* 117, 8-15.

Chen, S., Zou, Y., Yan, Z., Shen, W., Shi, S., Zhang, X., Wang, H., 2009. Carboxymethylated-bacterial cellulose for copper and lead ion removal. *Journal of Hazardous materials* 161, 1355-1359.

Chen, W.-f., Zhang, J., Zhang, X., Wang, W., Li, Y., 2016. Investigation of heavy metal (Cu, Pb, Cd, and Cr) stabilization in river sediment by nano-zero-valent iron/activated carbon composite. *Environmental Science and Pollution Research* 23, 1460-1470.

Cheng, D., Wen, Y., An, X., Zhu, X., Ni, Y., 2016. TEMPO-oxidized cellulose nanofibers (TOCNs) as a green reinforcement for waterborne polyurethane coating (WPU) on wood. *Carbohydrate Polymers*.

Cheng, X., Shi, H., Adams, C.D., Ma, Y., 2010. Assessment of metal contaminations leaching out from recycling plastic bottles upon treatments. *Environmental Science and Pollution Research* 17, 1323-1330.

Chernen'kova, T.V., Bochkarev, Y.N., Friedrich, M., Boettger, T., 2014. The impact of natural and anthropogenic factors on radial tree growth on the northern Kola Peninsula. *Contemporary Problems of Ecology* 7, 759-769.

Christophe, K., Guedenon, P., Nelly, K., Edorh, P.A., Adechina, R., 2011. Evaluation of heavy metals pollution of Nokoue Lake. *African Journal of Environmental Science and Technology* 5, 255-261.

Ciolacu, D., Gorgieva, S., Tampu, D., Kokol, V., 2011. Enzymatic hydrolysis of different allomorphic forms of microcrystalline cellulose. *Cellulose* 18, 1527-1541.

Cojocaru, C., Diaconu, M., Cretescu, I., Savic, J., Vasic, V., 2009. Biosorption of copper(II) ions from aqua solutions using dried yeast biomass. *Colloids and Surfaces a-Physicochemical and Engineering Aspects* 335, 181-188.

da Silva Perez, D., Montanari, S., Vignon, M.R., 2003. TEMPO-mediated oxidation of cellulose III. *Biomacromolecules* 4, 1417-1425.

Da'na, E., Sayari, A., 2012. Adsorption of heavy metals on amine-functionalized SBA-15 prepared by co-condensation: Applications to real water samples. *Desalination* 285, 62-67.

Das, N., Vimala, R., Karthika, P., 2008. Biosorption of heavy metals - An overview. *Indian Journal of Biotechnology* 7, 159-169.

Derringer, G., Suich, R., 1980. Simultaneous optimization of several response variables. *Journal of Quality Technology* 12, 214-219.

Divers, M.T., Elliott, E.M., Bain, D.J., 2014. Quantification of nitrate sources to an urban stream using dual nitrate isotopes. *Environmental Science & Technology* 48, 10580-10587.

Dong, X.M., Revol, J.-F., Gray, D.G., 1998. Effect of microcrystallite preparation conditions on the formation of colloid crystals of cellulose. *Cellulose* 5, 19-32.

Donohue, J., 2004. Copper in Drinking-water, Background document for development of WHO Guidelines for Drinking-water Quality. US Environmental Protection Agency.

Dore, A.J., Hallsworth, S., McDonald, A.G., Werner, M., Kryza, M., Abbot, J., Nemitz, E., Dore, C.J., Malcolm, H., Vieno, M., Reis, S., Fowler, D., 2014. Quantifying missing annual emission sources of heavy metals in the United Kingdom with an atmospheric transport model. *Science of the Total Environment* 479, 171-180.

Duan, C., Zhao, N., Yu, X., Zhang, X., Xu, J., 2013. Chemically modified kapok fiber for fast adsorption of Pb2+, Cd2+, Cu2+ from aqueous solution. *Cellulose* 20, 849-860.

Dufresne, A., 2006. Comparing the mechanical properties of high performances polymer nanocomposites from biological sources. *Nanoscience and Nanotechnology* 6, 1701-1172.

Dufresne, A., 2012. Nanocellulose: From Nature to High Performance Tailored Materials. De Gruyter.

Ebrahimzadeh, H., Tavassoli, N., Sadeghi, O., Amini, M.M., 2012. Optimization of solid-phase extraction using artificial neural networks and response surface methodology in combination with experimental design for determination of gold by atomic absorption spectrometry in industrial wastewater samples. *Talanta* 97, 211-217.

Eckhardt, D.A.V., 2012. Land Use and Water Quality.

Eichhorn, S.J., Dufresne, A., Aranguren, M., Marcovich, N.E., Capadona, J.R., Rowan, S.J., Weder, C., Thielemans, W., Roman, M., Renneckar, S., Gindl, W., Veigel, S., Keckes, J., Yano, H., Abe, K., Nogi, M., Nakagaito, A.N., Mangalam, A., Simonsen, J., Benight, A.S., Bismarck, A., Berglund, L.A., Peijs, T., 2009. Review: current international research into cellulose nanofibres and nanocomposites. *Journal of Materials Science* 45, 1-33.

Elazzouzi-Hafraoui, S., Nishiyama, Y., Putaux, J.-L., Heux, L., Dubreuil, F., Rochas, C., 2008. The shape and size distribution of crystalline nanoparticles prepared by acid hydrolysis of native cellulose. *Biomacromolecules* 9, 57-65.

Elmolla, E.S., Chaudhuri, M., Eltoukhy, M.M., 2010. The use of artificial neural network (ANN) for modeling of COD removal from antibiotic aqueous solution by the Fenton process. *Journal of Hazardous materials* 179, 127-134.

Engin, M.S., Uyanik, A., Cay, S., Icbudak, H., 2010. Effect of the adsorptive character of filter papers on the concentrations determined in studies involving heavy metal ions. *Adsorption Science and Technology* 28, 837-846.

EPA, 1995. Technology Transfer Network. United States Environmental Protection Agency

Eslamian, S., 2016. Urban Water Reuse Handbook. CRC Press.

Eyley, S., Thielemans, W., 2011. Imidazolium grafted cellulose nanocrystals for ion exchange applications. *Chemical Communications* 47, 4177-4179.

Eyley, S., Thielemans, W., 2014. Surface modification of cellulose nanocrystals. *Nanoscale* 6, 7764-7779.

Foo, K.Y., Hameed, B.H., 2010. Insights into the modeling of adsorption isotherm systems. *Chemical Engineering Journal* 156, 2-10.

Fu, F., Wang, Q., 2011. Removal of heavy metal ions from wastewaters: A review. *Journal of Environmental Management* 92, 407-418.

Gao, J.-J., Luo, W., Xi, X.-X., 2014. Estimation inventory of polycyclic aromatic hydrocarbons (PAHs) from anthropogenic sources and its impacts within the Yanghe Watershed, an important water-source site of Beijing, China. *Chinese Journal of Environmental Science* 35, 4573-4581.

Gardner, M., Comber, S., Scrimshaw, M.D., Cartmell, E., Lester, J., Ellor, B., 2012. The significance of hazardous chemicals in wastewater treatment works effluents. *Science of the Total Environment* 437, 363-372.

Gardner, M., Jones, V., Comber, S., Scrimshaw, M.D., Coello-Garcia, T., Cartmell, E., Lester, J., Ellor, B., 2013. Performance of UK wastewater treatment works with respect to trace contaminants. *Science of the Total Environment* 456, 359-369.

Gavrilescu, M., 2004. Removal of heavy metals from the environment by biosorption. *Engineering in Life Sciences* 4, 219-232.

Gengec, E., Ozdemir, U., Ozbay, B., Ozbay, I., Veli, S., 2013. Optimizing dye adsorption onto a waste-derived (modified charcoal ash) adsorbent using Box-Behnken and central composite design procedures. *Water Air and Soil Pollution* 224.

Geyikci, F., Kilic, E., Coruh, S., Elevli, S., 2012. Modelling of lead adsorption from industrial sludge leachate on red mud by using RSM and ANN. *Chemical Engineering Journal* 183, 53-59.

Ghorbani, F., Younesi, H., Ghasempouri, S.M., Zinatizadeh, A.A., Amini, M., Daneshi, A., 2008. Application of response surface methodology for optimization of cadmium biosorption in an aqueous solution by Saccharomyces cerevisiae. *Chemical Engineering Journal* 145, 267-275.

Ghosh, A., Das, P., Sinha, K., 2015. Modeling of biosorption of Cu(II) by alkalimodified spent tea leaves using response surface methodology (RSM) and artificial neural network (ANN). *Applied Water Science* 5, 191-199.

Ghosh, A., Sinha, K., Das Saha, P., 2013. Central composite design optimization and artificial neural network modeling of copper removal by chemically modified orange peel. *Desalination and Water Treatment* 51, 7791-7799.

Góes, M.M., Keller, M., Masiero Oliveira, V., Villalobos, L.D.G., Moraes, J.C.G., Carvalho, G.M., 2016. Polyurethane foams synthesized from cellulosebased wastes: Kinetics studies of dye adsorption. *Industrial Crops and Products* 85, 149-158.

Goldstein, J., Newbury, D.E., Joy, D.C., Lyman, C.E., Echlin, P., Lifshin, E., Sawyer, L., Michael, J.R., 2012. Scanning Electron Microscopy and X-ray Microanalysis: Third Edition. Springer US.

Gorelov, B.M., Dyakin, V.V., Kashin, G.N., Makhnjuk, V.I., Morozovskaya, D.V., Sidorchuk, V.A., 1994. Effect of physical and chemical adsorption of water molecules on YBa2Cu3O7– δ surface composition. *Journal of Electron Spectroscopy and Related Phenomena* 70, 161-165.

Gracia-Lor, E., Sancho, J.V., Serrano, R., Hernández, F., 2012. Occurrence and removal of pharmaceuticals in wastewater treatment plants at the Spanish Mediterranean area of Valencia. *Chemosphere* 87, 453-462.

Greenwood, R., Kendall, K., 1999. Selection of suitable dispersants for aqueous suspensions of Zirconia and Titania powders using Acoustophoresis. *Journal of the European Ceramic Society* 19, 479-488.

Guclu, G., Gurdag, G., Ozgumus, S., 2003. Competitive removal of heavy metal ions by cellulose graft copolymers. *Journal of Applied Polymer Science* 90, 2034-2039.

Habibi, Y., 2014. Key advances in the chemical modification of nanocelluloses. *Chemical Society Reviews* 43, 1519-1542.

Habibi, Y., Chanzy, H., Vignon, M.R., 2006. TEMPO-mediated surface oxidation of cellulose whiskers. *Cellulose* 13, 679-687.

Hamed, M.M., Khalafallah, M.G., Hassanien, E.A., 2004. Prediction of wastewater treatment plant performance using artificial neural networks. *Environmental Modelling & Software* 19, 919-928.

Hamid, H.A., Jenidi, Y., Thielemans, W., Somerfield, C., Gomes, R.L., Predicting the capability of carboxylated cellulose nanowhiskers for the remediation of copper from water using response surface methodology (RSM) and artificial neural network (ANN) models. *Industrial Crops and Products*.

Hamsaveni, D.R., Prapulla, S.G., Divakar, S., 2001. Response surface methodological approach for the synthesis of isobutyl isobutyrate. *Process Biochemistry* 36, 1103-1109.

Hanafiah, M.A.K.M., Alias, M., Khalir, W.K.A.W.M., Zakaria, H., 2014. Enhanced adsorption of Pb(II) on chemically treated neem (Azadirachta indica) leaf powder, in: Gupta, K.M. (Ed.), Material Science and Engineering Technology Ii, pp. 128-133.

Hanaki, K., 2008. Urban Environmental Management and Technology. Springer Japan.

Hasenmueller, E.A., Criss, R.E., 2013. Multiple sources of boron in urban surface waters and groundwaters. *Science of the Total Environment* 447, 235-247.

He, Z., Weng, H., Ho, H.-C., Ran, Q., Mao, M., 2014. Soil erosion and pollutant transport during rainfall-runoff processes. *Water Resources* 41, 604-611.

Health, P., 1992. Toxiocological Profile for Pyridine. U.S. Public Health.

Henze, M., VanLoosdrecht, M.C.M., Ekama, G.A., Brdjanovic, D., 2008. Biological Wastewater Treatment: Principles, Modelling and Design.

Himanshu, P., Vashi, R.T., 2014. COD and BOD removal from textile wastewater using naturally prepared adsorbents and their activations forms using sulphuric acid, Wastewater Engineering: Types, Characteristics and Treatment Techniques (Book Chapter) Communicated.

Ho, Y.S., 2004. Selection of optimum sorption isotherm. Carbon 42, 2115-2116.

Horikawa, Y., Itoh, T., Sugiyama, J., 2006. Preferential uniplanar orientation of cellulose microfibrils reinvestigated by the FTIR technique. *Cellulose* 13, 309-316.

Hsieh, Y.-L., 2013. Cellulose nanocrystals and self-assembled nanostructures from cotton, rice straw and grape skin: a source perspective. *Journal of Materials Science* 48, 7837-7846.

Hu, J., Gan, F., Triantafyllidou, S., Nguyen, C.K., Edwards, M.A., 2012. Copperinduced metal release from lead pipe into drinking water. *CORROSION* 68, 1037-1048.

Hughes, T.A., Gray, N.F., 2013a. Co-treatment of acid mine drainage with municipal wastewater: performance evaluation. *Environmental Science and Pollution Research* 20, 7863-7877.

Hughes, T.A., Gray, N.F., 2013b. Removal of metals and acidity from acid mine drainage using municipal wastewater and activated sludge. *Mine Water and the Environment* 32, 170-184.

Hutton, M., Symon, C., 1986. The quantities of cadmium, lead, mercury and arsenic entering the UK environment from human activities. *Science of the Total Environment* 57, 129-150.

Hwang, H.-H., Panno, S.V., Hackley, K.C., 2015. Sources and changes in groundwater quality with increasing urbanization, Northeastern Illinois. *Environmental & Engineering Geoscience* 21, 75-90.

Ileri, O., Cay, S., Uyanik, A., Erduran, N., 2014. Removal of common heavy metals from aqueous solutions by waste salvadora persica L. Branches (Miswak). *International Journal of Environmental Research* 8, 987-996.

Iqbal, J., Tirmizi, S.A., Mirza, M.L., Iqbal, J., 2005. Adsorption status of some transition metal ions on pretreated fish scales. *Journal of the Chemical Society of Pakistan* 27, 77-81.

Ishizaki, K., Komarneni, S., Nanko, M., 2013. Porous Materials: Process technology and applications. Springer US.

Isobe, N., Chen, X., Kim, U.J., Kimura, S., Wada, M., Saito, T., Isogai, A., 2013. TEMPO-oxidized cellulose hydrogel as a high-capacity and reusable heavy metal ion adsorbent. *Journal of Hazardous materials* 260, 195-201.

Javadian, H., Ghorbani, F., Tayebi, H., Asl, S.H., 2015. Study of the adsorption of Cd (II) from aqueous solution using zeolite-based geopolymer, synthesized from coal fly ash; kinetic, isotherm and thermodynamic studies. *Arabian Journal of Chemistry* 8, 837-849.

Jelic, A., Gros, M., Ginebreda, A., Cespedes-Sánchez, R., Ventura, F., Petrovic, M., Barcelo, D., 2011. Occurrence, partition and removal of pharmaceuticals in sewage water and sludge during wastewater treatment. *Water Research* 45, 1165-1176.

Jiang, F., Kittle, J.D., Tan, X., Esker, A.R., Roman, M., 2013. Effects of sulfate groups on the adsorption and activity of cellulases on cellulose substrates. *Langmuir* 29, 3280-3291.

Jin, L., Wei, Y., Xu, Q., Yao, W., Cheng, Z., 2014. Cellulose nanofibers prepared from TEMPO-oxidation of kraft pulp and its flocculation effect on kaolin clay. *Journal of Applied Polymer Science* 131.

Kadirvelu, K., Thamaraiselvi, K., Namasivayam, C., 2001. Removal of heavy metals from industrial wastewaters by adsorption onto activated carbon prepared from an agricultural solid waste. *Bioresource Technology* 76, 63-65.

Karamouz, M., Moridi, A., Nazif, S., 2010. Urban Water Engineering and Management. CRC Press.

Karnitz, J.O., Alves Gurgel, L.V., de Freitas, R.P., Gil, L.F., 2009. Adsorption of Cu(II), Cd(II), and Pb(II) from aqueous single metal solutions by mercerized cellulose and mercerized sugarcane bagasse chemically modified with EDTA dianhydride (EDTAD). *Carbohydrate Polymers* 77, 643-650.

Karnitz, O., Jr., Alves Gurgel, L.V., Perin de Melo, J.C., Botaro, V.R., Sacramento Melo, T.M., de Freitas Gil, R.P., Gil, L.F., 2007. Adsorption of heavy metal ion from aqueous single metal solution by chemically modified sugarcane bagasse. *Bioresource Technology* 98, 1291-1297.

Kasnejad, M.H., Esfandiari, A., Kaghazchi, T., Asasian, N., 2012. Effect of preoxidation for introduction of nitrogen containing functional groups into the structure of activated carbons and its influence on Cu (II) adsorption. *Journal of the Taiwan Institute of Chemical Engineers* 43, 736-740.

Kaszuba, M., Corbett, J., Watson, F.M., Jones, A., 2010. High-concentration zeta potential measurements using light-scattering techniques. *Philosophical Transactions of the Royal Society of London A: Mathematical, Physical and Engineering Sciences* 368, 4439-4451.

Kay, A.R., 2004. Detecting and minimizing zinc contamination in physiological solutions. *BMC Physiology* 4, 1.

Klemm, D., Philipp, B., Heinze, T., Heinze, U., Wagenknecht, W., 2004. General Considerations on Structure and Reactivity of Cellulose: Section 2.1–2.1.4, Comprehensive Cellulose Chemistry. Wiley-VCH Verlag GmbH & Co. KGaA, pp. 9-29.

Krishna, D., Sree, R.P., 2013. Response surface modeling and optimization of Chromium (VI) removal from waste water using custard apple peel powder. *2013* 11, 8.

Kulbat, E., Olanczuk-Neyman, K., Quant, B., Geneja, M., Haustein, E., 2003. Heavy metals removal in the mechanical-biological wastewater treatment plant "Wschod" in Gdansk. *Polish Journal of Environmental Studies* 12, 635-641.

Kurniawan, T.A., Chan, G. Y. S., Lo, W. H., Babel, S., 2006. Physico-chemical treatment techniques for wastewater laden with heavy metals. *Chemical Engineering Journal* 118, 83-98.

Kushwaha, J.P., Srivastava, V.C., Mall, I.D., 2010. Organics removal from dairy wastewater by electrochemical treatment and residue disposal. *Separation and Purification Technology* 76, 198-205.

Kvien, I., Tanem, B.S., Oksman, K., 2005. Characterization of cellulose whiskers and their nanocomposites by atomic force and electron microscopy. *Biomacromolecules* 6, 3160-3165.

Kyzas, G., Fu, J., Matis, K., 2013. The change from past to future for adsorbent materials in treatment of dyeing wastewaters. *Materials* 6, 5131.

Labet, M., Thielemans, W., 2011. Improving the reproducibility of chemical reactions on the surface of cellulose nanocrystals: ROP of epsilon-caprolactone as a case study. *Cellulose* 18, 607-617.

Landry, V., Alemdar, A., Blanchet, P., 2011. Nanocrystalline cellulose: Morphological, physical, and mechanical properties. *Forest Products Journal* 61, 104-112.

Langmuir, I., 1918. The adsorption of gases on plane surfaces of glass, mica and platinum. *Journal of American Chemical Society*, 1361-1403.

Levlin, E., 2010. Conductivity measurements for controlling municipal wastewater, In *proceedings* of a Polish-Swedish-Ukrainian Seminar, pp. 51-62.

Li, L., Li, X., Yan, C., Guo, W., Yang, T., Fu, J., Tang, J., Hu, C., 2014. Optimization of methyl orange removal from aqueous solution by response surface methodology using spent tea leaves as adsorbent. *Frontiers of Environmental Science & Engineering* 8, 496-502.

Li, Q., Chai, L., Wang, Q., Yang, Z., Yan, H., Wang, Y., 2010. Fast esterification of spent grain for enhanced heavy metal ions adsorption. *Bioresource Technology* 101, 3796-3799.

Lim, S.-R., Schoenung, J.M., 2010. Human health and ecological toxicity potentials due to heavy metal content in waste electronic devices with flat panel displays. *Journal of Hazardous materials* 177, 251-259.

Limousin, G., Gaudet, J.P., Charlet, L., Szenknect, S., Barthes, V., Krimissa, M., 2007. Sorption isotherms: A review on physical bases, modeling and measurement. *Applied Geochemistry* 22, 249-275.

Liu, D., Wang, S., Ma, Z., Tian, D., Gu, M., Lin, F., 2014a. Structure - color mechanism of iridescent cellulose nanocrystal films. *Royal Society of Chemistry Advances* 4, 39322-39331.

Liu, F., Luo, X., Lin, X., Liang, L., Chen, Y., 2009. Removal of copper and lead from aqueous solution by carboxylic acid functionalized deacetylated konjac glucomannan. *Journal of Hazardous materials* 171, 802-808.

Liu, H., Feng, S., Zhang, N., Du, X., Liu, Y., 2014b. Removal of Cu(II) ions from aqueous solution by activated carbon impregnated with humic acid. *Frontiers of Environmental Science & Engineering* 8, 329-336.

Liu, J., Yan, M., Zhang, Y.-K., Du, K.-F., 2011. Study of glutamate-modified cellulose beads for Cr(III) adsorption by response surface methodology. *Industrial & Engineering Chemistry Research* 50, 10784-10791.

Liu, J.-S., Guo, L.-C., Luo, X.-L., Chen, F.-R., Zeng, E.Y., 2014c. Impact of anthropogenic activities on urban stream water quality: a case study in Guangzhou, China. *Environmental Science and Pollution Research* 21, 13412-13419.

Liu, M., Yuan, Q., Jia, H., Li, S., Wang, X., Wang, C., 2013. Cysteine-modified orange peel for removal of Cu(II) from aqueous solutions. *Water Science and Technology* 67, 2444-2450.

Liu, M.H., Deng, Y., Zhan, H.Y., Zhang, X.S., 2002. Adsorption and desorption of copper(II) from solutions on new spherical cellulose adsorbent. *Journal of Applied Polymer Science* 84, 478-485.

Liu, P., Oksman, K., Mathew, A.P., 2016. Surface adsorption and self-assembly of Cu(II) ions on TEMPO-oxidized cellulose nanofibers in aqueous media. *Journal of Colloid and Interface Science* 464, 175-182.

Lu, D., Cao, Q., Cao, X., Luo, F., 2009. Removal of Pb(II) using the modified lawny grass: Mechanism, kinetics, equilibrium and thermodynamic studies. *Journal of Hazardous materials* 166, 239-247.

Lu, P., Hsieh, Y.L., 2010. Preparation and properties of cellulose nanocrystals: Rods, spheres, and network. *Carbohydrate Polymers* 82, 329-336.

Lu, P., Hsieh, Y.L., 2012. Preparation and characterization of cellulose nanocrystals from rice straw. *Carbohydrate Polymers* 87, 564-573.

Lu, Y., Weng, L., Cao, X., 2006. Morphological, thermal and mechanical properties of ramie crystallites—reinforced plasticized starch biocomposites. *Carbohydrate Polymers*, 198-204.

Lysenko, G., Squires, V., Verheye, W.H., 2010. Interactions: Food, Agriculture And Environment - Volume I.

Ma, L., Zhang, Y., Cao, J., Yao, J., 2014. Preparation of unmodified cellulose nanocrystals from Phyllostachys heterocycla and their biocompatibility evaluation. *BioResources* 9, 210-217.

Maheshwari, U., Gupta, S., 2016. Removal of Cr(VI) from wastewater using activated neem bark in a fixed-bed column: interference of other ions and kinetic modelling studies. *Desalination and Water Treatment* 57, 8514-8525.

Malakootian, M., Toolabi, A., Moussavi, S.G., Ahmadian, M., 2011. Equilibrium and kinetic modeling of heavy metals biosorption from three different real industrial wastewaters onto ulothrix zonata algae. *Australian Journal of Basic and Applied Sciences* 5.

Mandal, S., Mahapatra, S.S., Sahu, M.K., Patel, R.K., 2015. Artificial neural network modelling of As(III) removal from water by novel hybrid material. *Process Safety and Environmental Protection* 93, 249-264.

Marsalek, J., Cisneros, B.J., Karamouz, M., Malmquist, P.A., Goldenfum, J.A., Chocat, B., 2008. Urban Water Cycle Processes and Interactions: Urban Water Series - UNESCO-IHP. Taylor & Francis.

Mason, L.G., 2007. Focus on Hazardous Materials Research. Nova Science Publishers.

Mata, Y.N., Blazquez, M.L., Ballester, A., Gonzalez, F., Munoz, J.A., 2008. Characterization of the biosorption of cadmium, lead and copper with the brown alga Fucus vesiculosus. *Journal of Hazardous materials* 158, 316-323.

Matoug, I.O.M., 2013. Study of metals contamination in different filter papers. *Journal Of Environmental Science, Toxicology And Food Technology* 3, 59-61.

McBain, J.W., 1909. The mechanism of the adsorption ("sorption") of hydrogen by carbon. *The London, Edinburgh, and Dublin Philosophical Magazine and Journal of Science* 18, 916-935.

McDowall, D.J., Gupta, B.S., Stannett, V.T., 1984. Grafting of vinyl monomers to cellulose by ceric ion initiation. *Progress in Polymer Science* 10, 1-50.

Mehdinia, A., Shegefti, S., Shemirani, F., 2015. Removal of Lead(II), Copper(II) and Zinc(II) Ions from Aqueous Solutions Using Magnetic Amine-Functionalized Mesoporous Silica Nanocomposites. *Journal of the Brazilian Chemical Society* 26, 2249-2257.

Meireles, M.R., Almeida, P.E., Simões, M.G., 2003. A comprehensive review for industrial applicability of artificial neural networks. *IEEE transactions on industrial electronics* 50, 585-601.

Melignani, E., Isabel de Cabo, L., Maria Faggi, A., 2015. Copper uptake by Eichhornia crassipes exposed at high level concentrations. *Environmental Science and Pollution Research* 22, 8307-8315.

Merino-Solís, M., Villegas, E., de Anda, J., López-López, A., 2015. The effect of the hydraulic retention time on the performance of an ecological wastewater treatment system: An anaerobic filter with a constructed wetland. *Water* 7, 1149.

Mobasherpour, I., Salahi, E., Ebrahimi, M., 2014. Thermodynamics and kinetics of adsorption of Cu(II) from aqueous solutions onto multi-walled carbon nanotubes. *Journal of Saudi Chemical Society* 18, 792-801.

Mohajeri, S., Aziz, H.A., Zahed, M.A., Mohajeri, L., Bashir, M.J., Aziz, S.Q., Adlan, M.N., Isa, M.H., 2011. Multiple responses analysis and modeling of

Fenton process for treatment of high strength landfill leachate. *Water Sci Technol* 64, 1652-1660.

Mohan, S., Singh, Y., Verma, D.K., Hasan, S.H., 2015. Synthesis of CuO nanoparticles through green route using citrus limon juice and its application as nanosorbent for Cr(VI) remediation: process optimization with RSM and ANN-GA based model. *Process Safety and Environmental Protection* 96, 156-166.

Moon, R.J., Martini, A., Nairn, J., Simonsen, J., Youngblood, J., 2011. Cellulose nanomaterials review: structure, properties and nanocomposites. *Chemical Society Reviews* 40, 3941-3994.

Morandi, G., Heath, L., Thielemans, W., 2009. Cellulose nanocrystals grafted with polystyrene chains through Surface-Initiated Atom Transfer Radical Polymerization (SI-ATRP). *Langmuir* 25, 8280-8286.

Mudhoo, A., Sharma, S.K., Garg, V.K., Tseng, C.-H., 2011. Arsenic: An overview of applications, health, and environmental concerns and removal processes. *Critical Reviews in Environmental Science and Technology* 41, 435-519.

Munoz, I., Jose Gomez-Ramos, M., Aguera, A., Francisco Garcia-Reyes, J., Molina-Diaz, A., Fernandez-Alba, A.R., 2009. Chemical evaluation of contaminants in wastewater effluents and the environmental risk of reusing effluents in agriculture. *Trac-Trends in Analytical Chemistry* 28, 676-694.

Nada, A.M.A., El-Wakil, N.A., Hassan, M.L., Adel, A.M., 2006. Differential adsorption of heavy metal ions by cotton stalk cation-exchangers containing multiple functional groups. *Journal of Applied Polymer Science* 101, 4124-4132.

Nadeem, R., Hanif, M.A., Mahmood, A., Jamil, M.S., Ashraf, M., 2009. Biosorption of Cu(II) ions from aqueous effluents by blackgram bran (BGB). *Journal of Hazardous materials* 168, 1622-1625.

Nasr, M.S., Moustafa, M.A.E., Seif, H.A.E., El Kobrosy, G., 2012. Application of artificial neural network (ANN) for the prediction of EL-AGAMY wastewater treatment plant performance-EGYPT. *Alexandria Engineering Journal* 51, 37-43.

Nhapi, I.B., N.; Murenzi, R.; Sekomo, C. B.; Wali, U. G., 2011. Removal of heavy metals from industrial wastewater using rice husks. *The Open Environmental Engineering Journal* 4, 170-180.

O'Connell, D.W., Birkinshaw, C., O'Dwyer, T.F., 2006a. A chelating cellulose adsorbent for the removal of Cu(II) from aoueous solutions. *Journal of Applied Polymer Science* 99, 2888-2897.

O'Connell, D.W., Birkinshaw, C., O'Dwyer, T.F., 2006b. A modified cellulose adsorbent for the removal of nickel(II) from aqueous solutions. *Journal of Chemical Technology and Biotechnology* 81, 1820-1828.

O'Connell, D.W., Birkinshaw, C., O'Dwyer, T.F., 2008a. Heavy metal adsorbents prepared from the modification of cellulose: a review. *Bioresource Technology* 99, 6709-6724.

O'Connell, D.W., Birkinshaw, C., O'Dwyer, T.F., 2008b. Removal of copper, nickel and lead from wastewater using a modified cellulose material: A

Comparison, in: Zamorano, M., Brebbia, C.A., Kungolos, A., Popov, V., Itoh, H. (Eds.), Waste Management and the Environment Iv, pp. 809-818.

Oezer, A., Guerbuez, G., Calmli, A., Koerbahti, B.K., 2008. Investigation of nickel(II) biosorption on Enteromorpha prolifera: Optimization using response surface analysis. *Journal of Hazardous materials* 152, 778-788.

Ofomaja, A.E., Naidoo, E.B., Modise, S.J., 2010a. Dynamic studies and pseudosecond order modeling of copper(II) biosorption onto pine cone powder. *Desalination* 251, 112-122.

Ofomaja, A.E., Unuabonah, E.I., Oladoja, N.A., 2010b. Competitive modeling for the biosorptive removal of copper and lead ions from aqueous solution by Mansonia wood sawdust. *Bioresource Technology* 101, 3844-3852.

Oguz, E., Ersoy, M., 2010. Removal of Cu(II) from aqueous solution by adsorption in a fixed bed column and Neural Network Modelling. *Chemical Engineering Journal* 164, 56-62.

Ohwoavworhua, F.O., Adelakun, T.A., 2010. Non-wood Fibre Production of Microcrystalline Cellulose from Sorghum caudatum: Characterisation and Tableting Properties. *Indian Journal of Pharmaceutical Sciences* 72, 295-301.

Okita, Y., Saito, T., Isogai, A., 2010. Entire surface oxidation of various cellulose microfibrils by TEMPO-mediated oxidation. *Biomacromolecules* 11, 1696-1700.

Oksman, K., Niska, K.O., Sain, M., 2006. Cellulose Nanocomposites: Processing, Characterization, and Properties. American Chemical Society.

Oloibiri, V., Ufomba, I., Chys, M., Audenaert, W.T.M., Demeestere, K., Van Hulle, S.W.H., 2015. A comparative study on the efficiency of ozonation and coagulation-flocculation as pretreatment to activated carbon adsorption of biologically stabilized landfill leachate. *Waste Management* 43, 335-342.

Oshima, T., Kondo, K., Ohto, K., Inoue, K., Baba, Y., 2008. Preparation of phosphorylated bacterial cellulose as an adsorbent for metal ions. *Reactive & Functional Polymers* 68, 376-383.

Osobamiro, M.T., Atewolara-Odule, O.C., 2015. Determination of physicochemical parameters and levels of some heavy metals in industrial wastewater. *International Journal of Scientific & Engineering Research* 6.

Pagnanelli, F., Mainelli, S., Vegliò, F., Toro, L., 2003. Heavy metal removal by olive pomace: biosorbent characterisation and equilibrium modelling. *Chemical Engineering Science* 58, 4709-4717.

Pakravan, P., Akhbari, A., Moradi, H., Azandaryani, A.H., Mansouri, A.M., Safari, M., 2015. Process modeling and evaluation of petroleum refinery wastewater treatment through response surface methodology and artificial neural network in a photocatalytic reactor using poly ethyleneimine (PEI)/titania (TiO2) multilayer film on quartz tube. *Applied Petrochemical Research* 5, 47-59.

Park, D., Yun, Y.S., Park, J.M., 2010. The past, present, and future trends of biosorption. *Biotechnology and Bioprocess Engineering* 15, 86-102.

Payen, A., 1838. Mémoire sur la composition du tissu propre des plantes et du ligneux. *Comptes Rendus* 7, 1052-1056.

Pazwash, H., 2011. Urban Storm Water Management. CRC Press.

Peng, Y., Gardner, D.J., Han, Y., 2012. Drying cellulose nanofibrils: in search of a suitable method. *Cellulose* 19, 91-102.

Pereira, F.V., Gurgel, L.V.A., de Aquino, S.F., Gil, L.F., 2009. Removal of Zn^{2+} from electroplating wastewater using modified wood sawdust and sugarcane bagasse. *Journal of Environmental Engineering-Asce* 135, 341-350.

Perkin Elmer, I., 1996. Analytical Methods for Atomic Absorption Spectrometry. The Perkin-Elmer Corporation.

Perkin Elmer, I., 2004. Sensitivity Versus Detection Limit.

Perkin Elmer, I., 2008. Atomic Spectroscopy: A Guide to Selecting the Appropriate Technique and System. PerkinElmer, Inc.

Pilkington, J.L., Preston, C., Gomes, R.L., 2014. Comparison of response surface methodology (RSM) and artificial neural networks (ANN) towards efficient extraction of artemisinin from Artemisia annua. *Industrial Crops and Products* 58, 15-24.

Podstawczyk, D., Witek-Krowiak, A., Dawiec, A., Bhatnagar, A., 2015. Biosorption of copper(II) ions by flax meal: Empirical modeling and process optimization by response surface methodology (RSM) and artificial neural network (ANN) simulation. *Ecological Engineering* 83, 364-379.

Postek, M.T., Vladar, A., Dagata, J., Farkas, N., Ming, B., Wagner, R., Raman, A., Moon, R.J., Sabo, R., Wegner, T.H., Beecher, J., 2011. Development of the metrology and imaging of cellulose nanocrystals. *Measurement Science & Technology* 22.

Pourjavadi, A., Doulabi, M., Alamolhoda, A.A., Tavakkoli, E., Amirshekari, S., 2013. Synthesis and characterization of salep sulfate and its utilization in preparation of heavy metal ion adsorbent. *Journal of Applied Polymer Science* 130, 3001-3008.

Rajemahadik, C.F., Kulkarni, S.V., Kulkarni, D.G.S., 2013. Efficient removal of heavy metals from electroplating wastewater using electrocoagulation. *International Journal of Scientific and Research Publications* 3.

Ramwell, C.T., Kah, M., Johnson, P.D., 2014. Contribution of household herbicide usage to glyphosate and its degradate aminomethylphosphonic acid in surface water drains. *Pest Management Science* 70, 1823-1830.

Ranjan, D., Mishra, D., Hasan, S.H., 2011. Bioadsorption of Arsenic: An Artificial Neural Networks and Response Surface Methodological Approach. *Industrial & Engineering Chemistry Research* 50, 9852-9863.

Rao, K.S., Anand, S., Rout, K., Venkateswarlu, P., 2012. Response surface optimization for removal of cadmium from aqueous solution by waste agricultural biosorbent psidium guvajava L. leaf powder. *Clean-Soil Air Water* 40, 80-86.

Rastegar, S.O., Mousavi, S.M., Shojaosadati, S.A., Sheibani, S., 2011. Optimization of petroleum refinery effluent treatment in a UASB reactor using response surface methodology. *Journal of Hazardous materials* 197, 26-32.

Ratcliffe, H.E., Swanson, G.M., Fischer, L.J., 1996. Human exposure to mercury: A critical assessment of the evidence of adverse health effects. *Journal of Toxicology and Environmental Health* 49, 221-270.

Reddy, D.H.K., Lee, S.M., Seshaiah, K., 2012. Biosorption of toxic heavy metal ions from water environment using honeycomb biomass—An industrial waste material. *Water, Air, & Soil Pollution* 223, 5967-5982.

Reddy, D.H.K., Seshaiah, K., Reddy, A. V. R., Lee, S. M., 2012. Optimization of Cd(II), Cu(II) and Ni(II) biosorption by chemically modified moringa oleifera leaves powder. *Carbohydrate Polymers* 88, 1077-1086.

Revitt, D.M., Lundy, L., Coulon, F., Fairley, M., 2014. The sources, impact and management of car park runoff pollution: A review. *Journal of Environmental Management* 146, 552-567.

Rhind, S.M., 2009. Anthropogenic pollutants: a threat to ecosystem sustainability? *Philosophical Transactions of the Royal Society Biological Sciences* 364, 3391-3401.

Rodushkin, I., Engstroem, E., Baxter, D.C., 2010. Sources of contamination and remedial strategies in the multi-elemental trace analysis laboratory. *Analytical and Bioanalytical Chemistry* 396, 365-377.

Rouquerol, J., Rouquerol, F., Llewellyn, P., Maurin, G., Sing, K.S.W., 2013. Adsorption by Powders and Porous Solids: Principles, Methodology and Applications. Elsevier Science.

Sadrzadeh, M., Mohammadi, T., Ivakpour, J., Kasiri, N., 2009. Neural network modeling of Pb²⁺ removal from wastewater using electrodialysis. *Chemical Engineering and Processing: Process Intensification* 48, 1371-1381.

Saha, P.D., 2013. Mathematical modeling of the reduction of Safranin onto chemically modified rice husks in stirred tank reactor using response surface methodology and artificial neural network. *Bioremediation Journal* 17, 52-60.

Sahan, T., Ceylan, H., Sahiner, N., Aktas, N., 2010. Optimization of removal conditions of copper ions from aqueous solutions by Trametes versicolor. *Bioresource Technology* 101, 4520-4526.

Saiano, F., Ciofalo, M., Cacciola, S.O., Ramirez, S., 2005. Metal ion adsorption by Phomopsis sp biomaterial in laboratory experiments and real wastewater treatments. *Water Research* 39, 2273-2280.

Saito, T., Hirota, M., Tamura, N., Kimura, S., Fukuzumi, H., Heux, L., Isogai, A., 2009. Individualization of nano-sized plant cellulose fibrils by direct surface carboxylation using TEMPO catalyst under neutral conditions. *Biomacromolecules* 10, 1992-1996.

Saito, T., Isogai, A., 2004. TEMPO-mediated oxidation of native cellulose: the effect of oxidation conditions on chemical and crystal structures of the water-insoluble fractions. *Biomacromolecules* 5, 1983-1989.

Saito, T., Isogai, A., 2006. Introduction of aldehyde groups on surfaces of native cellulose fibers by TEMPO-mediated oxidation. *Colloids and Surfaces A: Physicochemical and Engineering Aspects* 289, 219-225.

Saito, T., Shibata, I., Isogai, A., Suguri, N., Sumikawa, N., 2005. Distribution of carboxylate groups introduced into cotton linters by the TEMPO-mediated oxidation. *Carbohydrate Polymers* 61, 414-419.

Santos, S.C.R., Boaventura, R.A.R., 2008. Adsorption modelling of textile dyes by sepiolite. *Applied Clay Science* 42, 137-145.

Sarı, A., Tuzen, M., 2009. Kinetic and equilibrium studies of biosorption of Pb(II) and Cd(II) from aqueous solution by macrofungus (Amanita rubescens) biomass. *Journal of Hazardous materials* 164, 1004-1011.

Sarkar, M., Majumdar, P., 2011. Application of response surface methodology for optimization of heavy metal biosorption using surfactant modified chitosan bead. *Chemical Engineering Journal* 175, 376-387.

Savic, I.M., Stojiljkovic, S.T., Stojanovic, S.B., Moder, K., 2012. Modeling and optimization of Fe(III) adsorption from water using bentonite clay: comparison of central composite design and artificial neural network. *Chemical Engineering & Technology* 35, 2007-2014.

Schäfer, A.I., Fane, A.G., Waite, T.D., 2005. Nanofiltration: Principles and Applications. Elsevier Advanced Technology.

Sciban, M., Radetic, B., Kevresan, D., Klasnja, M., 2007. Adsorption of heavy metals from electroplating wastewater by wood sawdust. *Bioresource Technology* 98, 402-409.

Sedighi, M., Ghasemi, M., Hassan, S.H.A., Daud, W.R.W., Ismail, M., Abdallah, E., 2012. Process optimization of batch biosorption of lead using Lactobacillius bulgaricus in an aqueous phase system using response surface methodology. *World Journal of Microbiology & Biotechnology* 28, 2047-2055.

Sengil, I.A., Oezacar, M., 2008. Biosorption of Cu(II) from aqueous solutions by mimosa tannin gel. *Journal of Hazardous materials* 157, 277-285.

Settle, F.A., 1997. Handbook of Instrumental Techniques for Analytical Chemistry. Prentice Hall PTR.

Shanmugaprakash, M., Sivakumar, V., 2013. Development of experimental design approach and ANN-based models for determination of Cr(VI) ions uptake rate from aqueous solution onto the solid biodiesel waste residue. *Bioresource Technology* 148, 550-559.

Shojaeimehr, T., Rahimpour, F., Khadivi, M.A., Sadeghi, M., 2014. A modeling study by response surface methodology (RSM) and artificial neural network (ANN) on Cu²⁺ adsorption optimization using light expended clay aggregate (LECA). *Journal of Industrial and Engineering Chemistry* 20, 870-880.

Shukla, S.R., Pai, R.S., 2005a. Adsorption of Cu(II), Ni(II) and Zn(II) on dye loaded groundnut shells and sawdust. *Separation and Purification Technology* 43, 1-8.

Shukla, S.R., Pai, R.S., 2005b. Adsorption of Cu(II), Ni(II) and Zn(II) on modified jute fibres. *Bioresource Technology* 96, 1430-1438.

Silverstein, R.M., Webster, F.X., Kiemle, D.J., Bryce, D.L., 2014. Spectrometric Identification of Organic Compounds. Wiley.

Singh, P., Shook, J.R., DiPalma, M., 2015. Discovering Our World: Humanity's Epic Journey from Myth to Knowledge. Pitchstone Publishing.

Singh, R., Chadetrik, R., Kumar, R., Bishnoi, K., Bhatia, D., Kumar, A., Bishnoi, N.R., Singh, N., 2010. Biosorption optimization of lead(II), cadmium(II) and

copper(II) using response surface methodology and applicability in isotherms and thermodynamics modeling. *Journal of Hazardous materials* 174, 623-634.

Singh, S.N., Srivastava, G., Bhatt, A., 2012. Physicochemical determination of pollutants in wastewater in Dheradun. *Current World Environment* 7, 133-138.

Sinha, K., Chowdhury, S., Das Saha, P., Datta, S., 2013. Modeling of microwaveassisted extraction of natural dye from seeds of Bixa orellana (Annatto) using response surface methodology (RSM) and artificial neural network (ANN). *Industrial Crops and Products* 41, 165-171.

Siqueira, G., Bras, J., Dufresne, A., 2010. Cellulosic bionanocomposites: A review of preparation, properties and applications. *Polymers* 2, 728-765.

Sjöström, E., 1993. Wood Chemistry: Fundamentals and Applications. Academic Press.

Spellman, F.R., 2013. Handbook of Water and Wastewater Treatment Plant Operations, Third Edition. CRC Press.

Srivastava, P., Hasan, S.H., 2011. Biomass of Mucor Hiemalis for the biosorption of cadmium from aqueous solutions: Equilibrium and kinetic studies.

Sugashini, S., Begum, K.M.M.S., 2013. Optimization using central composite design (CCD) for the biosorption of Cr(VI) ions by cross linked chitosan carbonized rice husk (CCACR). *Clean Technologies and Environmental Policy* 15, 293-302.

Takaichi, S., Saito, T., Tanaka, R., Isogai, A., 2014. Improvement of nanodispersibility of oven-dried TEMPO-oxidized celluloses in water. *Cellulose* 21, 4093-4103.

Tan, W.S., Ting, A.S.Y., 2014. Alginate-immobilized bentonite clay: adsorption efficacy and reusability for Cu(II) removal from aqueous solution. *Bioresource Technology* 160, 115-118.

Tchobanoglous, G., Burton, F.L., Metcalf, Eddy, 1991. Wastewater engineering: treatment, disposal, and reuse. McGraw-Hill.

Texier, A.C., Andrès, Y., Faur-Brasquet, C., Le Cloirec, P., 2002. Fixed-bed study for lanthanide (La, Eu, Yb) ions removal from aqueous solutions by immobilized Pseudomonas aeruginosa: experimental data and modelization. *Chemosphere* 47, 333-342.

Thirumavalavan, M., Lai, Y.L., Lin, L.C., Lee, J.F., 2010. Cellulose-based native and surface modified fruit peels for the adsorption of heavy metal ions from aqueous solution: langmuir adsorption isotherms. *Journal of Chemical and Engineering Data* 55, 1186-1192.

Thomas, R., 2013. Practical Guide to ICP-MS: A Tutorial for Beginners, Third Edition. CRC Press.

Tiryaki, S., Aydın, A., 2014. An artificial neural network model for predicting compression strength of heat treated woods and comparison with a multiple linear regression model. *Construction and Building Materials* 62, 102-108.

Tovar Gomez, R., Moreno Virgen, M.R., Dena Aguilar, J.A., Hernandez Montoya, V., Bonilla Petriciolet, A., Montes Moran, M.A., 2013. Modeling of

fixed-bed adsorption of fluoride on bone char using a hybrid neural network approach. *Chemical Engineering Journal* 228, 1098-1109.

Tunay, O., Tasli, R., Orhon, D., 1992. Factors affecting the performance of hydroxide precipitation of metals, in: Dalton, C.S., Wukasch, R.F. (Eds.), Proceedings of the 46th Industrial Waste Conference, May 14, 15, 16, 1991, Purdue University, pp. 467-478.

Turan, N.G., El, E.B., Ozgonenel, O., 2013a. Prediction of Heavy Metal Removal by Different Liner Materials from Landfill Leachate: Modeling of Experimental Results Using Artificial Intelligence Technique. *The Scientific World Journal* 2013, 5.

Turan, N.G., Mesci, B., Ozgonenel, O., 2011. The use of artificial neural networks (ANN) for modeling of adsorption of Cu(II) from industrial leachate by pumice. *Chemical Engineering Journal* 171, 1091-1097.

Turan, N.G., Mesci, B., Ozgonenel, O., 2013b. Response surface modeling of Cu(II) removal from electroplating waste by adsorption: application of BoxBehnken experimental design. *Clean-Soil Air Water* 41, 304-312.

Turek, N.F., Kasten, L., Lytle, D.A., Goltz, M.N., 2011. Impact of plumbing age on copper levels in drinking water. *Journal of Water Supply Research and Technology-Aqua* 60, 1-15.

UK, 2008. Proposals for environmental quality standards for annex VIII substances. UK Technical Advisory Group on the WFD.

UKWIR, 2015. Municipal Sewage, One Man's Waste, Another Man's Treasure! UK Water Industry Research (UKWIR). Ref:15/PM/03/5, 2015

Unuabonah, E.I., Guenter, C., Weber, J., Lubahn, S., Taubert, A., 2013. Hybrid Clay: A new highly efficient adsorbent for water treatment. *ACS Sustainable Chemistry & Engineering* 1, 966-973.

USEPA, 1998. Waste Water Collection, Treatment and Storage. United States Environmental Protection Agency

Vaughan, T., Seo, C.W., Marshall, W.E., 2001. Removal of selected metal ions from aqueous solution using modified corncobs. *Bioresource Technology* 78, 133-139.

Vera Candioti, L., De Zan, M.M., Cámara, M.S., Goicoechea, H.C., 2014. Experimental design and multiple response optimization. Using the desirability function in analytical methods development. *Talanta* 124, 123-138.

Vijayabhanu, R., Radha, V., 2013. A new feature selection algorithm using modified PSO for an anaerobic wastewater treatment system. *International Journal of Emerging Technologies in Computational and Applied Sciences* 4, 280-288.

Volesky, B., Holan, Z. R., 1995. Biosorption of heavy metals. *Biotechnology Progress* 11, 235-250.

Wågberg, L., Hägglund, R., 2001. Kinetics of Polyelectrolyte Adsorption on Cellulosic Fibers. *Langmuir* 17, 1096-1103.

Wang, B., Sain, M., Oksman, K., 2007. Study of structural morphology of hemp fiber from the micro to the nanoscale. *Applied Composite Materials* 14, 89-103.

Wang, J., Chen, C., 2006. Biosorption of heavy metals by Saccharomyces cerevisiae: A review. *Biotechnology Advances* 24, 427-451.

Wang, J., Chen, C., 2009. Biosorbents for heavy metals removal and their future. *Biotechnology Advances* 27, 195-226.

Wang, L.K., Chen, J.P., Hung, Y.T., Shammas, N.K., 2009. Heavy Metals in the Environment. CRC Press.

Wang, S., Wei, M., Huang, Y., 2013. Biosorption of multifold toxic heavy metal ions from aqueous water onto food residue eggshell membrane functionalized with ammonium thioglycolate. *Journal of Agricultural and Food Chemistry* 61, 4988-4996.

Wang, Z., Hattendorf, B., Günther, D., 2006. Analyte response in laser ablation Inductively Coupled Plasma Mass Spectrometry. *Journal of the American Society for Mass Spectrometry* 17, 641-651.

Wankasi, D., Horsfall Jnr, M., Ibuteme Spiff, A., 2006a. Sorption kinetics of Pb2+ and Cu2+ ions from aqueous solution by Nipah palm (Nypa fruticans Wurmb) shoot biomass. *Electronic Journal of Biotechnology* 9, 0-0.

Wankasi, D., Jnr., M.H., Spiff, A.I., 2006b. Sorption kinetics of Pb^{2+} and Cu^{2+} ions from aqueous solution by Nipah palm (Nypa fruticans Wurmb) shoot biomass.

WEF, 2007. Operation of Municipal Wastewater Treatment Plants Manual of Practice:Characterization and Sampling of Wastewater, 6th ed. Water Environment Federation (WEF) p. 41.

Welker, C.M., Balasubramanian, V.K., Petti, C., Rai, K.M., DeBolt, S., Mendu, V., 2015. Engineering plant biomass lignin content and composition for biofuels and bioproducts. *Energies* 8, 7654-7676.

WHO, 2006. Guidelines for the safe use of wastewater, excreta and greywater: wastewater use in agriculture, WHO Press, World Health Organization, p. 222.

Wiese, S.B.O., MacLeod, C.L., Lester, J.N., 1997. A recent history of metal accumulation in the sediments of the Thames Estuary, United Kingdom. *Estuaries* 20, 483-493.

Witek-Krowiak, A., Chojnacka, K., Podstawczyk, D., Dawiec, A., Pokomeda, K., 2014. Application of response surface methodology and artificial neural network methods in modelling and optimization of biosorption process. *Bioresource Technology* 160, 150-160.

Wong, K.K., Lee, C.K., Low, K.S., Haron, M.J., 2003. Removal of Cu and Pb from electroplating wastewater using tartaric acid modified rice husk. *Process Biochemistry* 39, 437-445.

Xu, L., Yao, Q., Deng, J., Han, Z., Zhang, Y., Fu, Y., Huber, G.W., Guo, Q., 2015. Renewable N-heterocycles production by thermocatalytic conversion and ammonization of biomass over ZSM-5. *ACS Sustainable Chemistry & Engineering* 3, 2890-2899.

Xu, X., Geng, W., Song, J., Yang, Y., Liu, M., 2011. Adsorption of Cd(II) and Cu(II) by Epichlorohydrin and Cysteine Modified Bagasse. *Asian Journal of Chemistry* 23, 1377-1380.

Yang, H., Liu, J., Yang, J., 2011. Leaching copper from shredded particles of waste printed circuit boards. *Journal of Hazardous materials* 187, 393-400.

Yang, Y.-Y., Toor, G.S., Williams, C.F., 2015. Pharmaceuticals and organochlorine pesticides in sediments of an urban river in Florida, USA. *Journal of Soils and Sediments* 15, 993-1004.

Yanna, L., Wang, C., Jia, Y., Wang, W., Ma, X., Du, J., Pu, G., Tian, X., 2014. Effects of sulfuric, nitric, and mixed acid rain on litter decomposition, soil microbial biomass, and enzyme activities in subtropical forests of China. *Applied Soil Ecology* 79, 1-9.

Ye, J., Zhang, P., Hoffmann, E., Zeng, G., Tang, Y., Dresely, J., Liu, Y., 2014. Comparison of Response Surface Methodology and Artificial Neural Network in Optimization and Prediction of Acid Activation of Bauxsol for Phosphorus Adsorption. *Water Air and Soil Pollution* 225.

Yu, L., Han, M., He, F., 2013a. A review of treating oily wastewater. *Arabian Journal of Chemistry*.

Yu, X., Tong, S., Ge, M., Wu, L., Zuo, J., Cao, C., Song, W., 2013b. Adsorption of heavy metal ions from aqueous solution by carboxylated cellulose nanocrystals. *Journal of Environmental Sciences* 25, 933-943.

Zhang, S., Sun, B., Wang, W., Zhu, M., Chen, J., 2006. Adsorption of Cd²⁺ and Cu²⁺ by oxidized cellulose from TEMPO-mediated selective oxidation of alkaline natural cellulose pulp. *Journal of Macromolecular Science Part A: Pure and Applied Chemistry* 43, 1895-1906.

Zhao, B.X., Wang, P., Zheng, T., Chen, C. Y., Shu, J., 2006. Preparation and adsorption performance of a cellulosic-adsorbent resin for copper(II). *Journal of Applied Polymer Science* 99, 2951-2956.

Zhou, Y., Zhang, L., Fu, S., Zheng, L., Zhan, H., 2012. Adsorption behavior of Cd2+, Pb2+, and Ni2+ from aqueous solutions on cellulose-based hydrogels. *BioResources* 7, 2752-2765.

Zhu, C.-S., Wang, L.-P., Chen, W.-b., 2009. Removal of Cu(II) from aqueous solution by agricultural by-product: Peanut hull. *Journal of Hazardous materials* 168, 739-746.

Zhu, R., Yongsheng, W., Qihui, L., Qian, T., 2011. The Second SREE Conference on Chemical Engineering (CCE 2011) Effect of Pb on growth, accumulation and quality component of tea plant. *Procedia Engineering* 18, 214-219.

Zolgharnein, J., Asanjarani, N., Mousavi, S.N., 2011. Optimization and characterization of Ti(I) adsorption onto modified Ulmus carpinifolia tree leaves. *Clean-Soil Air Water* 39, 250-258.

Zolgharnein, J., Shahmoradi, A., Ghasemi, J.B., 2013a. Comparative study of Box-Behnken, central composite, and Doehlert matrix for multivariate optimization of Pb (II) adsorption onto Robinia tree leaves. *Journal of Chemometrics* 27, 12-20.

Zolgharnein, J., Shariatmanesh, T., Asanjarani, N., 2013b. Cercis siliquastrum tree leaves as an efficient adsorbent for removal of Ag(I): Response surface

optimization and characterization of biosorption. *Clean-Soil Air Water* 41, 1183-1195.

Zougagh, M., de Torres, A.G., Pavon, J.M.C., 2002. Determination of cadmium in water by ICP-AES with on-line adsorption preconcentration using DPTH-gel and TS-gel microcolumns. *Talanta* 56, 753-761.

Zugenmaier, P., 2008. Crystalline Cellulose and Derivatives: Characterization and Structures. Springer Berlin Heidelberg.

Zulkali, M.M.D., Ahmad, A.L., Norulakmal, N.H., 2006. Oryza sativa L. husk as heavy metal adsorbent: optimization with lead as model solution. *Bioresource Technology* 97, 21-25.

of known concentration	1 1
Concentration (mg/L)	Absorbance
0.0	0.000
2.0	0.069
4.0	0.136

0.203

0.268

0.331

6.0

8.0

10.0

Appendix A-	Calibration	curves and	data for	· AAS
-------------	-------------	------------	----------	-------



Table A.1: Data collected using certified 1000 mg/L calibration standard solution of Cu (in 2 % HNO₃) of known concentration

Figure A.1: Calibration Curve of Cu(II) from standard solutions of Cu (2-10 mg/L) $\,$

Concentration (mg/L)	Absorbance
0.0	0.003
10.0	0.049
20.0	0.095
30.0	0.143
40.0	0.195
50.0	0.247
60.0	0.292
70.0	0.341

Table A.2: Data collected using certified 1000 mg/L calibration standard solution of Cu (in 2 % HNO₃) of known concentration



Figure A.2: Calibration Curve of Cu(II) from standard solutions of Cu (10-70 mg/L) $\,$



Appendix B – BET plots and N2 adsorption-desorption isotherms





Figure B.1: BET surface area plot of (a) unmodified and (b) modified CNWs



Figure B.2: N_2 adsorption-desorption isotherm at 77K of (a) unmodified and (b) modified CNWs



Appendix C – Isotherms and kinetic studies



Figure C.1: Langmuir isotherm plots for the adsorption of Cu(II) onto modified CNW adsorbents (temperature=; pH=; contact time=; sorbent dosage)



Figure C.2: Freundlich isotherm plots for the adsorption of Cu(II) onto modified CNW adsorbents (temperature=; pH=; contact time=; sorbent dosage)

Table C.3: Kinetic parameters obtained from pseudo-first order for Cu(II) removal using modified CNW adsorbents at different initial Cu(II) concentration (a) 10 mg/L and (b) 60 mg/L

t	qt	qe	qe-qt	$log(q_e-q_t)$
0	0	1.91	1.91	0.28
30	1.85	1.91	0.06	-1.19
60	1.86	1.91	0.05	-1.34
90	1.88	1.91	0.03	-1.46
120	1.88	1.91	0.03	-1.53
180	1.90	1.91	0.01	-1.91
240	1.91	1.91	0.00	-2.53

(a) 10 mg/L

(b) 60 mg/L

t	qt	qe	qe-qt	$log(q_e-q_t)$
0	0.00	9.19	9.19	0.96
30	8.21	9.19	0.98	-0.01
60	8.58	9.19	0.61	-0.21
90	8.59	9.19	0.60	-0.23
120	8.97	9.19	0.22	-0.66
180	9.00	9.19	0.19	-0.73
240	9.18	9.19	0.01	-2.01

t= time; q_t = adsorption capacity at time t; q_e = adsorption capacity at equilibrium

Table C.4: Kinetic parameters obtained from pseudo-second order for Cu(II) removal using modified CNW adsorbents at different initial Cu(II) concentration (a) 10 mg/L and (b) 60 mg/L

t	qt	q _e	t/q_t
0	0	1.91	
30	1.85	1.91	16.26
60	1.86	1.91	32.19
90	1.88	1.91	48.00
120	1.88	1.91	63.81
180	1.90	1.91	94.85
240	1.91	1.91	125.85

(a) 10 mg/L

(b) 60 mg/L

t	q_t	q e	t/qt
0	0	9.19	
30	8.21	9.19	3.66
60	8.58	9.19	6.99
90	8.59	9.19	10.47
120	8.97	9.19	13.38
180	9.00	9.19	19.99
240	9.18	9.19	26.14

t= time; q_t = adsorption capacity at time t; q_e = adsorption capacity equilibrium



Appendix D – Optimisation studies (ANN)

Figure D.1: Experimental and ANN predicted Cu(II) removal for the training set. Condition for ANN: training algorithm: Levenberg-Marquardt, number of inputs: 4, number of hidden layer: 6, number of output: 1

Table D.1: ANN preliminary optimisation studies for wastewater matrix (a) at constant initial Cu(II) concentration, (b) at constant pH, and (c) at constant sorbent dosage

Sorbent dosage (g/L)	ANN Removal (%)
0.5	71.86
0.6	71.98
0.7	72.12
2.6	74.76
2.7	75.09
2.8	75.43
1.2	73.36
1.3	73.58
1.4	73.80
4.2	87.15
4.3	87.27
4.4	87.38
	Sorbent dosage (g/L) 0.5 0.6 0.7 2.6 2.7 2.8 1.2 1.3 1.4 4.2 4.3 4.4

(a) At constant initial Cu(II) concentration (3 mg/L)

(b) At constant pH (6.5)

Sorbent dosage (g/L)	Initial Cu(II) concentration (mg/L)	ANN Removal (%)
0.5	1	58.00
0.6	1	58.04
2.4	2	63.85
2.5	2	64.11
6.3	3	82.52
6.4	3	82.59
5.3	4	83.78
5.4	4	83.87
8.1	5	88.28
8.2	5	88.34

рН	Initial concentration (mg/L)	ANN Removal (%)
5	3.9	85.63
5	4	85.69
6	4.5	85.46
6	4.6	85.61
7	3	83.42
7	3.1	83.75
8	4.1	89.81
8	4.2	89.88
8	4.3	89.95
8	4.4	90.01

(c) At constant sorbent dosage (5.25 g/L) $\,$